SCHEDULE 1

The Colleges of Applied Arts and Technology Pension Plan

As Amended and Restated Effective January 1, 2018

Consolidated to December 22, 2023

Including amendments of:

January 1, 2018

May 1, 2018

June 1, 2018

October 1, 2018

January 1, 2020

September 1, 2021

January 1, 2022

February 16, 2022

March 14, 2022

October 28, 2022

November 9, 2022

April 28, 2023

July 27, 2023

December 7, 2023

December 22, 2023

March 25, 2024

September 19, 2024

September 26, 2024

The Colleges of Applied Arts and Technology

Pension Plan

TABLE OF CONTENTS

ARTICLE 1	10
ESTABLISHMENT OF THE PLAN	10
1.01 Establishment of the Plan	10
1.02—Amendments	10
1.03 Continuation of the Plan	10
1.04 Amendment of the Plan	11
1.05 Appendix G Design Two Provisions	11
ARTICLE 2	12
DEFINITIONS	12
ARTICLE 3	29
ELIGIBILITY AND MEMBERSHIP IN THE PLAN	29
3.01 Eligibility for Membership	29
3.02—Continuation of Membership	29
3.03 Membership and Benefit Determination Following Re-employment	30
3.04 Re-employment Following Grievance	35
ARTICLE 4	37
CONTRIBUTIONS	37
4.01 Member Current Service Contributions	37
4.02 Contributions During Periods of Paid Leave of Absence, Reduced Pay or	
Disability	37
4.03 Contributions for Periods of Purchasable Service	39

4.04	Payment of Member Contributions	44
4.05	Employer Contributions	44
4.06	Remittance of Contributions	45
4.07	- Overpayments	45
4.08	Deficiencies or Surplus While Plan is Ongoing	45
ARTICL	E 5	47
PENSI	ONABLE SERVICE	47
5.01	Pensionable Service	47
5.02	Computation of Pensionable Service	48
5.03	Pensionable Service for Periods of Concurrent Employment	48
5.04	Limitations on Pensionable Service	49
ARTICL	E-6	 50
NORM	1AL RETIREMENT	50
6.01	Normal Retirement Pension	 50
6.02	Maximum Pension Benefit	51
6.03	Pension Adjustment	52
6.04	Reduction of Benefits	52
ARTICL	E 7	53
EARL	Y RETIREMENT	53
7.01	Early Retirement Date	53
7.02	Early Retirement Pension	53
	- Unreduced Early Retirement Pension	
	Maximum Pension Benefit	
ARTICL	E 8	57
	DONIED DETIDEMENT	57

8.01 Postponed Retirement Date	57
8.02 Postponed Retirement Pension	57
ARTICLE 9	58
PAYMENT OF PENSIONS	58
9.01 Normal Form of Pension Benefit	58
9.02 Alternate Normal Form of Pension Benefit	58
9.03 Optional Form of Pension Benefit	59
9.04 Surviving Child or Children	60
9.05 Minimum Guarantee	61
9.06 Payment of Pensions	61
9.07 Excess Contributions	63
9.08 Shortened Life Expectancy	63
9.09 Small Pensions	64
9.10 Small Spousal Pensions	65
ARTICLE 10	66
BENEFITS ON TERMINATION OF EMPLOYMENT	66
10.01 Termination with Deferred Pension	66
10.02 Portability Options	66
10.03 Early Commencement of Deferred Pension	68
ARTICLE 11	69
DISABILITY	69
11.01 Accrual of Benefits and Disability Date	69
ARTICLE 12	71
BENEFITS ON DEATH PRIOR TO PENSION COMMENCEMENT.	71
12.01 Death Prior to Pension Commencement	71

12.02 Transfer of Entitlements	73
12.03—Limitation on Benefits	73
ARTICLE 13	74
RECIPROCAL AGREEMENTS	74
13.01—Reciprocal Agreement	74
13.02 Conditions of Transfer and Purchase	74
ARTICLE 14	76
INDEXATION	76
14.01 Inflation Protection Factor	76
14.02 Increases in Pensions in respect of service between 1991 an	d 200877
14.03 Increases in Pensions in respect of service after December 2	007 77
14.04 Application of Increases	77
14.05 Cap Benefit Increases	79
ARTICLE 15	80
TRUST FUND	80
15.01 Operation of the Trust Fund	80
15.02—Provision of Benefits	80
15.03 Expenses	80
15.04 Investments	81
ARTICLE 16	82
ADMINISTRATION OF THE PLAN	82
16.01—Administration	82
16.02—Indemnity	82
16.03—Information	82
ADTICLE 17	0.4

GENERAL PROVISIONS	84
17.01 Employment Rights	84
17.02 Board of Trustees Records	84
17.03 Reduction of Benefits	84
17.04 Incapacity to Receive Payments	84
17.05 Non-Assignability and Non-Commutability of Benefits	85
17.06 Determination of Age and Marital Status	85
17.07 Notice	85
17.08 Construction	86
ARTICLE 18	87
FUTURE OF THE PLAN	87
18.01 Continuation and Amendment of the Plan	87
18.02 No Reduction in Benefits	87
18.03 Termination of the Plan	87
18.04 Cessation of an Employer	88
APPENDIX A	89
PARTICIPATING EMPLOYERS	89
APPENDIX A.1	91
PARTICIPATING EMPLOYERS AND MEMBER CONTRIBUTIONS OF I	DESIGN
TWO EMPLOYEES	91
A.—PARTICIPATING EMPLOYERS UNDER APPENDIX A.1	91
The following constitute Employers, pursuant to Section 2.20 of the Plan:	91
APPENDIX B	205
TRANSFER OF NURSES TO CAAT PENSION PLAN	205
ADDENINY C	206

MEMBER CONTRIBUTIONS	 206
APPENDIX D	210
CONTRIBUTORY EARNINGS	210
APPENDIX E	212
ROYAL ONTARIO MUSEUM AND FOUNDATION	212
APPENDIX F	220
YOUTH SERVICES BUREAU OF OTTAWA	220
APPENDIX G DESIGN TWO PROVISIONS	223
APPENDIX H	269
TORSTAR CORPORATION AND SUBSIDIARIES	269
APPENDIX I	276
CANADIAN PRESS ENTERPRISES INC	 276
APPENDIX J	285
POSTMEDIA NETWORK INC.	285
APPENDIX K	292
COMMUNITY LIVING TORONTO (NON EXECUTIVE MEMBERS)	292
APPENDIX L	298
COMMUNITY LIVING TORONTO (EXECUTIVE MEMBERS)	298
APPENDIX M	303
UNITED WAY OF GREATER TORONTO	303
APPENDIX N	309
FP CANADIAN NEWSPAPERS LIMITED PARTNERSHIP	309
APPENDIX O	315
CATHOLIC CHARITIES OF THE ARCHDIOCESE OF TORONTO AND	
PARTICIPATING MEMBER AGENCIES	315

APPENDIX P	321
THE PRIORY OF CANADA OF THE MOST VENERABLE ORDER OF THE	
HOSPITAL OF ST. JOHN OF JERUSALEM AND PARTICIPATING COUNCILS	. 321
APPENDIX Q	328
SANOFI PASTEUR LIMITED	328
APPENDIX R	335
THE GEORGE HULL CENTRE FOR CHILDREN AND FAMILIES	
APPENDIX S.	
THE GLOBE AND MAIL INC	
APPENDIX T	
MC LEGALMANAGEMENT LIMITED PARTNERSHIP	348
APPENDIX U	354
THE COLLEGE OF FAMILY PHYSICIANS OF CANADA	354
APPENDIX V	362
ANGUS CONSULTING MANAGEMENT LIMITED	362
APPENDIX W	370
HEALTHCARE EXCELLENCE CANADA	370
APPENDIX X	380
LUTHER AN CHURCH—CANADA	380
ARTICLE 1	17
ESTABLISHMENT OF THE PLAN	
1.01 Establishment of the Plan	17
1.02 Amendments	17
1.03 Continuation of the Plan	17

1.04 Amendment of the Plan	<u></u> 18
1.05 Appendix G – Design Two Provisions	<u></u> 18
1.06 Appendix GP—Design Three Provisions	<u></u> 19
ARTICLE 2	<u></u> 19
<u>DEFINITIONS</u>	<u></u> 19
ARTICLE 3	<u></u> 36
ELIGIBILITY AND MEMBERSHIP IN THE PLAN	<u></u> 36
3.01 Eligibility for Membership	<u></u> 36
3.02 Continuation of Membership	<u></u> 36
3.03 Membership and Benefit Determination Following Re-employment	<u></u> 37
3.04 Re-employment Following Grievance	<u></u> 42
ARTICLE 4	<u></u> 44
CONTRIBUTIONS	<u></u> 44
4.01 Member Current Service Contributions	<u></u> 44
4.02 Contributions During Periods of Paid Leave of Absence, Reduced Pay	<u>or</u>
<u>Disability</u>	<u></u> 44
4.03 Contributions for Periods of Purchasable Service	<u></u> 46
4.04 Payment of Member Contributions	<u></u> 51
4.05 Employer Contributions	<u></u> 51
4.06 Remittance of Contributions	52
4.07 Overpayments	<u></u> 52
4.08 Deficiencies or Surplus While Plan is Ongoing	<u></u> 52
ARTICLE 5	<u></u> 54
PENSIONABLE SERVICE	<u></u> 54
5.01 Pangianghla Sarvica	5.1

5.02	Computation of Pensionable Service.	<u>.</u> 55
5.03	Pensionable Service for Periods of Concurrent Employment	<u>.</u> 55
5.04	Limitations on Pensionable Service	<u>.</u> 56
ARTICLE	6	<u>.</u> 57
NORM.	AL RETIREMENT	<u>.</u> 57
6.01	Normal Retirement Pension.	<u>.</u> 57
6.02	Maximum Pension Benefit	<u>.</u> 58
6.03	Pension Adjustment	<u>.</u> 59
6.04	Reduction of Benefits	<u>.</u> 59
ARTICLE	7	<u>.</u> 60
EARLY	RETIREMENT	<u>.</u> 60
7.01	Early Retirement Date	<u>.</u> 60
7.02	Early Retirement Pension	<u>.</u> 60
7.03	Unreduced Early Retirement Pension.	<u>.</u> 62
7.04	Maximum Pension Benefit	<u>.</u> 62
ARTICLE	8	<u>.</u> 64
POSTPO	ONED RETIREMENT	<u>.</u> 64
8.01	Postponed Retirement Date	<u>.</u> 64
8.02	Postponed Retirement Pension	<u>.</u> 64
ARTICLE	9	<u>.</u> 65
PAYMI	ENT OF PENSIONS	<u>.</u> 65
9.01	Normal Form of Pension Benefit	<u>.</u> 65
9.02	Alternate Normal Form of Pension Benefit.	<u>.</u> 65
9.03	Optional Form of Pension Benefit	<u>.</u> 66
9.04	Surviving Child or Children	67

9.05	Minimum Guarantee 6	8
9.06	Payment of Pensions 6	8
9.07	Excess Contributions	0'
9.08	Shortened Life Expectancy	0'
9.09	Small Pensions	1
9.10	Small Spousal Pensions	'2
ARTICLE	E 10	'3
BENEFIT	S ON TERMINATION OF EMPLOYMENT	'3
10.01	Termination with Deferred Pension	'3
10.02	Portability Options	'3
10.03	Early Commencement of Deferred Pension	′5
ARTICLE	<u> </u>	'6
DISAB	ILITY	'6
11.01	Accrual of Benefits and Disability Date	'6
ARTICLE	<u>5 12</u> 7	8'
BENEF	TITS ON DEATH PRIOR TO PENSION COMMENCEMENT7	'8
12.01	Death Prior to Pension Commencement	'8
12.02	Transfer of Entitlements	30
12.03	Limitation on Benefits	30
ARTICLE	<u>8 13</u> 8	31
RECIPI	ROCAL AGREEMENTS8	31
13.01	Reciprocal Agreement	31
13.02	Conditions of Transfer and Purchase	31
ARTICLE	<u>8</u> 14	3
INDEXA	FION	

14.01	Inflation Protection Factor	_83
14.02	Increases in Pensions in respect of service between 1991 and 2008	84
14.03	Increases in Pensions in respect of service after December 2007	_84
14.04	Application of Increases	84
14.05	Cap Benefit Increases	_86
ARTICLE	15	87
TRUST	FUND	87
15.01	Operation of the Trust Fund	_87
15.02	Provision of Benefits	87
15.03	Expenses	87
15.04	Investments	88
ARTICLE	16	89
ADMIN	ISTRATION OF THE PLAN	89
16.01	Administration	89
16.02	Indemnity	89
16.03	Information	89
ARTICLE	17	91
GENER.	AL PROVISIONS	91
<u>17.01 En</u>	nployment Rights	91
17.02	Board of Trustees Records	91
17.03	Reduction of Benefits	91
17.04	Incapacity to Receive Payments	91
17.05	Non-Assignability and Non-Commutability of Benefits	92
17.06	Determination of Age and Marital Status	92
17.07	N-4:	02

17.08 Construction.	93
ARTICLE 18	94
FUTURE OF THE PLAN	94
18.01 Continuation and Amendment of the Plan	94
18.02 No Reduction in Benefits	94
18.03 Termination of the Plan	94
18.04 Cessation of an Employer	95
APPENDIX A	96
PARTICIPATING EMPLOYERS	96
APPENDIX A.1	98
PARTICIPATING EMPLOYERS AND MEMBER CONTRIBUTIONS OF DESIGN	
TWO EMPLOYEES.	98
A. PARTICIPATING EMPLOYERS UNDER APPENDIX A.1	98
The following constitute Employers, pursuant to Section 2.20 of the Plan:	98
APPENDIX B 22	25
TRANSFER OF NURSES TO CAAT PENSION PLAN	25
APPENDIX C 22	26
MEMBER CONTRIBUTIONS22	26
APPENDIX D	30
CONTRIBUTORY EARNINGS2	30
APPENDIX E	32
ROYAL ONTARIO MUSEUM AND FOUNDATION23	32
APPENDIX F24	40
YOUTH SERVICES BUREAU OF OTTAWA24	40
ADDENDIV G DESIGN TWO DROVISIONS	12

APPENDIX GP—DESIGN THREE PROVISIONS	_289
APPENDIX H	<u>.</u> 300
TORSTAR CORPORATION AND SUBSIDIARIES	<u>.</u> 300
APPENDIX I	<u>.</u> 307
CANADIAN PRESS ENTERPRISES INC.	<u>.</u> 307
APPENDIX J	<u>.</u> 316
POSTMEDIA NETWORK INC.	<u>.</u> 316
APPENDIX K	<u>.</u> 323
COMMUNITY LIVING TORONTO (NON-EXECUTIVE MEMBERS)	<u>.</u> 323
APPENDIX L	<u>.</u> 329
COMMUNITY LIVING TORONTO (EXECUTIVE MEMBERS)	<u>.</u> 329
APPENDIX M	<u>.</u> 334
UNITED WAY OF GREATER TORONTO	<u>.</u> 334
APPENDIX N	<u>.</u> 309
FP CANADIAN NEWSPAPERS LIMITED PARTNERSHIP	<u>.</u> 309
APPENDIX O	<u>.</u> 315
CATHOLIC CHARITIES OF THE ARCHDIOCESE OF TORONTO AND	
PARTICIPATING MEMBER AGENCIES	<u>.</u> 315
APPENDIX P	<u>.</u> 321
THE PRIORY OF CANADA OF THE MOST VENERABLE ORDER OF THE	
HOSPITAL OF ST. JOHN OF JERUSALEM AND PARTICIPATING COUNCILS	<u>.</u> 321
APPENDIX Q	<u>.</u> 328
SANOFI PASTEUR LIMITED	<u>.</u> 328
APPENDIX R	<u>.</u> 335
THE GEORGE HULL CENTRE FOR CHILDREN AND FAMILIES	<u>.</u> 335
APPENDIX S	341

THE GLOBE AND MAIL INC	341
APPENDIX T	348
MC LEGALMANAGEMENT LIMITED PARTNERSHIP	348
APPENDIX U	354
THE COLLEGE OF FAMILY PHYSICIANS OF CANADA	354
APPENDIX V	362
ANGUS CONSULTING MANAGEMENT LIMITED	362
APPENDIX W	370
HEALTHCARE EXCELLENCE CANADA	370
APPENDIX X	380
LUTHED AN CHURCH CANADA	380

ARTICLE 1

ESTABLISHMENT OF THE PLAN

1.01 Establishment of the Plan

Effective June 1, 1967, the Colleges of Applied Arts and Technology Pension Plan was established to provide pensions and other benefits to eligible Employees of participating Colleges of Applied Arts and Technology.

1.02 Amendments

Since the establishment of the Plan, the Plan text has from time to time been amended, supplemented and restated. This amended and restated Plan text, effective January 1, 2018, shall apply with respect to the determination of benefits payable in the event of retirements, deaths or other terminations of membership occurring on or after January 1, 2018, unless specifically provided otherwise in the Plan.

Benefits payable under the Plan in respect of retirements, deaths or terminations of membership occurring prior to January 1, 2018 to Former Members, Retired Members or others entitled to such benefits shall continue to be paid in accordance with the terms of the Plan as constituted at the relevant date, unless otherwise provided or subsequently specifically amended by the terms of the Plan.

Pension benefits accrued under the Plan as amended and restated effective January 1, 2018 for service prior to January 1, 2018 in respect of Members at December 31, 2017 will be no greater or less than the benefits accrued under the terms of the Plan as in effect on December 31, 2017.

1.03 Continuation of the Plan

The Plan, as stated herein and as it may be amended from time to time, will remain in effect subject to its continued registration by

- (1) the relevant tax authorities as is necessary to ensure that Members are entitled to deduct the amount of their contributions and/or payments from their incomes before taxes under the provisions of the Income Tax Act or any other applicable tax laws as are now in effect or as may hereafter be amended or adopted; and
- (2) the Financial Services Regulatory Authority of Ontario or such successor entity pursuant to the provisions of the Act as amended from time to time.

1.04 Amendment of the Plan

In the event that any provision of the Plan is less favourable to Members, Former Members, Retired Members or others entitled to benefits from the Plan than is required by the terms of any applicable provincial or federal legislation, the Plan shall be deemed amended accordingly for those individuals whose jurisdiction of employment is not Ontario, but only to the extent necessary to remedy any such deficiencies.

For the avoidance of doubt, the Plan provisions will prevail where they exceed the minimum standards of any governing Canadian legislation applicable to benefit entitlements of Members, Former Members, Retired Members or eligible others.

1.05 Appendix G – Design Two Provisions

The Plan is amended effective June 1, 2018 by adding Appendix G.

Effective June 1, 2018, the benefits, rights and obligations of Design Two Employees shall be in accordance with the Design Two Provisions.

On and after January 1, 2019, only Members who are Full-time Employees shall be permitted to accrue benefits under the Design One Provisions.

Effective January 1, 2019, the benefits, rights and obligations of Other Than Regular Full-time Employees shall be in accordance with the Design Two Provisions.

1.06 Appendix GP—Design Three Provisions

The Plan is amended effective August 1, 2024, by adding Appendix GP.

ARTICLE 2

DEFINITIONS

The following words and phrases shall, for the purposes of the Plan, have the following meanings respectively, unless a different meaning is plainly required by the context.

2.01 "Act"

Shall mean the *Pension Benefits Act of Ontario*, *R.S.O. 1990* or when referencing a benefit entitlement in respect of a Member, Former Member or Retired Member shall mean the *Pension Benefits Act of Ontario*, *R.S.O. 1990* or any other Canadian legislation which applies to the benefit entitlements of a Member, Former Member, Retired Member or eligible other, and any future legislation amending, supplementing, superseding or incorporating it, including any regulations issued pursuant to the same, and including any other pension benefits legislation having jurisdiction over the Plan.

2.02 "Actuarial Equivalent"

Shall mean an actuarially equivalent value computed by reference to the rate of interest and actuarial tables which do not distinguish between males and females, adopted by the Plan Manager at the relevant time for purposes of the Plan, subject to any requirements of the Act and the Income Tax Act.

2.03 "Actuarial Value"

Shall mean the actuarial present value of the applicable pension and ancillary benefits determined as of the date a computation is required in accordance with, and subject to, the procedures and assumptions adopted by the Board of Trustees.

2.04 "Actuary"

Shall mean the actuary or firm of actuaries retained by, but independent of, the Board of Trustees and qualified through Fellowship in the Canadian Institute of Actuaries.

2.05 "Average YMPE"

Shall mean the average of the YMPE for the year in which the Member's Termination of Employment Date occurs and for each of the four preceding years.

2.06 "Beneficiary"

Shall be the person or persons named by a Member, Former Member, Retired Member or Spouse entitled to or in receipt of a pension to receive the death benefits, if any, payable to a Beneficiary under the Plan and the Beneficiary, if any, shall be so named by the Member, Former Member, Retired Member or Spouse:

- (1) by delivering the applicable written designation to the Plan Manager, or
- (2) by will, but only if the designation in the will relates expressly to benefits under the Plan either generally or specifically and if such will is provided to the Plan prior to any death benefits being paid.

If a Member, Former Member, Retired Member or Spouse does not designate, or does not validly designate, any Beneficiary or designates a Beneficiary who is not living on the date a determination is required, any benefit payable under the Plan to a Beneficiary of the Member, Former Member, Retired Member or Spouse will be paid to the estate of the Member, Former Member, Retired Member or Spouse. Furthermore, subject to applicable laws, each Member, Former Member, Retired Member or Spouse entitled to or in receipt of a pension shall have the right to change, revoke or amend any beneficiary designation by either of the methods provided above prior to their death and no such

change, revocation or amendment shall require the written consent of such Beneficiary previously designated by such Member, Former Member, Retired Member or Spouse.

2.07 "Board of Trustees"

Shall have the same meaning as in the Sponsorship and Trust Agreement.

2.08 "Child" or "Children"

Shall mean a dependent child or children of a Member, Former Member or Retired Member upon the death of the Member, Former Member or Retired Member and who

- (1) has not reached the age of 18 years, or
- (2) is in full-time attendance at an educational institution at the time of the Member, Former Member or Retired Member's death and has not reached the age of 25 years, or
- (3) is unable, due to mental or physical injury or disease, to support themselves and became disabled before the age of 18 years or before the age of 25 years if the child was in full-time attendance at an educational institution,

and has provided the Plan Manager with evidence of the above that the Plan Manager deems appropriate.

2.09 "Commuted Value"

Shall mean in relation to benefits that a person has a present or future entitlement to receive, a lump sum amount which is the actuarial present value of those benefits computed as of the date a determination is required, at the rate of interest and using the actuarial tables and rules for the computation of commuted values, subject to the Act and the Income Tax Act.

2.10 "Continuous Service"

Shall mean the period of unbroken employment of an Employee with an Employer and shall be deemed not to be broken by:

- (1) a leave of absence with full pay for any reason where the Employee's Employer has authorized the leave;
- (2) a leave of absence without pay for any reason where the Employee's Employer has authorized the leave;
- (3) statutory leaves under employment standards legislation applicable to the Member and which requires such Member to be allowed to continue participation in the Plan for the duration of such leave;
- (4) a period of layoff for so long as recall rights exist provided that the Employee has not transferred any benefit entitlement out of the Trust Fund pursuant to Section 10.02 or otherwise;
- (5) an absence due to strike or lockout, provided no termination of employment has occurred and the Employee returns to work following the end of the strike or lockout;
- (6) periods during which the Member meets the definition of Disabled in accordance with Section 2.15, and has not transferred any benefit entitlement out of the Trust Fund pursuant to Section 10.02 or otherwise.

2.11 "Contributory Earnings"

Shall mean:

- (1) Before January 1, 2019, basic salary and wages received by a Member from an Employer listed in Appendix A upon which contributions are made and such other forms of remuneration received by a Member from an Employer listed in Appendix A as determined in accordance with Appendix D.
- (2) On and after January 1, 2019, basic salary and wages received by a Member who is a Full-time Employee upon which contributions are made and such other forms of remuneration received by a Member who is a

- Full-time Employee from an Employer listed in Appendix A in respect of their employment as a Full-time Employee as determined in accordance with Appendix D.
- (3) For greater certainty, any payments or elements of remuneration that result from any arrangement the effect of which is to inflate Highest Average Pensionable Earnings without a bona fide change in employment, job function or pay scale shall be excluded from Contributory Earnings.

2.12 "Credited Interest"

Shall mean:

- interest on a Member's contributions at the applicable rate determined under (4) below:
 - (a) for the period prior to January 1, 1988, credited from the end of the calendar year in which the contribution is paid into the Trust Fund, and compounded annually thereafter to December 31, 1987, and
 - (b) for the period commencing on January 1, 1988, compounded annually and credited
 - at the end of each calendar year, on the balance of contributions at the beginning of the calendar year; plus
 - (ii) at the end of each calendar year, on the balance of contributions made during the year at a pro-rata portion of the rate determined under (4) below for that year; plus
 - (iii) at the end of the month in which the Member's Termination of Membership Date occurs, on the balance of contributions at the beginning of that calendar year and the balance of contributions made during that calendar year, at the appropriate pro-rata portion of the rate determined in

under (4) below for the preceding calendar year, for the complete calendar months in the current calendar year to the end of the month in which the Member's Termination of Membership Date occurs;

- (2) interest on lump sum payments out of the Trust Fund other than payments of Commuted Values, compounded annually and credited from the date at which a determination is required to the end of the calendar month in which payment is made at the rate which is credited to Member contributions under (1) above; and
- (3) interest on the payment of a Commuted Value out of the Trust Fund and calculated and compounded annually from the date as of which the Commuted Value was determined or re-determined to the end of the calendar month of payment, at the same rate that was used to determine the Commuted Value; and
- (4) the applicable interest rate for determinations under (1) and (2) above are:
 - (a) for the period ending December 31, 1977, three per cent per annum;
 - (b) for the period commencing on January 1, 1978 and ending on December 31, 1984, five per cent per annum;
 - (c) for the period commencing on January 1, 1985 and ending on December 31, 1987, an annual rate calculated at one per cent below the average of the monthly rates of interest paid by the Chartered Banks on 90 day deposit receipts quoted for the immediately preceding year in the Bank of Canada Review, determined on December 31st of each year;

- (d) for the period commencing on January 1, 1988, at the rate which is the greatest of (i) through (iii) below for the applicable calendar year:
 - the average of the rates quoted for each of the twelve months in the immediately preceding calendar year in the Bank of Canada Review for five year personal fixed term chartered bank deposits;
 - (ii) for the period prior to December 31, 2017, the Canada Savings Bond Rate applied to the last such bonds issued in the previous year plus 1%; and
 - (iii) the minimum prescribed by law.

2.13 "Deemed Contributory Earnings"

Shall mean the amount of Contributory Earnings a Member would have received during a Period of Reduced Pay, leave, work stoppage or other absence from work, as applicable, which amount shall be based on the Member's rate of Contributory Earnings in the period immediately preceding such Period of Reduced Pay, leave, work stoppage or other absence, adjusted from time to time, if applicable to the Member, by any increases in those elements of Contributory Earnings as a result of:

- (1) negotiated increases under a collective bargaining agreement;
- (2) increases resulting from normal progression through the earnings classification; or,
- (3) increases obtained through the grievance arbitration procedure under a collective bargaining agreement or through a similar mechanism.

2.14 "Deferred Pension"

Shall mean the pension benefit determined pursuant to Article 10 and as may be adjusted pursuant to Article 14.

2.14.1 "Design One Provisions"

Shall mean the provisions of the Plan excluding the Design Two Provisions except where the Design One Provisions are specifically referenced in the Design Two Provisions.

2.14.2 "Design Two Employee"

Shall mean a person who is employed by an Employer listed in Appendix A.1.

2.14.3 "Design Two Provisions"

Shall mean the provisions of Appendix G.

2.15 "Disabled"

Shall mean, in reference to a Member, disability due to physical or mental injury or disease which prevents the Member from performing the duties of the employment in which the Member was engaged before the disability commenced, as determined using such evidence the Plan Manager deems appropriate, in accordance with Applicable Laws.

- 2.16 "Disability Date" of a Member shall have the meaning provided in Section 11.01.
- 2.17 "Early Retirement Date" has the meaning set forth in Section 7.01.
- **2.18** "Early Retirement Pension" shall mean the pension benefit determined pursuant to Section 7.02.
- 2.19 "Employee" shall mean a person who is employed by an Employer and includes Full-time Employees, Other Than Regular Full-time Employees and Design Two Employees.

- **2.20** "Employer" shall mean any employer listed in Appendix A or Appendix A.1. Employer shall also include the Board of Trustees and subject to the approval of the Board of Trustees, any corporation controlled by the Board of Trustees.
- **2.21** "Excess Contributions" shall mean the excess of (1) over (2) as follows:
 - the Member's contributions made in accordance with Article 4 plus
 Credited Interest to the date a determination is required;
 - (2) 50% of the Commuted Value of the Member's pension benefit in respect of all Pensionable Service, calculated as of the same date as in subsection (1);

provided that the Member contributions in (1) above and the Member's pension benefit in (2) above will not apply to benefits resulting from the election by a Member and to the Member's contributions made to purchase Pensionable Service in respect of which the Employer makes no contributions pursuant to Section 4.05 in respect of the Member.

2.22 "Former Member" shall mean:

- a person who has attained their Termination of Membership Date and who is entitled to a benefit under the Design One Provisions, excluding Retired Members; or
- (2) except for the purposes of Article 14, a person who is a Former Member as defined under the Design Two Provisions.
- **2.23** <u>"Full-time Employee"</u> shall mean a person who is employed by, and holds a regular full-time position with, an Employer listed in Appendix A.
- 2.24 "Highest Average Pensionable Earnings" shall mean the average of the Member's Pensionable Earnings for the 60 consecutive full and partial months of their Pensionable Service during which the Member's Pensionable Earnings were the highest, expressed as an annual amount and computed as in (1) below, subject to

the provisions in (2) below and any applicable policies and procedures that may be adopted by the Board of Trustees.

- (1) A Member's Highest Average Pensionable Earnings will be an amount equal to the sum of the Member's Pensionable Earnings for each full and partial month of such 60 month period, to a total of 60 months, divided by 5.
- (2) For purposes of a computation hereunder:
 - (a) if a Member has less than 60 consecutive full and partial months of Pensionable Service in which the Member received or was deemed to have received Pensionable Earnings, the Member's Highest Average Pensionable Earnings will be computed over the actual period of Pensionable Service;
 - (b) if a Member has a prior period of employment reinstated as Pensionable Service under Section 3.03, the period between the prior periods of employment and the date the Member last became an Employee will not be considered a break in the consecutive period of Pensionable Service.
 - (c) If a Member returns to employment with an Employer prior to attaining their Termination of Membership Date, the period between the prior period of employment and the date the Member last became an Employee will not be considered a break in the consecutive period of Pensionable Service.
- 2.25 "Income Tax Act" shall mean the Income Tax Act (Canada) and any Regulations and Rules made pursuant thereto or issued by the Canada Revenue Agency or such successor entity, as amended from time to time.

- 2.25.1 "Jurisdiction of Employment" refers to the legislation that applies to a Member, Former Member or Retired Member based on the province of employment while a Member"
- 2.26 "Member" shall mean an individual who is enrolled in the Plan and continues to be entitled to benefits under the Plan but excludes Former Members and Retired Members.
- 2.27 "Non-Participating Employer" shall mean an employer who sponsors or contributes to a Canadian registered pension plan under which a person has accrued pension benefits and is no longer accruing benefits under such plan, but does not include an Employer or Predecessor Employer.
- 2.28 "Normal Retirement Date" of a Member or Former Member shall be the last day of the month coincident with or next following the Member's or Former Member's attainment of age 65.
- 2.29 "Normal Retirement Pension" shall mean the pension benefit provided in Section6.01 as of a Normal Retirement Date.
- **2.30** "Other Than Regular Full-time Employee" or "OTRFT Employee" shall mean any person who is employed by an Employer listed in Appendix A who is not a Full-time Employee.
- 2.31 "Pensionable Earnings" shall mean:
 - (1) a Member's Contributory Earnings or deemed Contributory Earnings on which the Member makes contributions pursuant to Article 4;
 - (2) for periods of Transferred In Service, the actual earnings received by the Member from the Non-Participating Employer for the period of Transferred In Service;

- (3) for any period of Pensionable Service under Section 5.01(4) in respect of which the Member meets the definition of Disabled in accordance with Section 2.15, the Member's Pensionable Earnings
 - (a) for the balance of the calendar year in which the Member becomes

 Disabled, commencing on the Member's Disability Date, will be
 deemed to be the rate of the Employee's Contributory Earnings in
 the period prior to the Member's Disability Date; and
 - (b) for each calendar year the Member remains Disabled after the end of the calendar year in which the Member became Disabled, the deemed rate of Pensionable Earnings in subsection (a) shall be increased on the January 1st of that year in accordance with the inflation protection factor as provided in Section 14.01 provided that in respect of the first calendar year in which such inflation protection factor is applied such increase shall be pro-rated for that portion of the prior year from the Disability Date to December 31;
- (4) for prior periods of Pensionable Service reinstated pursuant to Section 3.04(1), the Member's Contributory Earnings under Section 4.01 in respect of those periods.
- (5) For any calendar year in which a Member works less than full-time, the Member's Pensionable Earnings for the year will be annualized by dividing the Member's Contributory Earnings for the year by the Member's Pensionable Service for the year.
- (6) For greater certainty, before January 1, 2019, only a Member employed by an Employer listed in Appendix A may have Pensionable Earnings, and, on and after January 1, 2019, only a Member who is a Full-time Employee may have Pensionable Earnings.
- **2.32** "Pensionable Service" shall be determined in accordance with the provisions of Article 5.

- 2.33 "Period of Reduced Pay" shall mean a period throughout which:
 - (1) the Member renders services to an Employer and the Member's job classification does not change; and
 - (2) the remuneration received by the Member from an Employer is less than the remuneration the Member would have received if the Member had rendered services on the same basis as was in effect before the period of reduced services commenced.

To qualify for a Period of Reduced Pay, a Member must have been employed by one or more Employers on a full-time basis for not less than 36 months in the aggregate prior to the commencement of the Period of Reduced Pay.

- **2.34** "Plan" shall mean The Colleges of Applied Arts and Technology Pension Plan, as stated herein and as amended from time to time.
- **2.35** "Plan Manager" shall mean the person appointed by the Board of Trustees to hold that office.
- 2.36 "Postponed Retirement Date" has the meaning set forth in Section 8.01.
- 2.37 "Postponed Retirement Pension" has the meaning set forth in Section 8.02.
- **2.38** "Predecessor Employer" shall mean an employer whose operations were acquired or otherwise assumed by an Employer.
- **2.39** "Purchasable Service" shall mean the following periods of a Member's employment:
 - a leave of absence from an Employer without pay for any reason where the Member's Employer has authorized such leave;
 - (2) a period of layoff from an Employer for so long as recall rights exist provided that the Member has not transferred any benefit entitlement out

- of the Trust Fund as provided in Section 10.02 or otherwise and the Member is recalled and returns to work;
- (3) a statutory leave under employment standards legislation applicable to the Member and which requires such Member to be allowed to continue participation in the Plan for the duration of such leave from an Employer;
- (4) a period of employment as an Employee with any Employer before the Member became a contributor;
- (5) prior periods of Pensionable Service under the Plan prior to the date the Member last joined the Plan for which the Member is no longer entitled to Pensionable Service or in respect of which the Member transferred the Commuted Value out of the Trust Fund as provided in Section 10.02 or otherwise;
- (6) a period of employment with a Predecessor Employer;
- a Member's period of employment with an Employer during which a work stoppage occurred;
- (8) a period of employment with a Non-Participating Employer in respect of which benefits accrued under a registered pension plan while not accruing a pension under the Plan.
- 2.40 "Reciprocal Agreement" shall mean an agreement, as provided in Article 13, between the Board of Trustees and a person authorized for that purpose under a registered pension plan established by or for any Non-Participating Employer that provides for the transfer of pension credits to the Trust Fund on behalf of a Member for a period of former employment with that Non-Participating Employer.
- 2.41 "Registered Retirement Savings Arrangement" shall mean a registered retirement savings plan established in accordance with the Income Tax Act or a registered retirement income fund established in accordance with the Income Tax Act.

2.42 "Retired Member" shall mean:

- (1) a person who has either attained their Termination of Employment Date or Termination of Membership Date and who
 - (a) is receiving or has elected in accordance with the Design One Provisions to receive a Normal Retirement Pension; a Postponed Retirement Pension; an Early Retirement Pension; or a pension determined in accordance with Section 10.03; or
- (b) has attained their Normal Retirement Date under the Design One Provisions; or
- (2) except for the purposes of Article 14, a person who is a Retired Member as defined under the Design Two Provisions.
- **2.43** "Sponsors' Committee" shall have the same meaning as in the Sponsorship and Trust Agreement.
- 2.44 "Sponsorship and Trust Agreement" shall mean the agreement initially entered into by the Employers, and the Ontario Public Service Employees Union and the initial members of the Board of Trustees in respect of the administration of the Plan and the Trust Fund, as amended from time to time.
- 2.45 "Spouse" in respect of a Member, Former Member or Retired Member, shall be as defined under the Act as at the time a determination of marital status is required.
- 2.46 "Termination of Employment Date" of a Member shall mean the date on which a Member's termination of employment with an Employer occurs or the date of any other interruption of Continuous Service for any reason.
- 2.47 "<u>Termination of Membership Date</u>" of a Member shall mean the date that is twenty-four (24) consecutive months after the Termination of Employment Date

during which no contributions have been made by or on behalf of the Member in respect of current service, unless coincident with or after the Termination of Employment Date, and prior to the expiry of the twenty-four (24) month period:

- (1) the Member requests to transfer the Commuted Value of the Member's pension entitlement under the Plan to the pension fund of another pension plan pursuant to Section 10.02 and the administrator of the other pension plan agrees to accept the payment;
- (2) the Member elects an Early Retirement Date;
- (3) the Member attains their Normal Retirement Date;
- (4) The Member dies;

in which case, the Termination of Membership Date shall be applicable date above. Notwithstanding the foregoing:

- (5) in the event a Member resigns their membership in the Plan pursuant to Section 9.08, the Member shall be deemed to have attained their Termination of Membership Date as of the effective date of the Member's resignation under the Plan;
- (6) in the event a Member attains a Postponed Retirement Date, then the Member shall be deemed to have attained their Termination of Membership Date as of their Postponed Retirement Date.
- 2.48 "Transferred In Service" shall mean that portion of a Member's period of employment with a Non-Participating Employer that is purchased by a transfer of pension credits to the Trust Fund by way of a Reciprocal Agreement.
- 2.49 "Trust Fund" shall mean all of the assets of the Plan trust fund consolidated with all funds and assets received from time to time by way of contributions, transfers or other amounts, together with all increments, earnings and profits accrued

Formatted: English (Canada)

- thereon from the administration and investment of the said Trust Fund and from which pensions and other benefits under the Plan are to be paid.
- 2.50 "YMPE" shall mean in respect of any calendar year, the Year's Maximum Pensionable Earnings under the Canada Pension Plan applicable to that calendar year.
- 2.51 "Gender and Number" Reference to the third person will include the male gender and the female gender unless the context otherwise requires. Words importing the singular number may be construed to extend to and include the plural number, and words importing the plural number may be construed to extend to and include the singular.
- 2.52 "Headings Not Part of Plan" The titles given to the various Articles and Sections of the Plan are inserted for convenience and reference only and are not part of this Plan, and they shall not be considered in determining the purpose, meaning, or intention of any provision of the Plan.
- 2.53 "References" Unless otherwise indicated, a reference in the Design One Provisions to a provision, Section, or Article is a reference to a provision, Section, or Article of the Design One Provisions.

ARTICLE 3

ELIGIBILITY AND MEMBERSHIP IN THE PLAN

3.01 Eligibility for Membership

(1) Full-time Employees

Each Full-time Employee shall become a Member of the Plan upon the Employee's date of hire provided the Employee's date of hire is prior to the end of the year in which the Employee reaches age 71, or such other age as is specified in the Income Tax Act (Canada).

(2) Other Than Regular Full-time (OTRFT) Employees

Each Other Than Regular Full-time (OTRFT) Employee who becomes a Full-time Employee prior to electing to join the Plan will become a Member immediately upon becoming a Full-time Employee. Each Design Two Employee who becomes a Full-time Employee prior to joining the Plan will become a Member immediately upon becoming a Full-time Employee.

(3) Excluded Employees

An Employee who is participating and accruing benefits under the Ontario Teachers' Pension Plan, or The Pension Plan for NORCAT, or any other pension plan as designated by the Sponsors' Committee is not eligible to become a Member of the Plan.

3.02 Continuation of Membership

A person who becomes a Member of the Plan shall remain a Member until their Termination of Membership Date.

3.03 Membership and Benefit Determination Following Re-employment

(1) Membership on Re-employment Prior to Pension Commencement

(a) Prior Member With No Benefit Entitlement

A person who is neither a Former Member nor a Retired Member, but who was previously a Member of the Plan, and who is hired as a Full-time Employee will be considered a new Employee for all purposes of the Plan and the provisions of Section 3.01 shall apply.

(b) Former Member

If a Former Member is hired as a Full-time Employee, the Former Member shall become a Member immediately upon hire as a Full-time Employee and the provisions in (i), (ii), and (iii) shall apply.

- (i) If the person received payment of Excess Contributions in respect of a prior period of membership in the Plan under the Design One Provisions, any Deferred Pension in respect of such prior period of membership will remain frozen and the provisions of Section 3.03(3)(a) below will apply for the determination of the pension benefit to which the person is entitled under the Design One Provisions.
- (ii) Notwithstanding the provisions of (i) above, the person who received payment of Excess Contributions may elect to repay to the Trust Fund the amount of the refund of Excess Contributions, with Credited Interest from the date of the prior payment to the date of repayment and, provided the repayment is made within 6 months following the date of hire as a Full-time Employee, upon repayment, or if the person did not receive a payment of Excess Contributions in respect of a prior period of membership in the Plan under the Design One Provisions, the pension benefit to which the

Member is entitled under the Design One Provisions will be determined in accordance with the provisions of Section 3.03(3)(b) and that benefit will be in lieu of the Deferred Pension to which the Member was entitled in respect of the prior period of membership under the Design One Provisions.

(iii) Any pension benefit to which the Member may be entitled under the Design Two Provisions at the Member's subsequent Termination of Membership Date will not be affected by the Member's subsequent period of membership.

(2) Membership on Re-employment After Pension Commencement

(a) Before Normal Retirement Date

If a Retired Member is hired as a Full-time Employee before their Normal Retirement Date, pension payments will cease immediately and the Retired Member will become a Member and recommence contributions under Section 4.01 immediately on such hire, the pension benefit to which the Member will be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will be determined in accordance with Section 3.03(3)(c), and any pension benefit to which the Member may be entitled under the Design Two Provisions at the Member's subsequent Termination of Membership Date will not be affected by the Member's subsequent period of membership.

(b) After Normal Retirement Date

If a Retired Member is hired as a Full-time Employee after their Normal Retirement Date but before the end of the year in which

the Retired Member reaches age 71, or such other age as is specified in the Income Tax Act (Canada), the Retired Member may elect either:

- (i) to continue to receive pension payments and not become a Member of the Plan, or
- (ii) to become a Member of the Plan in which case pension payments will cease immediately and the Retired Member will become a Member and recommence contributions under Section 4.01 immediately on rehire, the pension benefit to which the Member will be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will be determined in accordance with Section 3.03(3), and any pension benefit to which the Member may be entitled under the Design Two Provisions at the Member's subsequent Termination of Membership Date will not be affected by the Member's subsequent period of membership.

If the Retired Member is hired by an Employer after the end of the year in which the Retired Member reaches age 71, or such other age as is specified in the Income Tax Act (Canada), the Retired Member will continue to receive pension payments and shall not become a Member of the Plan.

(3) Determination of Pension Benefits for Re-employed Members

(a) Re-employed Former Members – No Re-Payment of Excess

If a Former Member is hired as a Full-time Employee and Section 3.03(1)(b)(i) applies, the benefit to which the Member will be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will be the sum of:

- the Deferred Pension to which the Member is entitled in respect of such prior period of Pensionable Service, and
- (ii) the pension benefit determined under the Plan based on the Member's Pensionable Service and Highest Average Pensionable Earnings for the period after the Member's re-employment.

(b) Re-employed Members

If a Former Member is hired as a Full-time Employee and Section 3.03(1)(b)(ii) applies, the benefit to which the Member will be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will be the greater of (i) and (ii) below:

- (i) the benefit determined pursuant to (a) above, and
- (ii) the pension benefit determined under the Plan based on
 - A. the Member's Pensionable Service in respect of the applicable Deferred Pension to which the Member is entitled in respect of such prior period of Pensionable Service, plus Pensionable Service accrued after the Member's re-employment;
 - B. Highest Average Pensionable Earnings based on the Member's Pensionable Earnings during such prior period of membership and after the Member's reemployment and the period of absence between the prior period of membership and the Member's reemployment will not be considered a break in the consecutive period of Pensionable Earnings; and
 - the provisions of the Plan in effect at the Member's subsequent Termination of Membership Date; and

the benefit so determined will be Member's Pension in respect of all such periods of Pensionable Service.

(c) Re-Employed Retired Members

If a Retired Member is hired as a Full-time Employee and is required or elects under Section 3.03(2) to become a Member of the Plan, the benefit to which the Member will be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will be the greater of (i) and (ii), plus (iii):

- (i) The sum of A and B below:
 - A. the pension in pay at the time the Member was reemployed and rejoined the Plan, plus any inflation
 protection that would have been received during the
 period of re-employment, adjusted to what the
 Retired Member would have been receiving under
 the normal form of pension in the event they elected
 an optional form; plus
 - B. the pension accrued during the period of reemployment, as in Section 3.03(3)(a)(ii);
- (ii) the pension benefit determined under Section 6.01, based on A, B and C below, and reduced as per D:
 - A. the Member's Pensionable Service accrued under the Plan prior to retirement, plus Pensionable Service accrued after the Member's re-employment;
 - B. Highest Average Pensionable Earnings based on the Member's Pensionable Earnings during the prior period of membership and after the Member's re-

employment and the period of absence between the prior period of membership and the Member's reemployment will not be considered a break in the consecutive period of Pensionable Earnings;

- C. the provisions of the Plan in effect at the Member's subsequent Termination of Membership Date, and;
- D. the benefit determined under subsections A, B, and C above will be actuarially adjusted to reflect any pension payments received as a Retired Member prior to the Member's re-employment.
- (iii) the pension benefit determined under Section 7.02(1)(b) based on the Member's Pensionable Service, Highest Average Pensionable Earnings and provisions of the Plan as determined in subsection (ii) above.

3.04 Re-employment Following Grievance

If a person who is neither a Former Member nor a Retired Member but who was previously a Member, has their employment as a Full-time Employee reinstated as a result of a grievance or other legal proceeding, the Member shall recommence contributions under Article 4 as of the effective date of the reinstatement and, following any determination of a past service pension adjustment under the Income Tax Act, shall make contributions on any retroactive salary or wages received as a result of the reinstatement of employment and the Member may either:

 elect to repay any Commuted Value plus any Excess Contributions or contribution refund, with Credited Interest from the date of payment to the date of repayment in which case,

- (a) the Member's Pensionable Service to which the repaid amounts are applicable will be reinstated;
- (b) the benefit to which the Member will be entitled at the Member's subsequent Termination of Membership Date will be determined without regard to the interruption in the Member's employment; and
- (c) if the amount to be repaid under this Section 3.04 is in respect of pre-1992 Pensionable Service, the amount being repaid must be transferred directly from another registered pension plan on behalf of the Member or from a registered retirement savings arrangement of which the Member is the annuitant;

or,

(2) elect not to repay the Commuted Value plus any Excess Contributions received in which case the Member will be considered a new Employee for all purposes of the Plan effective as of the date of the reinstatement of employment.

ARTICLE 4

CONTRIBUTIONS

4.01 Member Current Service Contributions

- (1) Every Member shall contribute to the Trust Fund in accordance with the rates detailed in Appendix C in respect of the applicable periods.
- (2) The contributions of a Member in a calendar year may not exceed the least of:
 - (a) the contribution in respect of the Member's Contributory Earnings for any year that would result in a pension accrual for the year for the Member equal to the defined benefit limit for the year, as defined in the Income Tax Act, if the following conditions were met:
 - the Member's Highest Average Pensionable Earnings at the end of the year were equal to the Member's Contributory Earnings for the year;
 - (ii) the Average YMPE, at the end of the year, were equal to the YMPE for the year; and
 - (iii) the Member had attained age 65 at the end of the year;
 - (b) the maximum amount that may be permitted for registered pension plans under the provisions of the Act, the Income Tax Act or any other applicable legislation as may be in force from time to time.

4.02 <u>Contributions During Periods of Paid Leave of Absence, Reduced Pay or Disability</u>

(1) Periods of Paid Leave of Absence

Each Member who is absent on an authorized leave of absence with full pay for any reason where the Member's Employer has authorized such leave shall continue to contribute to the Trust Fund in respect of such absence in accordance with Section 4.01 as applicable, based on the Member's Contributory Earnings.

(2) Periods of Reduced Pay and Paid Sick Leave

Each Member who is on a Period of Reduced Pay or a period of paid sick leave at less than full pay shall contribute to the Trust Fund in respect of the period in accordance with Section 4.01 as applicable, based on

- (a) actual Contributory Earnings received during the period, plus the difference between the actual Contributory Earnings received during the period and Deemed Contributory Earnings for the period, provided in the event the Member is on a Period of Reduced Pay the Member and the Employer have agreed to the inclusion of such difference,
- (b) In the absence of an agreement regarding Reduced Pay Periods as provided above, a Member may elect to include the difference between actual Contributory Earnings and Deemed Contributory Earnings by paying twice the amount of contributions determined in accordance with Section 4.01 in respect of such difference; and,
- (c) the YMPE as in effect for the period during which contributions are made.

(3) Periods of Disability or Partial WSIB Benefits

- (a) Each member who is determined to be Disabled and is accruing Pensionable Service under Section 11.01(1) will cease contributions under Section 4.01 during the period of accrual under Section 11.01(1).
- (b) If a Member is receiving partial loss of earnings benefits under workers compensation benefits in respect of an Employer and is receiving Contributory Earnings while in receipt of such benefits:

- (i) the Member will not be required to contribute under Section 4.01 during the first 12 consecutive months during which he or she is in receipt of such benefits in respect of the same work related injury or illness and during which he is accruing Pensionable Service under Section 11.01(2);
- (ii) the Member will be required to contribute under Section4.01 following the expiry of the initial 12 month period in(i) above based on the Contributory Earnings actually being received by the Member.

(4) Limit

Contributions under this Section 4.02 shall be subject to the terms and restrictions in Section 4.04 and Section 5.04 and the Act, the Income Tax Act, and any other applicable legislation.

4.03 Contributions for Periods of Purchasable Service

A Member may elect, prior to attaining their Termination of Employment Date, to include in Pensionable Service any period of Purchasable Service by contributing to the Trust Fund as provided in this Section 4.03 and only the periods so purchased will be included in Pensionable Service. All contributions made pursuant to this Section 4.03 and the periods of Pensionable Service purchased by those contributions shall be subject to the terms and restrictions in Section 4.04 and Section 5.04 and the Act, the Income Tax Act, including but not limited to any past service adjustment certification procedures and any other applicable legislation.

For the purposes of Section 4.03, but excluding Section 4.03(5), Actuarial Value shall mean the higher of (i) Actuarial Value, and (ii) twice the Member's rate of contributions under Section 4.01 at the date the purchase request is received for the period of service being purchased based on the Member's rate of Contributory Earnings and the YMPE at the time the purchase request is received.

(1) Periods of Service with an Employer Prior to Becoming a Member

(a) Each Member may elect to contribute to the Trust Fund in respect of periods of employment with an Employer before the Member became a contributor under Section 4.01. The contribution required to purchase the prior period of employment is equal to 100% of the Actuarial Value of the service being purchased and shall be payable to the Trust Fund in a lump sum. If the Actuarial Value is not paid to the Trust Fund within three months of the date the Member is advised of the amount of the Actuarial Value, a new request to purchase will be required and the Actuarial Value will be re-determined as of the date the new request to purchase is received.

(2) Periods of Parental, Pregnancy or Emergency Leave

- (a) Every Member who is absent on a statutory leave under employment standards legislation applicable to the Member and which requires such Member to be allowed to continue participation in the Plan for the duration of such leave from an Employer, shall, unless the Member specifically elects to have such leave excluded as Pensionable Service, contribute to the Trust Fund in respect of such absence. The Member's contributions will be determined in accordance with Section 4.01 as applicable during the leave and based on the Member's Deemed Contributory Earnings and the YMPE as in effect during the period of absence.
- (b) If the Member elected to exclude this leave as Pensionable Service and subsequently elects, within six months after the end of the leave that this leave be included as Pensionable Service, the Member's contribution will be an amount equal to the amount of contributions determined in accordance Section 4.03(2)(a) on

Deemed Contributory Earnings and shall be payable to the Trust Fund in a lump sum within six months after the end of the leave.

(c) A Member who did not make an election under (b) above or did make an election but did not make the required payment within six months after the end of the absence, may elect to have such leave included as Pensionable Service if the Member pays 100% of Actuarial Value of the service being purchased, which shall be payable to the Trust Fund in a lump sum within three months of the date the Member is advised of the amount of the Actuarial Value. If payment is not so made, a new request to purchase will be required in order to have such absence included as Pensionable Service and the Actuarial Value will be re-determined as of the date the new request to purchase is received.

(3) Periods of Leave of Absence Without Pay or Layoff

- (a) If a Member is
 - (i) on a leave of absence without pay for any reason where the Member's Employer has authorized such leave, as provided in Section 2.39(1), or
 - (ii) is absent on a period of layoff as provided in Section2.39(2) during which recall rights exist

the Member may elect to establish as Pensionable Service the period of such absence that constitutes Continuous Service by contributing to the Trust Fund in accordance with this Section.

(iii) If the Member so elects prior to six months after the end of the absence, the Member's contribution will be an amount equal to twice the amount of contributions determined in accordance with Section 4.01 for the period being purchased based on the Member's Deemed Contributory Earnings and the YMPE as in effect during the period of absence and the contribution shall be payable to the Trust Fund in a lump sum within six months after the end of the leave or layoff.

(b) A Member who did not make an election under (b) above or did make an election but did not make the required payment within six months after the end of the absence, may elect to have such leave included as Pensionable Service if the Member pays the full Actuarial Value of the service being purchased, which shall be payable to the Trust Fund in a lump sum within three months of the date the Member is advised of the amount of the Actuarial Value. If payment is not so made, a new request to purchase will be required in order to have such absence included as Pensionable Service and the Actuarial Value will be re-determined as of the date the new request to purchase is received.

(4) Periods of Work Stoppage

- (a) If a Member of a bargaining unit is absent from work due to work stoppage and provided the Member has not attained their Termination of Employment Date, the Member may elect following cessation of the work stoppage to establish as Pensionable Service the period of such absence that constitutes Continuous Service by contributing to the Trust Fund in accordance with this Section.
- (b) If the Member so elects prior to six months after the end of the work stoppage, the Member's contribution will be an amount equal to twice the amount of contributions determined in accordance with Section 4.01 as applicable during the absence and based on the Member's Deemed Contributory Earnings and the YMPE as in effect for the period of absence and the contribution shall be

- payable to the Trust Fund in a lump sum within six months after the end of the work stoppage.
- (c) A Member who did not make an election under subsection (b) above or did make an election but did not make the required payment within six months after the end of the absence, may elect to have such leave included as Pensionable Service if the Member pays the full Actuarial Value of the service being purchased, which shall be payable to the Trust Fund in a lump sum within three months of the date the Member is advised of the amount of the Actuarial Value. If payment is not so made, a new request to purchase will be required in order to have such absence included as Pensionable Service and the Actuarial Value will be re-determined as of the date the new request to purchase is received.

(5) Prior Service With An Employer, Predecessor Employer or Non-Participating Employer

A Member may elect to include in Pensionable Service prior periods of employment with (i) an Employer as provided in Section 2.39(5), (ii) a Predecessor Employer, or (iii) a Non-Participating Employer, if the Member pays the full Actuarial Value of the service being purchased, which shall be payable to the Trust Fund in a lump sum within three months of the date the Member is advised of the amount of the Actuarial Value. If payment is not so made, a new request to purchase will be required in order to have such prior periods of applicable employment with an Employer, Predecessor Employer, or Non-Participating Employer included as Pensionable Service and the Actuarial Value will be redetermined as of the date the new request to purchase is received. Provided however that if the period of Pensionable Service to be purchased occurred before 1992, then the payment must be transferred directly from another registered pension plan on behalf of the Member and

shall represent the Member's entitlement in that plan for the period that is being purchased.

4.04 Payment of Member Contributions

- (1) The contributions required to be made by a Member under Section 4.01 and 4.02 in any pay period shall be deducted from the Member's Contributory Earnings in the pay period. Any amounts payable under Section 4.03 shall be payable by the Member within the periods prescribed in that Section.
- (2) Notwithstanding any provisions of the Plan, in no event will a Member contribute for periods of absence without pay or the unpaid portion of Periods of Reduced Pay in excess of five years of full-time equivalent remuneration, plus three years of full-time equivalent remuneration for periods of parenting, throughout such Member's period of employment with any Employer after December 31, 1990, in accordance with the Act and the Income Tax Act. Contributions made during a calendar year for any prior period of service shall be subject to the Income Tax Act, including but not limited to the prescribed compensation rules and limits, and the past service pension adjustment procedures and rules.

4.05 **Employer Contributions**

In order to fund the benefits under the Plan, each Employer listed in Appendix A, subject to Section 4.07, shall contribute to the Trust Fund an amount equal to the contributions made by Members who are employed by that Employer that are required by the following Sections:

- (1) Section 4.01;
- (2) Section 4.02, but excluding Section 4.02(2)(c); and
- (3) Section 4.03(2)(a) or (b), whichever is applicable.

4.06 Remittance of Contributions

The contributions made by a Member under the Plan together with the amount required to be paid by the Employer in respect of such contributions shall be paid by the Employer to the financial institution designated by the Board of Trustees so that they shall be received by that financial institution on a business day that is no more than thirty days following the month in respect of which the contributions were made.

When an Employer fails to pay to the Board of Trustees the contributions within the time specified above, there shall be charged to the Employer and deposited to the Trust Fund one and one-half per cent of the amount of the unpaid contributions on the first day of each subsequent month, prorated for partial months, until the total amount due has been paid.

Notwithstanding other provisions of this Section 4.06, all Member contributions shall be remitted by the Employers to the Board of Trustees in accordance with the Act and the Income Tax Act.

4.07 Overpayments

Any overpayment of a contribution by an Employer or a Member in accordance with this Article 4 may be refunded to the Employer or the Member, as the case may be, provided it is in accordance with applicable legislation.

4.08 Deficiencies or Surplus While Plan is Ongoing

Employer and Member current service contributions made in accordance with Section 4.01 and Section 4.05 shall be inclusive of contributions required to be made in respect of any going concern unfunded liability identified by the Actuary in a valuation filed pursuant to the Act. However, and notwithstanding any other terms of the Plan, in the event a going concern unfunded liability is identified by the Actuary in a valuation filed pursuant to the Act and the contributions made in

accordance with Section 4.01 and Section 4.05 are not sufficient to cover contributions required to be made in respect of such going concern unfunded liability, then in the absence of an amendment to Appendix C by the Sponsors' Committee, the contributions required to be made under Section 4.01 and Section 4.05 shall be increased accordingly, provided any such increase in contributions is borne equally by Members and Employers. Contributions shall not be required in respect of any solvency deficiency identified by the Actuary in a valuation filed pursuant to the Act. Any such contributions to the Plan shall not exceed the maximum contributions permitted, under the Act and the Income Tax Act. While the Plan remains ongoing any funding excess shall be applied in the manner determined by the Sponsors' Committee, subject to the provisions of Article 14.

ARTICLE 5

PENSIONABLE SERVICE

<u>5.01</u> Pensionable Service

The Pensionable Service of a Member means any period of Continuous Service in respect of which contributions were made by or on behalf of the Member including:

- (1) any period of Continuous Service during which the Member made contributions to the Plan under Section 4.01 or 4.02;
- (2) any prior period of Purchasable Service with an Employer, Predecessor Employer or Non-Participating Employer in respect of which the Member made contributions to the Trust Fund under Article 4.
- (3) periods of Transferred In Service for which a Member made contributions or a transfer of funds was made on behalf of the Member and determined to constitute Pensionable Service under Article 13;
- (4) any period during which the Member meets the definition of Disabled in accordance with Section 2.15, determined pursuant to the provisions of Article 11 to constitute Pensionable Service.

For greater certainty, in respect of any Member who is working less than full-time, the applicable period of pensionable service will be pro-rated accordingly. For greater certainty, in no event, shall any period in respect of which contributions were made by or on behalf of a Member under the Design Two Provisions be included in the Pensionable Service of that Member for the purposes of calculating the amount of their benefit under the Design One Provisions.

5.02 Computation of Pensionable Service

Pensionable Service will be computed in accordance with (1), (2), and (3) below, subject to the limitations in Section 5.03 and Section 5.04 below:

- (1) Pensionable Service will be computed in years and part years;
- (2) in no event will a Member accrue more than the equivalent of 1 year of Pensionable Service in any one calendar year;
- (3) Such other policies as may be adopted by the Board of Trustees from time to time. In no event shall Pensionable Service include any period prior to an earlier Termination of Membership Date or prior to employment with an Employer, on or after the Effective Date of the Plan, except as specifically provided in the Plan.

<u>5.03</u> Pensionable Service for Periods of Concurrent Employment

(1) Full-time Employment with Two Employers

If a Member is a Full-time Employee with one Employer (the "Original Employer") and is subsequently employed on a full-time basis by another Employer (the "Subsequent Employer"), no Contributions will be made and no Pensionable Earnings and Pensionable Service will be recognized in respect of employment with the Subsequent Employer as long as the Employee continues to be recognized as a Full-time Employee with the Original Employer.

(2) Full-time and Other Than Regular Full-time and Design Two Employment

If a Member is a Full-time Employee with one Employer (the "Full-time Employer") and is concurrently employed by another Employer as an OTRFT Employee (an "OTRFT Employer"), or a Design Two Employee (a "Design Two Employer"), that Member shall only accrue benefits under the Design One Provisions and no Contributions will be made and no

Pensionable Earnings and Pensionable Service will be recognized in respect of employment with any OTRFT Employer or Design Two Employer as long as the Employee continues to be recognized as a Full-time Employee with the Full-time Employer.

(3) Maximum Pensionable Service

In no event will an Employee with concurrent employment with more than one Employer in a calendar year accrue in aggregate more than the equivalent of 12 months of Pensionable Service in respect of that calendar year.

5.04 Limitations on Pensionable Service

- (1) Notwithstanding any other provisions of the Plan, in no event will a Member receive Pensionable Service during periods of absence without pay, other than a period in which the Member is Disabled, or the unpaid portion of Periods with Reduced Pay in excess of five years of full-time equivalent remuneration, plus three years of full-time equivalent remuneration for periods of parenting, throughout the Member's periods of employment with any Employer after December 31, 1990 in accordance with the Act and the Income Tax Act, including, but not limited to the rules and restrictions regarding prescribed compensation.
- (2) A person may not accrue Pensionable Service and concurrently receive a pension under the Plan payable from the Trust Fund in respect of their own Pensionable Service.

ARTICLE 6

NORMAL RETIREMENT

6.01 Normal Retirement Pension

Each Member who has attained their Termination of Employment Date and who has attained their Termination of Membership Date coincident with their Normal Retirement Date will be entitled to receive from the Trust Fund an annual pension benefit, as of their Normal Retirement Date, computed as in (1), (2) and (3) below and payable in accordance with the provisions of Article 9:

- (1) 1.3% of the Member's Highest Average Pensionable Earnings up to the Average YMPE, multiplied by the Member's years, including part years, of Pensionable Service excluding any full calendar year where a Member
 - (a) for periods of service prior to January 1, 2013, did not contribute to the Canada Pension Plan as provided under Section 4.01(2) or by reason of having attained age 65; or
 - (b) for periods of service prior to January 1, 2013, was a Member of the Plan at December 31, 2004 and at that date had attained 35 years of Pensionable Service; or
 - (c) is accruing Pensionable Service pursuant to Section 11.01 in respect of a period in which the Member was Disabled that commenced prior to July 1, 2016, plus
- (2) 2% of the Member's Highest Average Pensionable Earnings up to the Average YMPE multiplied by the Member's years, including part years, of Pensionable Service that were not included in (1) above; plus
- (3) 2% of the Member's Highest Average Pensionable Earnings in excess of the Average YMPE multiplied by the Member's years, including part years, of Pensionable Service.

Formatted: Highlight

Formatted: Highlight

Formatted: Highlight

Formatted: Highlight

6.02 Maximum Pension Benefit

Notwithstanding any other provisions of the Plan, the amount of lifetime pension benefits provided under the Design One Provisions, in respect of any Member at Termination of Membership Date or termination of the Plan, or otherwise, including any amounts payable to the Member's former Spouse upon marriage breakdown, as the case may be, when added to the amount of pension benefits provided to the Member in respect of a period of employment that constitutes Pensionable Service under this Plan by any other registered pension plan to which the Employer contributes or has contributed will, under no circumstances, exceed a pension benefit that at an annual rate is the lesser of:

- the defined benefit limit for the year of commencement, as defined in the Income Tax Act, multiplied by the Member's years of pensionable service with the Employer;
- (2) 2% of the average of the Member's best three consecutive years of compensation as defined in the Income Tax Act, multiplied by the Member's years of pensionable service with the Employer;

or such other maximum limitation as may be prescribed by the Income Tax Act.

For the purposes of this Section 6.02, a Member shall not be credited with more than 35 years of pensionable service for service prior to January 1, 1992, and the terms "defined benefit limit" and "pensionable service" shall have the meanings ascribed to them under the Income Tax Act and "pensionable service" shall refer to Pensionable Service under this Plan.

Notwithstanding the foregoing, if a Member elects to contribute after June 7, 1990 to the Plan in respect of pre-1990 service, the amount of lifetime retirement benefits that can be provided for each such year, subject to the exemptions provided in the Income Tax Act, is limited to the greater of:

(1) \$1,150.00; and

(2) 2/3 of the defined benefit limit for the year in which the lifetime retirement benefits commence to be paid.

6.03 Pension Adjustment

Notwithstanding any provisions of the Plan, the annual benefit accrual rate under the Plan shall not exceed the maximum permissible pension adjustment rate as defined in the Income Tax Act and the benefit formula shall not exceed the maximum benefit formula for registered pension plans as permitted under and in accordance with the Income Tax Act.

6.04 Reduction of Benefits

Notwithstanding any other provisions of the Plan, benefits under the Plan may be reduced or contributions made in accordance with the Plan may be refunded, as appropriate, if such benefits or contributions exceed the maximum limits as prescribed under the Income Tax Act in order to avoid revocation of the Plan, subject to the Act and with the consent of the Financial Services Regulatory Authority of Ontario.

ARTICLE 7

EARLY RETIREMENT

7.01 Early Retirement Date

- (1) A Member who has attained their Termination of Employment Date prior to Normal Retirement Date and who coincident with or prior to their Termination of Membership Date has attained age 50 and 20 years of Pensionable Service;
- (2) age 55;
- (3) age plus years of Pensionable Service total 85 years;

may elect an Early Retirement Pension as of the Member's Early Retirement Date, being the last day of the any month coincident with or following the attainment of the conditions set out in (1), (2) or (3) above, subject to the provisions of this Article 7. For the purposes of Article 7, age is calculated in years and part years on the last day of the month in which a determination is required.

7.02 Early Retirement Pension

(1) Determination of Early Retirement Pension

Each Member who has elected an Early Retirement Date, subject to the provisions of this Article 7, will be entitled to receive from the Trust Fund an annual pension benefit and payable in accordance with the provisions of Article 9, computed as the sum of (a) and (b) below:

(a) an annual lifetime pension computed pursuant to Section 6.01 based on the Member's Pensionable Service, Highest Average Pensionable Earnings and Average YMPE as of the Member's Termination of Employment Date; plus (b) subject to the provisions of Section 9.06(1)(b), an annual bridge pension benefit in an amount equal to 0.7% of the Member's Highest Average Pensionable Earnings up to the Average YMPE multiplied by the Member's years, including part years, of Pensionable Service as are included in the determination under Section 6.01(1),

and the sum of the amounts determined in (a) and (b) above will be reduced by three per cent for each full year and prorated for partial years that the Member's age at their Early Retirement Date precedes the Member's age, had the Member's Continuous Service continued uninterrupted, at the last day of the month in which the Member would attain the earliest of the conditions in (c), (d) and (e) below:

- (c) Normal Retirement Date;
- (d) attained age plus years of Pensionable Service totals 85 years; and
- (e) age 60 and 20 years of Pensionable Service.

(2) Pension Payments

In the event that:

- (a) the application to commence an Early Retirement Pension is not received by the Plan Manager within one month following the later of the Member's Termination of Employment Date and the date on which any condition set out in (1), (2) or (3) of Section 7.01 is attained co-incident with or following a Member's Termination of Employment Date, then the Member's Early Retirement Date shall be the end of the month in which the application is received.
- (b) if the application is received more than 12 months after the Member's Termination of Membership Date, the person is entitled to a Deferred Pension determined pursuant to Article 10 and the provisions of Section 10.03 applicable to early commencement of Deferred Pension will apply.

7.03 Unreduced Early Retirement Pension

Every Member who, at the Member's Early Retirement Date has satisfied the conditions in either 7.02(1)(d) or (e), is entitled to an Early Retirement Pension determined in accordance with Section 7.02(1) without any reduction for commencement prior to Normal Retirement Date, provided the Member's application to commence their Early Retirement Pension is received within 12 months of the Member's Termination of Membership Date.

7.04 Maximum Pension Benefit

The Member's pension benefit payable on early retirement will be subject to the provisions of Section 6.02 and any applicable maximum permitted benefit provisions of the Income Tax Act, provided further that notwithstanding any other provision of the Plan, for pensionable service from January 1, 1992, the sum of the annual retirement benefit payable to a Member under the Design One Provisions and any other registered pension plan in which the Employer participates in respect of the Member (other than the Canada or Quebec Pension Plan) shall not, in the year of commencement, prior to age 65, exceed the sum of:

- the defined benefit limit for the year of commencement multiplied by the Member's pensionable service with the Employer after December 31, 1991; plus
- (2) 25% of the average of the YMPE for the year of commencement and the two immediately preceding years multiplied by a fraction the numerator of which is the Member's Pensionable Service (not exceeding 35 years) accrued after December 31, 1991 and the denominator of which is 35

and the sum so determined will be reduced by three per cent for each full year and prorated for partial years that the Member's age at pension commencement date precedes the Member's age, had the Member's Continuous Service continued uninterrupted, at the last day of the month in which the Member would attain the

earliest of the conditions specified in Section 8503(3)(c) of the Regulations of the Income Tax Act.

For the purposes of this Section 7.04, the terms "defined benefit limit" and "pensionable service" shall have the meanings ascribed to them under the Income Tax Act and "pensionable service" shall refer to Pensionable Service under this Plan.

ARTICLE 8

POSTPONED RETIREMENT

8.01 Postponed Retirement Date

If a Member has not attained their Termination of Employment Date coincident with or prior to their Normal Retirement Date, pension benefits under the Plan will not commence until the Member's Postponed Retirement Date which will be when the earlier of (1) and (2) below occurs

- (1) November 30th of the year in which the Member attains age 71, or such other age as is specified in the Income tax Act (Canada), and
- (2) the last day of the month in which the Member's Termination of Employment Date occurs,

and the Member will continue to make contributions under Section 4.01 until the month in which the Member's Postponed Retirement Date occurs.

8.02 Postponed Retirement Pension

Each Member who attains a Postponed Retirement Date under Section 8.01 will be entitled to receive from the Trust Fund an annual pension benefit, computed pursuant to Section 6.01 based on the Member's Pensionable Service, Highest Average Pensionable Earnings and Average YMPE as of the Member's Postponed Retirement Date, payable in accordance with the provisions of Article 9.

The Member's Postponed Retirement Pension under this Section 8.02 will be subject to the provisions of Section 6.02.

ARTICLE 9

PAYMENT OF PENSIONS

9.01 Normal Form of Pension Benefit

The normal form of pension payable under the Plan in respect of a Member, Former Member or Retired Member's entitlement under Articles 6, 7, 8 or 10 shall be a pension payable in equal monthly instalments for the person's lifetime and shall be payable in that form unless the alternate normal form under Section 9.02 applies or the person elects an optional form under Section 9.03.

9.02 Alternate Normal Form of Pension Benefit

(1) Spouse at Pension Commencement

If a Member, Former Member or Retired Member has a Spouse at the date the first pension payment is due pursuant to Article 6, 7, 8 or 10, in lieu of the normal form provided under Section 9.01, the pension shall be paid in an alternate normal form as a joint and survivor annuity.

The pension benefit so payable for their lifetime will be the benefit otherwise payable under Section 9.01 and on their death, 60% of the lifetime benefit he or she was receiving or ought to have been receiving on the date of death will be payable to their Spouse for the Spouse's lifetime.

(2) Spouse at Date of Death

(a) If the Member, Former Member or Retired Member did not have a Spouse at the date their first pension payment was due or if the Member, Former Member or Retired Member is predeceased by the Spouse as of the date the Member, Former Member or Retired Member's first pension payment was due or if such Spouse has ceased to have an entitlement pursuant to the Act and the Member, Former Member or Retired Member is survived by a subsequent Spouse, then 60% of the benefit the Member, Former Member, or Retired Member was receiving on the date of death will be payable to the Member, Former Member or Retired Member's Spouse at the date of death for the Spouse's lifetime.

(3) In the event legislation governing benefits provides for a minimum pension payable to the Member, Former Member or Retired Member's Spouse that is greater than 60% of the lifetime benefit payable to the Member, Former Member or Retired Member then this joint and survivor annuity will be the Actuarial Equivalent of the joint and survivor annuity otherwise payable under Section 9.02(1).

9.03 Optional Form of Pension Benefit

- (1) If a Member, Former Member or Retired Member has a Spouse at the date their first pension payment is due pursuant to Article 6, 7, 8 or 10, in lieu of the normal form provided under Section 9.02, the Member, Former Member or Retired Member may elect prior to the commencement of pension payments to receive pension benefits in the form of a joint and survivor annuity providing for a survivor pension to the Member, Former Member or Retired Member's Spouse of 75% of the lifetime benefit payable to the Member, Former Member or Retired Member and this optional joint and survivor annuity will be the Actuarial Equivalent of the joint and survivor annuity otherwise payable to the Member, Former Member or Retired Member and the Spouse under Section 9.02(1).
- (2) If the Spouse in respect of which the election under this Section 9.03 is made predeceases the Member, Former Member or Retired Member or otherwise ceases to have an entitlement pursuant to the Act and the Member, Former Member or Retired Member is survived by a subsequent Spouse, the benefit payable to that subsequent Spouse will be 60% of the lifetime benefit to which the Member would have been entitled at the date

of the Member, Former Member or Retired Member's death if the election in subsection (1) above had not been made.

9.04 Surviving Child or Children

- (1) If a Retired Member in receipt of a pension dies and is not survived by a Spouse, or if a Spouse in receipt of a survivor pension under Section 9.02 or Section 9.03 dies, and the Retired Member or Spouse is survived by an eligible Child or Children, the survivor pension otherwise payable to a Spouse under Section 9.02 or Section 9.03, whichever is applicable at the date of death, will be divided among and paid to the eligible Child or Children for so long as the Child or Children meet the applicable definition of Child in accordance with Section 2.08.
- (2) If two or more Children are entitled to a survivor benefit under subsection (1) above, the survivor benefit otherwise payable to a Spouse will be apportioned equally among the Children and it will be reapportioned on the death of a Child or as each of them no longer meets the applicable definition of Child in accordance with Section 2.08. For greater clarity, once a Child no longer meets the applicable definition of Child in accordance with Section 2.08, they will have no further entitlement to a survivor benefit under this Section 9.04.
- (3) Notwithstanding the provisions of this Section 9.04 and effective January 1, 2018, the survivor benefit payable to an eligible Child or Children under subsection (1) or (2) may be reduced as appropriate, if the benefit exceeds the maximum limit prescribed under paragraph 8503(2)(d) of the Regulations of the Income Tax Act. For greater certainty, in no event shall a benefit payable to an eligible Child or Children exceed the limit set out in paragraph 8503(2)(d) of the Regulations of the Income Tax Act.

9.05 Minimum Guarantee

Where a Retired Member who is in receipt of a pension dies leaving no Spouse nor Children or where a pension has become payable to the Spouse or Children of a Retired Member and on the cessation of the entitlement of such Spouse or Children no further pension is payable in respect of the Retired Member, the amount, if any, by which the sum of 60 months of the Member's lifetime pension benefit as calculated at the time the Member commenced receiving the pension exceeds the total of the benefits paid to or in respect of the Retired Member, Spouse and Children shall be refunded to the Beneficiary of the last to die of the Retired Member and the Spouse.

9.06 Payment of Pensions

(1) Commencement and Duration of Pension Payments

Unless otherwise provided in the Plan, a pension benefit provided by the Plan is payable in equal monthly instalments no later than the first banking day of each calendar month

- (a) commencing no later than the first banking day of the month following:
 - (i) in the case of a Retired Member who has elected in accordance with the Plan to receive a Normal Retirement Pension, Early Retirement Pension or Postponed Pension, the Member's Normal, Early or Postponed Retirement Date, as applicable, or such other date as a Former Member may elect under Section 10.03; and
 - (ii) in the case of a Spouse or Child, the month immediately following the month in which the Retired Member's or Spouse's death, whichever is applicable, occurs; and
- (b) ceasing with the last monthly instalment payable prior to

- in the case of a Retired Member or Spouse, the death of the Retired Member or Spouse, whichever is applicable; and
- (ii) in the case of each Child, the date the Child attains the age of 18 years, or if earlier, the date the Child's death occurs;

provided that any benefit payable in respect of a Retired Member's entitlement under Section 7.02(1)(b) will cease with the last monthly instalment prior to the earlier of the Retired Member's death and the date the Retired Member attains age 65.

(2) Late Payment of Pension Benefits

In the event that a person becomes entitled to receive a monthly pension from the Trust Fund and pension payments do not commence within two months of the date on which the person is entitled to commence receiving pension payments, such person shall only be entitled to Credited Interest on the late payment or such other amount as determined under the applicable policy and procedures as determined by the Plan Manager. No person shall be entitled to receive liquidated damages in respect of any pension payment which is overdue because of a delay which is directly or indirectly caused by the act, neglect, default or omission of that person.

(3) Cessation of Pension Following Re-employment

If a Retired Member is re-employed after commencement of pension payments and following re-employment again becomes a Member and resumes contributions pursuant to Section 3.03(2), the re-employed Retired Member's pension shall be suspended during the period of re-employment but not beyond the end of the year in which the Employee reaches age 71, or such other age as is specified in the Income Tax Act. A person shall not concurrently receive a pension payment from the Plan in respect of their own Pensionable Service and accrue pension benefits under the Plan.

9.07 Excess Contributions

Where a Member of the Plan, or in the case of the death of the Member, the Member's Spouse or Beneficiary, becomes entitled to a benefit as a result of attaining their Termination of Membership Date, and the Member has Excess Contributions, the Member, or in case of death, the Member's Spouse or Beneficiary, as applicable, shall receive a refund of the Excess Contributions, determined at the Member's Termination of Membership Date, with Credited Interest to the date of refund, and applicable recipient may elect that the refund be transferred to a Registered Retirement Savings Arrangement, subject to the Act and the Income Tax Act. However, in the event the Plan is not advised within two months of the Excess Contribution becoming payable as to how the entitlement is to be distributed, the amount may be paid to the applicable recipient less applicable withholding tax.

9.08 Shortened Life Expectancy

The Plan Manager, in their discretion, may permit the commutation and payment, on a non-locked in basis, of a benefit payable to a person entitled to a benefit under the Plan by reason of an illness or physical disability of the person that will shorten such person's life expectancy to twenty-four months or less, subject to the Act and the Income Tax Act. The following conditions shall apply in such circumstances:

- (1) In the case of a Member, the Member shall, in writing, resign membership in the Plan. The Member's benefit under the Plan shall be calculated as though the Member attained their Termination of Membership Date as of the effective date of the Member's resignation under the Plan.
- (2) The person's illness or physical disability shall be certified by a physician who is licensed to practice medicine in a jurisdiction in Canada.
- (3) If the person has a Spouse on the date the benefit is to be commuted, the Spouse shall provide a written, signed waiver:

- (a) consenting to the commutation of the person's benefit;
- (b) agreeing that no further benefit shall be payable from the Plan; and
- (c) certifying that the Spouse has received independent legal advice regarding the provisions of this Section 9.08.
- (4) If the person whose benefit is commuted pursuant to this Section 9.08 recovers from such disability and once again becomes a Member of the Plan,
 - (a) the Member shall be treated for all purposes of the Plan as a new Member and no years of Pensionable Service shall be reinstated; and,
 - (b) notwithstanding any other provision of the Plan, the Member shall not be permitted to make contributions to the Plan for any prior years of Pensionable Service in respect of which their benefit was commuted pursuant to this Section 9.08.

9.09 Small Pensions

In the event that the annual pension payable under this Plan to a Former Member or Retired Member at their Normal Retirement Date is less than 4% of the YMPE at the Former Member's or Retired Member's Termination of Employment Date or if the Commuted Value of the pension is less than 20% of the YMPE at the Former Member's or Retired Member's Termination of Employment Date, or is less than such other threshold as may be permitted in accordance with the Act or the Income Tax Act for the payment of lump sum amounts, the Member will receive a lump sum payment, unless otherwise determined by the Plan Manager, in the amount of the Commuted Value of the pension to which the Former Member or Retired Member is entitled. The Former Member or Retired Member may elect to transfer such amount to a Registered Retirement Savings

Arrangement. In the absence of such an election, the amount may be paid less

applicable withholding tax. A transfer or payment in accordance with this section shall fully discharge all liability with respect to the Former Member's or Retired Member's pension.

9.10 Small Spousal Pensions

In the event that the alternate normal pension payable under this Plan is less than 4% of the YMPE at the Member's date of death or if the Commuted Value of the benefit is less than 20% of the YMPE at the Member's date of death, or is less than such other threshold as may be permitted in accordance with the Act or the Income Tax Act from time to time for the payment of lump sum amounts, the Spouse will receive a lump sum payment, unless otherwise determined by the Plan Manager, and subject to the Act, in the amount of the Commuted Value of the pension to which the Spouse is entitled. The Spouse may elect to transfer such amount to a Registered Retirement Savings Arrangement. In the absence of such an election, the amount may be paid less applicable withholding tax. A transfer or payment in accordance with this section shall fully discharge all liability with respect to the alternate normal pension.

BENEFITS ON TERMINATION OF EMPLOYMENT

10.01 Termination with Deferred Pension

If a Member attains their Termination of Membership Date and is not then entitled to make an election under Section 7.01 to commence an Early Retirement Pension and has not attained their Normal Retirement Date, the Member will be entitled to receive a Deferred Pension in the amount accrued to the Member under Section 6.01 using the Member's Pensionable Service, Highest Average Pensionable Earnings and Average YMPE at the Member's Termination of Employment Date and shall be payable at the Member's Normal Retirement Date in accordance with Article 9 or such other date as the Member may elect under Section 10.03.

10.02 Portability Options

- (1) Subject to the provisions of (2) below:
 - (a) a Member who has attained their Termination of Employment Date but has not attained their Termination of Membership Date, provided the Member is not then entitled to make an election under Section 7.01 to commence and Early Retirement Pension and has not attained their Normal Retirement Date, may elect, in lieu of any other benefit under the Plan, to have the Commuted Value of their entitlement under the Plan transferred, on a locked-in basis, to the pension fund related to another registered pension plan within the meaning of the Income Tax Act, in order to provide benefits in respect of the Member under a defined benefit provision or a money purchase provision of that other plan, provided the administrator of the other pension plan agrees to accept the payment, or

a Former Member, provided the Former Member is not then (b) entitled to make an election under Section 7.01 to commence an Early Retirement Pension and has not attained their Normal Retirement Date, may elect, in lieu of any other benefit under the Plan, to have the Commuted Value of their Deferred Pension transferred, on a locked-in basis, to a Registered Retirement Savings Arrangement as prescribed in the regulations to the Act in the name of the Former Member or transferred, on a locked-in basis, to the pension fund related to another registered pension plan within the meaning of the Income Tax Act, in order to provide benefits in respect of the Member under a defined benefit provision or a money purchase provision of that other plan, provided the administrator of the other pension plan agrees to accept the payment, provided that, if a Former Member does not elect the option of a Commuted Value transfer under this Section 10.02 within six months of their Termination of Membership Date, the Former Member shall no longer be able to exercise any portability under this section 10.02(1)(b).

Notwithstanding the foregoing, a Member, who has attained their Termination of Employment Date but has not attained their Termination of Membership Date, or a Former Member, provided the Member or Former Member has not attained their Normal Retirement Date, may elect a Commuted Value transfer to the pension fund related to another registered pension plan within the meaning of the Income Tax Act, in order to provide benefits in respect of the Former Member, provided the other pension plan is not an Individual Pension Plan within the meaning of the Income Tax Act, and provided the administrator of the other pension plan agrees to accept the payment.

- (2) Any transfer pursuant to Section (1) above:
 - (a) shall be subject to such conditions and restrictions as may be prescribed by the Act and shall be subject to the provisions of the Income Tax Act governing transfers to a defined benefit provision, a money purchase provision, or a Registered Retirement Savings Arrangement; and
 - (b) shall be made as soon as practicable after the Plan Manager receives the Member's transfer option election.

10.03 Early Commencement of Deferred Pension

On or after the attainment of age 55, or attainment of age 50 if the Member had completed 20 years of Pensionable Service at the Former Member's Termination of Employment Date, a Former Member may elect to have payment of the Deferred Pension commence prior to the Former Member's Normal Retirement Date on the first day of any month subsequent to the Former Member's election being received by the Plan Manager.

The Former Member's pension benefit will be determined and payable as follows:

- (1) The Former Member's Deferred Pension in respect of their entitlement under Section 6.01 will be reduced by five per cent for each full year and prorated for partial years that the Former Member's age at the date pension benefits commence is less than 65 years and will be payable in the normal form under Section 9.01 or 9.02, whichever is applicable, or the optional form as may be elected under Section 9.03 and for the periods provided in Section 9.06.
- (2) The Former Member's Deferred Pension in respect of their entitlement under Section 7.02(1)(b) will be reduced by five per cent for each full year and prorated for partial years that the Former Member's age at the date

pension benefits commence is less than 65 years and will be payable for the periods provided in Section 9.06.

ARTICLE 11

DISABILITY

11.01 Accrual of Benefits and Disability Date

- (1) If a Member is determined to be Disabled and has not elected to receive any benefits under Article 10, the Member will cease to make contributions under Section 4.01 and will accrue Pensionable Service from the date that the Plan Manager determines is the date the Member's period of disability commenced, as determined using such evidence the Plan Manager deems appropriate (the Member's "Disability Date") to the earliest of:
 - (a) the Member's Normal Retirement date;
 - (b) an Early Retirement Date if so elected by the Member provided the Member has met the criteria set out in Section 7.01 and such Member will be deemed to have attained their Termination of Membership Date as of such election;
 - (c) the date the Member ceases to be Disabled; and
 - (d) the date of the Member's death.

For purposes of this Section 11.01(1) a Member will cease to be Disabled at the earlier of the date it is determined that the Member no longer satisfies the definition of Disabled. The Member will continue to be considered to be an Employee during the period of accrual of Pensionable Service under this Section 11.01(1). Notwithstanding the foregoing, a Member shall be deemed to no longer be Disabled as of the date the

Member commences any occupation for compensation or profit other than employment associated with a rehabilitation program that is acceptable to the Plan Manager. If a Member ceases to be Disabled and does not recommence active employment with an Employer within two months of ceasing to be Disabled, the Member will be deemed to have attained their Termination of Employment Date for purposes of the Plan as of the date the Member ceases to be Disabled.

- (2) If a Member is receiving partial loss of earnings benefits under the applicable workers compensation benefits in respect of an Employer and the provisions of Section 11.01(1) do not apply, the Member will cease to make contributions under Section 4.01 and will accrue Pensionable Service from the date as of which the Member first becomes entitled to such partial loss of earnings benefits to the earliest of
 - (a) 12 months from the date as of which such partial loss of earnings benefits first commenced, and
 - (b) the date such partial loss of earnings benefits cease.

BENEFITS ON DEATH PRIOR TO PENSION COMMENCEMENT

12.01 Death Prior to Pension Commencement

(1) Spousal Pension

If a Member, Retired Member or Former Member dies before payment of the first instalment of their pension is due and is survived by a Spouse, the Spouse will be entitled to

- (a) an immediate pension, or a deferred pension payable at the Spouse's age 65, which is the Actuarial Equivalent of the benefit accrued to the Member, Retired Member or Former Member to the date of their death, or
- (b) a lump sum payment, transferred to a Registered Retirement Savings Arrangement, or paid less applicable withholding tax, which is the Commuted Value of the benefit accrued to the Member, Retired Member or Former Member's date of death,

as elected by the Spouse on forms prescribed for the purpose by the Plan Manager and filed within the time prescribed by the Act and the Income Tax Act.

Unless the Spouse elects a deferred pension or the lump sum under (b) above within the time prescribed by the Act, the Spouse will receive the immediate pension payable under subsection (a) commencing on the first day of the month following the death of the Member, Retired or Former Member and will be payable to the Spouse until the Spouse dies. Upon the death of the Spouse, the amount, if any, by which the sum of 60 payments payable to the Spouse in the amounts first payable to the Spouse

exceeds the total payments received by the Spouse will be paid to the Spouse's Beneficiary.

If the Spouse elects a deferred pension under subsection (a) and dies before commencement of pension payments, the Commuted Value of the Spouse's pension will be payable to the Spouse's Beneficiary.

(2) Children's Death Benefit

If a Member, Retired Member or Former Member dies before payment of the first instalment of their pension is due and is not survived by a Spouse but is survived by an eligible Child or Children, a pension equal to 50% of the pension accrued to the Member, Former Member or Retired Member to the date of death shall be paid to the eligible Child or Children until the Child no longer meets the applicable definition of Child in accordance with Section 2.08.

If two or more Children are entitled to the pension under this Section 12.01, the pension will be apportioned equally among the Children and it will be reapportioned on the death of a Child or as each of them no longer meets the applicable definition of Child in accordance with Section 2.08. For greater clarity, once a Child no longer meets the applicable definition of Child under Section 2.08, they will have no further entitlement to a Children's Death Benefit under this Section 12.01(2). Provided, however, that no Children's Death Benefit shall be payable in respect of a Member or Former Member if the Jurisdiction of Employment applicable to the Member or Former Member does not permit a Children's Death Benefit to be deducted from the death benefit otherwise payable in respect of the Member or Former Member.

(3) Beneficiary's Death Benefit

If a Member, Retired Member or Former Member dies before payment of the first instalment of their pension is due and is not survived by a Spouse entitled to a benefit under Section 12.01(1), there shall be payable to the Member's Beneficiary a lump sum payment which is the Commuted Value of the benefit accrued to the death of the Member, Former Member or Retired Member less the Commuted Value of any benefits payable pursuant to Section 12.01(2).

12.02 Transfer of Entitlements

A Spouse who elects a lump sum cash settlement under the Plan as a result of a Member, Retired Member or Former Member who dies before payment of the first instalment of their pension is due may direct that such amount be transferred to the pension fund of another pension plan in respect of the Spouse, provided the administrator of the other pension plan agrees to accept the payment, or to a Registered Retirement Savings Arrangement as prescribed in the regulations to the Act in the name of the Spouse, subject to the Act and the Income Tax Act.

12.03 Limitation on Benefits

No death benefit payable pursuant to this Article 12 shall exceed the maximum pre-retirement death benefits under the Income Tax Act.

RECIPROCAL AGREEMENTS

13.01 Reciprocal Agreement

The Board of Trustees may, within their absolute discretion, enter into a Reciprocal Agreement with the person authorized for that purpose under a registered pension plan established by or for any Non-Participating Employer for the transfer of pension credits from that other pension plan by a Member who has ceased membership in that other pension plan to the Trust Fund to purchase Transferred In Service on a reciprocal or other basis, subject to the following provisions:

(1) the Reciprocal Agreement shall contain such provisions as may be prescribed by the Board of Trustees, including the basis of determination of the Transferred In Service to be purchased by the transfer of pension credits and the time limits within which the transfer of pension credits must occur;

(2) the Pensionable Service under this Plan of a Member for whom an amount of money is transferred to the Trust Fund under the Reciprocal Agreement shall include the period of former employment which under the terms of the Reciprocal Agreement will constitute Transferred in Service for purposes of the determination of the Member's benefits under this Plan.

13.02 Conditions of Transfer and Purchase

The transfer of pension credits under Section 13.01 shall be subject to the following provisions:

 the Transferred In Service included in the Member's Pensionable Service will in no event exceed the total period of service with the Non-Participating Employer; Formatted: Highlight

Formatted: Highlight

(2) if the amount transferred to the Trust Fund from the Non-Participating
Employer's pension plan is insufficient to purchase as Transferred In
Service the total period of service with the Non-Participating Employer,
the Member may, subject to Income Tax Act regulations, purchase the
balance of the period of service eligible to be Transferred In Service by
paying to the Trust Fund the Actuarial Value of the balance of the period.

INDEXATION

14.01 Inflation Protection Factor

(1) **Definition of Inflation Protection Factor**

Subject to Section 14.01(2) below, inflation protection factor means 75% of the increase (or decrease, as applicable) in the average Consumer Price Index for Canada, as reported by Statistics Canada, for the 12 month period ending September 30 immediately prior to the effective date of the application of the inflation protection factor, as compared to the average Consumer Price Index for the 12 month period ending September 30 in the prior calendar year. If the above calculation in respect of the 12 month period results in an inflation protection factor in excess of 8%, the inflation protection factor in respect of that period shall be deemed to be 8%. If the above calculation in respect of the 12 month period results in an inflation protection factor which is less than 0%, the inflation protection factor in respect of that period shall be deemed to be 0%.

(2) Application of the Inflation Protection Factor

If the calculation under Section 14.01(1) in respect of the 12 month period results in a positive inflation protection factor in excess of the 8% maximum, the amount in excess of the 8% maximum shall be carried forward and applied to increase the inflation protection factor in the following 12 month period. If the calculation under Section 14.01(1) in respect of the 12 month period results in a negative inflation protection factor, such amount shall be carried forward and applied to decrease the inflation protection factor in the following 12 month period.

14.02 Increases in Pensions in respect of service between 1991 and 2008

Pensions payable to Retired Members and to the surviving Spouse of a deceased Member, Former Member or Retired Member and the annual amount of Deferred Pension payable to a Former Member and the deferred pension payable to a surviving Spouse under Section 12.01, will be increased on an annual basis each January 1 in accordance with Section 14.04 in respect of service attributable to each calendar year after December 31, 1991 and before January 1, 2008.

14.03 Increases in Pensions in respect of service after December 2007

Pensions payable to Retired Members and to the surviving Spouse of a deceased Member, Former Member or Retired Member and the annual amount of Deferred Pension payable to a Former Member and the deferred pension payable to a surviving Spouse under Section 12.01, shall be increased on an annual basis pursuant to Section 14.04 in respect of service attributable to each calendar year after December 31, 2007 but only if the most recent actuarial valuation of the Plan for funding purposes discloses prior unutilized funding excesses or cumulative gains, and in the following order of priority:

- increases in respect of the immediately preceding calendar year,
 followed by,
- (2) increases in respect of previous years where the maximum increase permitted by section 14.04 had not previously been effected in respect of such years, in chronological order commencing with the first calendar year following 2007 in respect of which the maximum increase permitted by Section 14.04 has not previously been effected.

14.04 Application of Increases

Any increases pursuant to Sections 14.02 and 14.03 will be made each January 1 in accordance with the following provisions:

(1) where a monthly pension is payable to a person on December 31 of the prior year in respect of a pension that was being paid on January 1 of the

- prior year, the monthly amount of pension payable to the person on January 1 of the current year shall be increased by the inflation protection factor;
- (2) where a person who would have been entitled to the increase provided for in subsection (1) above dies before the January 1 in which the increase is payable, the monthly amount of pension payable in respect of that deceased person to another person or persons, shall be increased by the inflation protection factor for the current year;
- (3) where a person started receiving a pension during the period from January 1 of the prior year to December 31 of the prior year, the monthly pension payable to the person on January 1 of the current year shall be increased by the inflation protection factor prorated for that portion of the prior year from the commencement of the pension to December 31. This subsection is not applicable to those persons described in subsection (1) and (2).
- (4) Where a person who would have been entitled to the increase provided for in subsection (3) dies before the January 1 on which the inflation protection factor is payable, the monthly pension payable in respect of that deceased person to another person or persons shall be increased on January 1 in such year in the manner provided for in subsection (3), using the pension commencement date of the first such deceased person entitled to a pension.
- (5) The annual amount of Deferred Pension payable to a Former Member and the deferred survivor pension payable to a surviving Spouse under Section 12.01 on December 31 of the prior year shall be increased by the inflation protection factor on January 1 of the current year if the Former Member or Spouse became entitled to receive the Deferred Pension or deferred survivor pension on or before January 1 of the prior year;

Formatted: Highlight

(6) if the Former Member or Spouse became entitled to receive the Deferred Pension or deferred survivor pension during the period from January 1 of the prior year to December 31 of the prior year, the Deferred Pension or the deferred survivor pension shall be increased in the current year by the inflation protection factor prorated for that portion of the prior year from the Termination of Membership Date, or date of death of the Member or Former Member, whichever is applicable, to December 31. When payment of the Deferred Pension or deferred survivor pension commences, the amount of the pension in payment shall be increased on the following January 1 by the inflation protection factor.

14.05 Cap Benefit Increases

- (1) The aggregate benefit increases effected pursuant to Sections 14.03 in respect of the period covered by an actuarial valuation of the Plan for funding purposes shall not exceed the difference between: (a) the prior unutilized funding excesses plus cumulative gains that are disclosed in such actuarial valuation of the Plan and, (b) the amounts required to liquidate the going concern unfunded liability of the Plan, if any, disclosed in such valuation.
- (2) For greater certainty, in no circumstances shall the aggregate benefit increases effected pursuant to Section 14.03 in respect of a particular year exceed the inflation protection factor in respect of those benefits for that year.
- (3) For greater certainty, the cumulative increase provided under Section 14.02 or 14.03 to any person shall not at any time exceed the cumulative increase in the Consumer Protection Index from the year the first increase was so provided to the year of the most recent increase so provided under Section 14.02 or 14.03.

TRUST FUND

15.01 Operation of the Trust Fund

The Board of Trustees shall administer the Plan and the Trust Fund. The Board of Trustees will maintain the Trust Fund during the term of this Plan for the purposes of accepting contributions and providing the benefits in accordance with this Plan. The Trust Fund shall be administered by the Board of Trustees in accordance with the terms of the Sponsorship and Trust Agreement, the Plan, and in compliance with the provisions of the Act, the Income Tax Act, and any other applicable legislation governing the investment of pension funds.

15.02 Provision of Benefits

All benefits under the Plan will normally be paid out of the Trust Fund. However, the Board of Trustees may at any time in its sole discretion purchase out of the Trust Fund from a life insurance company licensed to do business in Canada, benefits of equal amount and payable under the same conditions as the pension to which any retiring Member or other person is entitled under the Plan, so long as such action will not result in the Plan ceasing to be approved or registered for the purpose of the Income Tax Act or any other applicable legislation including the Act. In the event of such a purchase, the Member's or other person's rights under the Plan in respect of said pension benefits shall be deemed to be discharged in full.

15.03 Expenses

All normal and reasonable costs, charges, fees and expenses incurred in the operation of the Trust Fund and the Plan, including the reasonable fees and expenses of the Board of Trustees, Sponsors' Committee, and each of their respective members shall be paid from the Trust Fund in accordance with the terms of the Sponsorship and Trust Agreement and applicable legislation.

In addition, the Board of Trustees may, by resolution, pay, or cause to be paid, from the assets of the Plan such amounts as are necessary to reasonably compensate any person or persons for expenses incurred by such person or persons in the establishment of the Plan governance structure contemplated by the Sponsorship and Trust Agreement including, but not limited to, the establishment of the Board of Trustees, the establishment of the Sponsors' Committee, the establishment of the Trust Fund or the amendment or restatement of the Plan.

15.04 Investments

Subject to the provisions of the Act, the Income Tax Act, and any other appropriate legislation, the Board of Trustees shall direct the investment of the Pension Fund in the manner provided in the Sponsorship and Trust Agreement.

ADMINISTRATION OF THE PLAN

16.01 Administration

The Board of Trustees shall be the administrator of the Plan for purposes of the Act and the Income Tax Act and shall be responsible for the overall operation and administration of the Plan in accordance with its power, duties and responsibilities as provided in the terms of the Sponsorship and Trust Agreement, including the appointment of the Plan Manager. Subject to the limitations of the Plan, the Board of Trustees from time to time shall establish rules for the administration of the Plan and the transaction of business and may delegate certain of its functions to the Plan Manager. The determination of the Board of Trustees, or any applicable committee thereof, as to any disputed question shall be conclusive.

16.02 Indemnity

The employees of the Board of Trustees responsible for the day-to-day administration of the Plan shall use ordinary care and diligence in the performance of their duties, but no such employee shall be personally liable by virtue of contract, agreement, bond or other instrument made or executed by such person or on behalf of such person as an employee of the Board of Trustees nor for any loss unless resulting from such person's own gross negligence or wilful misconduct.

16.03 Information

Each eligible Employee, former Employee, Member, Former Member, Retired Member or any other individual entitled to receive or in receipt of a benefit under the Plan shall provide such information as the Board of Trustees may require in order to establish any eligibility for, or to commence or continue to receive, a pension or other benefit from this Plan.

The Board of Trustees shall provide each eligible Employee, former Employee, Member, Former Member, Retired Member or any other individual entitled to receive or in receipt of a benefit under the Plan with such documents as are required to be disclosed under the Act.

GENERAL PROVISIONS

17.01 Employment Rights

The establishment and implementation of this Plan shall not constitute an enlargement of any rights which a Member has apart from this Plan.

17.02 Board of Trustees Records

Wherever the records of the Board of Trustees are used for the purposes of this Plan, such records shall be conclusive of the facts with which they are concerned.

17.03 Reduction of Benefits

Notwithstanding any other provisions of the Plan, benefits under the Plan may be reduced or contributions made in accordance with the Plan may be refunded as appropriate if such benefits or contributions exceed the maximums limits as prescribed under the Income Tax Act in order to avoid revocation of the Plan, subject to the Act.

17.04 Incapacity to Receive Payments

If, in the opinion of the Board of Trustees, any person receiving a benefit under the terms of this Plan is incapable, either because they are a minor child or as a result of physical or mental infirmity, of managing their own affairs or are incapable of providing a valid receipt for such benefit, the Board of Trustees may authorize any payment to which such person is entitled to be made to a curator or administrator appointed by the Court, or in the absence thereof, payment shall be made to such person's Spouse, child or other person on behalf of such person and such payment shall be a complete discharge of the obligations of the Plan to make payment.

17.05 Non-Assignability and Non-Commutability of Benefits

Other than as required by law or as permitted under the Act benefits under the Plan shall not be subject to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, surrender or charge or to attachment or legal process for debts of the person receiving such benefits, or given as security, nor shall such benefits confer upon any Member, or any other person, any right or interest in the benefit which is capable of surrender or commutation except as specifically provided for in this Plan. Any attempt or agreement to alienate, sell, transfer, assign, pledge, encumber, surrender, give as security or charge any such benefit shall not be recognized and shall be treated as void by the Board of Trustees.

17.06 Determination of Age and Marital Status

Payment of benefits under the Plan shall not be made until the person entitled to payment of the benefits delivers to the Board of Trustees such evidence as the Board of Trustees may require of

- (1) satisfactory proof of age of the person and any other person who may become entitled to payment of the benefit and any other information required to calculate and pay the benefit; and
- (2) if the benefit is payable to a Spouse or Child, satisfactory proof of marital status or the relationship between the Member, Former Member or Retired Member and the Spouse or Child claiming entitlement to a benefit from the Plan.

17.07 Notice

Any notice or election to be given, made or communicated pursuant to or for any purpose of the Plan shall be given, made or communicated, as the case may be, in such manner as the Board of Trustees or Plan Manager shall determine. It shall be the duty of each Member, Former Member, Retired Member and Spouse or Beneficiary, as the case may be, to keep the Plan Manager or Board of Trustees

informed of the correct mailing address and of any other information required by the Board of Trustees or Plan Manager in the administration of the Plan. Any notices required or permitted to be given hereunder shall be deemed given if directed to the addressee at such address and mailed by regular mail.

17.08 Construction

The Plan shall be governed and construed in accordance with the laws applicable in the Province of Ontario

FUTURE OF THE PLAN

18.01 Continuation and Amendment of the Plan

The Sponsors' Committee expects and intends to maintain this Plan in force indefinitely but necessarily reserves the right to amend or discontinue the Plan, either in whole or in part, at any time or times, subject always to the requirements of the Income Tax Act and the provisions of the Act.

18.02 No Reduction in Benefits

For so long as the Plan remains ongoing, no amendment to the Plan shall operate to reduce the benefits as outlined in the Plan which have accrued to Members or other persons entitled to benefits under the Plan prior to the date of such amendment, except as otherwise required or permitted by the Sponsorship and Trust Agreement, or applicable laws.

18.03 Termination of the Plan

- (1) In the event of the termination of the Plan, the Employer shall not be obligated to make any further contributions to the Plan with respect to either past or future service for the Employees after the date of such termination of the Plan, except as may be specifically required by the Act or other applicable legislation to ensure that payments due to the Trust Fund as at the time of termination, but not yet paid by that time, will be made.
- (2) Should the Plan be wholly discontinued or terminated, all required contributions on deposit in the Trust Fund shall be used to provide benefits as outlined in the Plan for Members, Former Members, Retired Members, their respective estates, Spouses, Beneficiaries, or others, in accordance with their respective shares of the Trust Fund through the purchase of

annuity contracts from an insurance company licensed to do business in Canada, or by the transfer of the benefits to which the respective Members, Former Members, Retired Members, and other persons are entitled to the pension plans of subsequent employers or to approved retirement savings plans or by the continuation of the Trust Fund for the provision of deferred pensions as determined by the Board of Trustees or by the payment of cash refunds, or in such other manner as required by law or approved by the government authorities, subject to the requirements of the Income Tax Act and the Act.

18.04 Cessation of an Employer

In the event that any Employer shall cease to participate in the Plan, including as a result of having been wound up or becoming bankrupt, contribution amounts collected on behalf of members or owing from the Employer and required to be remitted to the Plan under Section 4.05 shall be remitted to the Plan promptly.

APPENDIX A

PARTICIPATING EMPLOYERS

The following constitute Employers, pursuant to Section 2.20 of the Plan, as of May 1, 2018.

College Employers

- 1. Algonquin College of Applied Arts and Technology
- 2. Cambrian College of Applied Arts and Technology
- 3. Canadore College of Applied Arts and Technology
- 4. Centennial College of Applied Arts and Technology
- 5. Collège Boréal d'arts appliqués et de technologie
- 6. Collège d'arts appliqués et de technologie La Cité collégiale
- 7. Conestoga College Institute of Technology and Advanced Learning
- 8. Confederation College of Applied Arts and Technology
- 9. Durham College of Applied Arts and Technology
- 10. Fanshawe College of Applied Arts and Technology
- 11. George Brown College of Applied Arts and Technology
- 12. Georgian College of Applied Arts and Technology
- 13. Humber College Institute of Technology and Advanced Learning
- 14. Lambton College of Applied Arts and Technology
- 15. Loyalist College of Applied Arts and Technology
- 16. Mohawk College of Applied Arts and Technology
- 17. Niagara College of Applied Arts and Technology
- 18. Northern College of Applied Arts and Technology
- 19. St. Clair College of Applied Arts and Technology
- 20. St. Lawrence College of Applied Arts and Technology
- 21. Sault College of Applied Arts and Technology
- 22. Seneca College of Applied Arts and Technology
- 23. Sheridan College Institute of Technology and Advanced Learning
- 24. Sir Sandford Fleming College of Applied Arts and Technology

Non-College Employers

- 1. Ontario Colleges Library Service
- 2. Ontario College Application Service
- 3. College Employer Council
- 4. Northern Centre for Advanced Technology (closed group of eight)
- 5. OntarioLearn Online Collaborative
- 6. OPSEU Local 415
- 7. OPSEU Local 562
- 8. Colleges Ontario
- 9. Mohawk Students' Association
- 10. OPSEU Local 110
- 11. Ignite Student Union (formerly Humber Students' Federation)
- 12. Fanshawe Student Union
- 13. The Royal Ontario Museum
- 14. The Royal Ontario Museum Foundation
- 15. The Board of Trustees
- 16. Sheridan Student Union
- 17. Youth Services Bureau of Ottawa
- 18. Canadian Collegiate Athletic Association
- 19. Lambton College Student Administrative Council
- 20. Community Living Toronto (In respect only of its employees subject to Appendix L and in respect of any other employees of Community Living Toronto who, on an after October 1, 2019, hold a full-time executive position. For the avoidance of doubt, should any such employee cease to hold a full-time executive position while remaining employed with Community Living Toronto then the applicable provisions of Appendix A.1 shall apply.)
- 21. Healthcare Excellence Canada—effective October 1, 2022 and only in respect of its Active Transferring Plan Members (as that term is defined in Appendix W) that were hired prior to October 1, 2020 and are employed on a regular full-time basis with Healthcare Excellence Canada

APPENDIX A.1

PARTICIPATING EMPLOYERS AND MEMBER CONTRIBUTIONS OF DESIGN TWO EMPLOYEES

A. PARTICIPATING EMPLOYERS UNDER APPENDIX A.1

The following constitute Employers, pursuant to Section 2.20 of the Plan:

- 1. Torstar Corporation effective October 1, 2018 and subject to the terms of Appendix H and B.(2) and C.(1) below
- 2. Toronto Star Newspapers Limited effective October 1, 2018 and subject to the terms of Appendix H and B.(2) and C.(1) below, as applicable
- 3. Metroland Media Group Ltd effective October 1, 2018 and subject to the terms of Appendix H and B.(2) and C.(1) below, as applicable
- 4. Performance Printing Ltd. effective January 1, 2019 and subject to the terms of B.(2) and C.(1) below
- 5. Autocatch.com Inc. effective January 1, 2019 and subject to the terms of B.(2) and C.(1) below
- 6. Free Daily News Group Inc. effective January 1, 2019 and subject to the terms of B.(2) and C.(1) below
- 7. EyeReturn Marketing Inc. effective January 1, 2019 and subject to the terms of B.(2) and C.(1) below
- 8. Canadian Press Enterprises Inc. effective July 1, 2019 and subject to the terms of Appendix I and B.(3) and C.(3) below
- 9. Shareholder Association for Research & Education effective January 1, 2019 and subject to the terms of B.(4) and C.(3) below
- 10. OPSEU Local 416 effective June 1, 2019 and subject to the terms of B.(1) and C.(2) below.
- 11. London Cross Cultural Learner Centre effective July 1, 2019 and subject to the terms of B.(5) and C.(2) below
- 12. Postmedia Network Inc. effective July 1, 2019 and subject to the terms of Appendix J and B.(6) and C.(4) below
- 13. Klos Concepts Ltd effective September 1, 2019 and subject to the terms of B.(7) and C.(2) below
- 14. University of Saskatchewan effective September 1, 2019 and subject to the terms of B.(8) and C.(5) below
- 15. Community Living Toronto effective October 1, 2019 and subject to the terms of Appendix K and B.(9) and C.(6) below
- 16. United Way of Greater Toronto—effective January 1, 2020 and subject to the terms of Appendix M and B.(10) and C.(2) below
- 17. FP Canadian Newspapers Limited Partnership—effective January 1, 2020 and subject to the terms of Appendix N and B.(11) and C.(7) below
- 18. Wright Henry LLP—effective January 1, 2020 and subject to the terms of B.(12) and C.(8) below

- 19. Catholic Charities of the Archdiocese of Toronto—effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 20. Catholic Cross Cultural Services—effective January 1, 2020 and subject to the terms of Appendix O and B.(14) and C.(2) below
- 21. Catholic Family Services of Durham— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 22. Catholic Family Services of Peel-Dufferin—effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 23. Catholic Family Services of Simcoe County— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 24. Catholic Family Services of Toronto— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 25. Catholic Settlement House Day Nursery— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 26. Mary Centre of the Archdiocese of Toronto—effective January 1, 2020 and subject to the terms of Appendix O and B.(15) and C.(2) below
- Our Place Community of Hope Centre (Metropolitan Toronto)— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 28. Rosalie Hall— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- Sancta Maria House— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 30. Silent Voice Canada Inc.— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 31. St Michael's Homes— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 32. LA Centre for Active Living— effective January 1, 2020 and subject to the terms of Appendix O and B.(13) and C.(2) below
- 33. Rose of Durham—effective January 1, 2020 and subject to the terms of Appendix O and B.(16) and C.(2) below
- 34. Skills Compétences Canada Ontario—effective January 1, 2020 and subject to the terms of B.(12) and C.(2) below
- 35. Fraserken Marketing Services Inc. DBA Fraser Group—effective March 1, 2020 and subject to the terms of B.(17) and C.(2) below
- 36. Vopak Terminals of Canada Inc.—effective June 1, 2020 and subject to the terms of B.(18) and C.(9) below
- 37. Canadian Institute of Actuaries—effective June 1, 2020 and subject to the terms of B.(19) and C.(2) below
- 38. Brink's Canada Limited—effective July 1, 2020 in respect of employees of the Employer who are members of a Unifor bargaining unit (excluding the Montreal Unifor bargaining unit); effective January 1, 2021 in respect of employees of the Employer who are members of the Teamster Local 927 bargaining unit and are employed in New Brunswick and Prince Edward Island; and effective February 1, 2022 in respect of employees of the Employer who are members of the Teamster Local 927 bargaining unit and are employed in Nova Scotia; and effective February 1, 2022 in respect of employees of the Employer who are members of

- the Newfoundland and Labrador Association of Public and Private Employees and are employed in Newfoundland and Labrador and subject to the terms of B.(12) and C.(10) below
- 39. Saint John Airport Inc.—effective July 1, 2020 and subject to the terms B.(20) and C.(2) below
- Forensic Examiners Inc.—effective July 1, 2020 and subject to the terms B.(1) and C.(14)
- 41. Injured Workers Community Legal Clinic—effective July 1, 2020 and subject to the terms B.(21) and C.(2)
- 42. Canadian Bar Insurance Association—effective August 1, 2020 and subject to the terms B.(12) and C.(2)
- 43. Simon Muller Professional Corporation—effective August 1, 2020 and subject to the terms B.(1) and C.(14) below
- 44. Alam Professional Corporation—effective August 1, 2020 and subject to the terms B.(1) and C.(14) below
- 45. Lawyers Financial Advisory Services—effective August 1, 2020 and subject to the terms B.(12) and C.(2) below
- 46. Gordon Prisco Law Professional Corporation—effective September 1, 2020 and subject to the terms B.(1) and C.(2) below
- 47. Ross Gascho Professional Corporation—effective September 1, 2020 and subject to the terms B.(1) and C.(14) below
- 48. Ontario Teachers Insurance Plan—effective September 7, 2020 and subject to the terms B.(22) and C.(11) below
- 49. OTIP/RAEO Benefits Incorporated—effective September 7, 2020 and subject to the terms B.(22) and C.(11) below
- 50. Curo Claims Services Inc.—effective September 7, 2020 and subject to the terms of B.(22) and C.(11) below
- 51. OTIP/RAEO Insurance Brokers Inc.—effective September 7, 2020 and subject to the terms of B.(22) and C.(11) below
- 52. TW Insurance Brokers Inc.—effective September 7, 2020 and subject to the terms of B.(22) and C.(11) below
- 53. TW Services Ltd.—effective September 7, 2020 and subject to the terms of B.(22) and C.(11) below
- 54. Vancouver Foundation—effective October 1, 2020 and subject to the terms of B.(23) and C.(12) below
- 55. British Columbia Unclaimed Property Society—effective October 1, 2020 and subject to the terms of B.(23) and C.(13) below
- 56. Municipal Benchmarking Networking of Canada—effective October 1, 2020 and subject to the terms of B.(1) and C.(2) below
- 57. Andrew Law Office—effective October 1, 2020 and subject to the terms B.(12) and C.(14) below
- 58. Idowu Ohioze Professional Corporation—effective October 1, 2020 and subject to the terms B.(24) and C.(14) below
- 59. Brenda Wong Law Professional Corporation—effective October 1, 2020 and subject to the terms B.(12) and C.(14) below
- 60. Lisa Belcourt Professional Corporation—effective October 1, 2020 and subject to the terms B.(12) and C.(14) below

- 61. Patrick Sloan Professional Corporation—effective October 1, 2020 and subject to the terms B.(12) and C.(14) below
- 62. Randy Bauslaugh Professional Corporation—effective October 1, 2020 and subject to the terms B.(1) and C.(2)
- 63. College of Physicians and Surgeons of Newfoundland and Labrador—effective November 1, 2020 and subject to the terms of B.(1) and C.(2) below
- 64. Bradley K Bullock Professional Corporation—effective November 1, 2020 and subject to the terms B.(20) and C.(14), and effective January 1, 2022 subject to B.(60) and C.(2) below
- 65. Audrey Lee Professional Corporation—effective January 1, 2021 and subject to the terms B.(25) and C.(14) below
- 66. Richardson Hall LLP—effective January 1, 2021 and subject to the terms B.(12) and C.(2) below
- 67. Jonathan M. Richardson Professional Corporation—effective January 1, 2021 and subject to the terms B.(12) and C.(2) below
- 68. Andrew Harrison Profession Corporation—effective January 1, 2021 and subject to the terms B.(1) and C.(2) below
- 69. Greenwood Defence Law Professional Corporation—effective January 1, 2021 and subject to the terms B.(12) and C.(14) below
- 70. Ibiyemi O. Philip Professional Corporation—effective January 1, 2021 and subject to the terms B.(12) and C.(14) below
- 71. Jonathan Rodger Professional Corporation—effective January 1, 2021 and subject to the terms B.(1) and C.(14) below
- 72. Travis Sippel Law Corporation—effective January 1, 2021 and subject to the terms B.(20) and C.(2) below
- 73. C. Matthews Professional Corporation—effective January 1, 2021 and subject to the terms B.(26) and C.(14) below
- 74. Stacy L. Maurier Professional Corporation—effective January 1, 2021 and subject to the terms B.(27) and C.(14) below
- 75. Greater Toronto Airports Authority—effective January 1, 2021 and subject to the terms B.(28) and C.(15) below
- 76. R. Snyder-Penner Professional Corporation—effective February 1, 2021 and subject to the terms B.(29) and C.(2) below
- 77. Wayforth LLP—effective February 1, 2021 and subject to the terms B.(29) and C.(2) below
- 78. Sobering Professional Corporation—effective February 1, 2021 and subject to the terms B.(29) and C.(2) below
- 79. Hirschorn Law Professional Corporation—effective February 1, 2021 and subject to the terms B.(1) and C.(2) below
- 80. Ian Goldman Law Corporation—effective February 1, 2021 and subject to the terms B.(1) and C.(2) below
- 81. Walker Sorensen LLP—effective February 1, 2021 and subject to the terms B.(12) and C.(2) below
- 82. Chown Smith and Scott Professional Corporation—effective March 1, 2021 and subject to the terms B.(30) and C.(14) below
- 83. Nanuq Law Corporation—effective March 1, 2021 and subject to the terms B.(31) and C.(14) below

- 84. Sackville Law Inc.—effective from March 1, 2021 to November 30, 2023 and subject to the terms B.(32) and C.(2) below
- 85. CaleyWray—effective April 1, 2021 and subject to the terms B.(12) and C.(2) below
- 86. Micheil Russell Barrister Solicitor Professional Corporation—effective April 1, 2021 and subject to the terms B.(1) and C.(2) below
- 87. Simner Corporation—effective April 1, 2021 and subject to the terms B.(1) and C.(2) below
- 88. Metriks Education Inc. —effective April 1, 2021 and subject to the terms B.(1) and C.(2) below
- 89. BCH Actuarial Service Inc. —effective April 1, 2021 and subject to the terms B.(1) and C.(2) below
- The Priory of Canada of the Most Venerable Order of the Hospital of St. John of Jerusalem—effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 91. L'Ordre de Saint-Jean, Conseil du Quebec effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 92. St. John Council for Alberta- effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 93. St. John Council for Federal District—effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 94. St. John Council for Manitoba, Northwest Territories and Nunavut– effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 95. St. John Council for New Brunswick Inc. effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 96. St. John Council for Newfoundland Properties Inc. effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 97. St. John Council for Nova Scotia/Prince Edward Island—effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 98. St. John Council for Ontario—effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 99. St. John Council for Saskatchewan Inc. effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 100. St. John Society (British Columbia & Yukon) effective April 1, 2021 and subject to the terms of Appendix P and B.(33) and C.(2) below
- 101. Sanofi Pasteur Limited—effective April 1, 2021 and subject to the terms of Appendix Q and B.(34) and C.(16) below
- 102. Allevato Quail & Roy Law Corporation (prior to April 1, 2022 operating as "Allevato Quail & Roy, Barristers & Solicitors")—effective April 1, 2021 and subject to the terms B.(35) and C.(2) below
- 103. The Globe and Mail Inc.—effective May 1, 2021 and subject to the terms of Appendix S and B.(36) and C.(2) below
- 104. Digital Research Alliance of Canada (prior to October 1, 2021 operating as "11593765 Canada Association" or "New Digital Research Infrastructure Organization")—effective May 1, 2021 and subject to the terms B.(37) and C.(17) below

- 105. Roni Excavating Ltd.—effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 106. Rion Estimating and Consulting Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 107. Grand Bovino Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 108. Orin Contractors Corp. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 109. Orin Enterprises Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 110. Iron Forming Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 111. Iron Shoring Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 112. Iron Construction and Developments Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 113. New Age Forming Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 114. Orin Contracting Services Inc. —effective May 1, 2021 and subject to the terms B.(38) and C.(18) below
- 115. Guild House Technical Institute—effective May 1, 2021 and subject to the terms B.(1) and C.(2) below
- 116. Ontario Crown Attorney's Association—effective May 1, 2021 and subject to the terms B.(1) and C.(8) below
- 117. Barbara Hendrickson Professional Corporation—effective May 1, 2021 and subject to the terms B.(1) and C.(14) below
- 118. Societe juridique personelle David Schulze Inc. —effective May 1, 2021 and subject to the terms B.(1) and C.(14) below
- 119. George Jeerakathil Law Consulting Professional Corporation—effective May 1, 2021 and subject to the terms B.(1) and C.(14) below
- 120. Goldman Hine LLP—effective May 1, 2021 and subject to the terms B.(12) and C.(2) below
- 121. Huron Perth Community Legal Clinic—effective May 1, 2021 and subject to the terms B.(12) and C.(2) below
- 122. Travis Moffat Professional Corporation—effective May 1, 2021 and subject to the terms B.(1) and C.(2) below
- 123. Antonios T. Antoniou Professional Corporation—effective May 1, 2021 and subject to the terms B.(1) and C.(14) below
- 124. Amanda E. Bafaro Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below
- 125. Andrew Leonard Keesmaat Professional Corporation—effective June 1, 2021 and subject to the terms B.(1) and C.(14) below
- 126. Beresford Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below
- 127. Chandra L. Flett Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below

- 128. Hardie Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below
- 129. Jacqueline Bart Immigration Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below
- James K Elsley Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(1) and C.(14) below
- O'Connor Richardson Professional Corporation—effective June 1, 2021 and subject to the terms B.(12) and C.(14) below
- 132. Yusra Siddiquee Law Professional Corporation—effective June 1, 2021 and subject to the terms B.(1) and C.(2) below
- 133. Accertaclaim Servicorp Inc.—effective June 1, 2021 and subject to the terms B.(39) and C.(2) below
- 134. Colleges and Institutes Canada—effective June 1, 2021 and subject to the terms B.(40) and C.(2) below
- 135. Peel Word Processing Inc.—effective June 1, 2021 and subject to the terms B.(41) and C.(8) below
- 136. Aeric Inc. (Operating as The Conference Board of Canada)—effective June 1, 2021 and subject to the terms B.(42) and C.(2) below
- 137. Ontario Dental Association—effective June 1, 2021 and subject to the terms B.(43) and C.(19) below
- 138. Huntington University—effective June 13, 2021 and subject to the terms B.(44) and C.(2) below
- 139. The George Hull Centre for Children and Families—effective July 1, 2021 and subject to Appendix R and the terms B.(45) and C.(20) below
- 140. Hamilton Community Legal Clinic—effective July 1, 2021 and subject to the terms B.(12) and C.(2) below
- 141. Krauss Law Professional Corporation—effective July 1, 2021 and subject to the terms B.(1) and C.(14) below
- 142. Nipissing Community Legal Clinic—effective July 1, 2021 and subject to the terms B.(21) and C.(14) below
- 143. Allan Ludkiewicz, Barristers and Solicitors—effective July 1, 2021 and subject to the terms B.(46) and C.(2) below
- 144. University of Victoria Students' Society—effective July 1, 2021 and subject to the terms B.(47) and C.(21) below
- 145. Tanglao Professional Corporation—effective August 1, 2021 and subject to the terms B.(1) and C.(2) below
- 146. Bradbury Sippel Law Corporation—effective August 1, 2021 and subject to the terms B.(12) and C.(8) below
- 147. Chatham-Kent Legal Clinic—effective August 1, 2021 and subject to terms B.(21) and C.(14) below
- 148. Brookshire Law Office—effective August 1, 2021 and subject to the terms B.(48) and C.(2) below
- 149. Teryl Scott Lawyers Inc.—effective August 1, 2021 and subject to the terms B.(48) and C.(2) below
- 150. Miskin Law Professional Corporation—effective August 1, 2021 and subject to the terms B.(49) and C.(2) below

- 151. Canadian Superior Courts Judges Association—effective August 1, 2021 and subject to the terms B.(50) and C.(8) below
- 152. Jenkins, Newman & Shipley Law Professional Corporation—effective August 1, 2021 and subject to the terms B.(12) and C.(14) below
- 153. Downsview Community Legal Services—effective August 1, 2021 and subject to the terms B.(21) and C.(2) below
- 154. Association of Justice Counsel—effective August 1, 2021 and subject to the terms B.(51) and C.(22) below
- 155. McRae Law—effective September 1, 2021 and subject to the terms B.(1) and C.(2) below
- 156. Peter Loucks Professional Corporation—effective September 1, 2021 and subject to the terms B.(12) and C.(2) below
- 157. United Way Centraide North East Ontario—effective October 1, 2021 and subject to the terms B.(52) and C.(2) below
- 158. Clinique Juridique Roy McMurtry Legal Clinic (Stormont, Dundas & Glengarry)—effective October 11, 2021 and subject to the terms B.(21) and C.(2) below
- 159. National Association of Federal Retirees—effective October 22, 2021 and subject to the terms B.(20) and C.(23) below
- 160. St. Stanislaus-St. Casimir's Polish Parishes Credit UnionLimited.—effective November 1, 2021 and subject to the terms B.(53) and C.(24) below
- 161. Workers' Health and Safety Legal Clinic—effective November 1, 2021 and subject to the terms B.(1) and C.(14) below
- Kinna-Aweya Legal Clinic—effective November 1, 2021 and subject to the terms
 B.(12) and C(2) below
- 163. Salih Law Professional Corporation—effective December 1, 2021 and subject to the terms B.(54) and C.(2) below
- 164. Above the Fold Media Inc.—effective January 1, 2022 and subject to the terms B.(20) and C.(2) below
- 165. 3Sab Networks Ltd.—effective January 1, 2022 and subject to the terms B.(55) and C.(2) below
- 166. Canadian Police Association/Association Canadienne Des Policiers—effective January 1, 2022 and subject to the terms B.(1) and C.(25) below
- 167. University of Sudbury—effective January 1, 2022 and subject to the terms B.(37) and C.(2) below
- 168. Silvia de Sousa Law Corporation—effective January 1, 2022 and subject to the terms B.(12) and C.(14) below
- 169. Thomas A. Bettle Professional Corporation—effective January 1, 2022 and subject to the terms B₂(12) and C₂(14) below
- 170. North Peel & Dufferin Community Legal Services Inc.—effective January 1, 2022 and subject to the terms B.(21) and C.(14) below
- 171. Advocacy Centre for Tenants Ontario—effective January 1, 2022 and subject to the terms B.(12) and C.(2) below
- 172. Charles A. Haché Professional Corporation—effective January 1, 2022 and subject to the terms B.(1) and C.(14) below
- 173. David Rosenfeld Professional Corporation—effective January 1, 2022 and subject to the terms B.(12) and C.(14) below

- 174. Alexander K.V. Shalashniy Legal Services (Prof. Corp.)—effective January 1, 2022 and subject to the terms B.(3) and C.(14) below
- 175. Speigel Nichols Fox LLP—effective January 1, 2022 and subject to the terms B.(56) and C.(2) below
- 176. Sternat Manaigre Law Corporation—effective January 1, 2022 and subject to the terms B.(56) and C.(8) below
- 177. Industrial Accident Victims Group of Ontario—effective January 1, 2022 and subject to the terms B.(57) and C.(2) below
- 178. Shift Works Strategic Inc.—effective January 1, 2022 and subject to the terms B.(1) and C.(2) below
- 179. Angela Chaisson Law Professional Corporation—effective January 1, 2022 and subject to the terms of B.(12) and C.(14) below
- 180. DNN Law Corporation—effective January 1, 2022 and subject to the terms of B.(1) and C.(14) below
- 181. Mayor Law Corporation—effective January 1, 2022 and subject to the terms B.(1) and C.(14) below
- 182. Mitch Frazer Professional Corporation—effective January 1, 2022 and subject to the terms B.(1) and C.(14) below
- 183. Kim Ferreira Professional Corporation—effective January 1, 2022 and subject to the terms B.(3) and C.(2) below
- 184. Allison Speigel Professional Corporation—effective January 1, 2022 and subject to B.(56) and C.(14)
- 185. Accerta Services Inc.—effective January 17, 2022 and subject the terms B.(12) and C.(2) below
- 186. Jamieson Laboratories Ltd.—effective January 30, 2022 and subject to the terms B.(58) and C.(26) below
- 187. Caza Saikaley srl/LLP—effective February 1, 2022 and subject to the terms of B.(59) and C.(14) below
- 188. Karen F. Redmond Law Corporation—effective February 1, 2022 and subject to the terms B.(1) and C.(2) below
- 189. Nancy Shapiro Professional Corporation—effective February 1, 2022 and subject to the terms B.(1) and C.(14) below
- 190. Miltons IP Professional Corporation—effective February 1, 2022 and subject to the terms B.(59) and C.(8) below
- 191. Mississauga Community Legal Services—effective February 1, 2022 and subject to the terms B.(21) and C.(27) below
- 192. Heritage Law Corporation—effective February 1, 2022 and subject to the terms B.(12) and C.(2) below
- 193. Burnside Law Group Limited—effective March 1, 2022 and subject to the terms B.(12) and C.(2) below
- 194. Darren M Cocchetto Law Professional Corporation—effective March 1, 2022 and subject to the terms B.(1) and C.(2) below
- 195. Ernest A. Schirru Professional Corporation—effective March 1, 2022 and subject to the terms B.(12) and C.(2) below
- 196. MC LegalManagement Limited Partnership—effective March 28, 2022 and subject to Appendix T and the terms B.(12) and C.(28) below

- 197. Aleisha Stevens Barrister Solicitor Professional Corporation—effective April 1, 2022 and subject to the terms B.(12) and C.(14) below
- 198. Brandin O'Connor Law Professional Corporation—effective <u>from April 1, 2022</u> to April 27, 2023 and subject to the terms B.(61) and C.(2) below
- 199. CL Burchett Professional Corporation—effective April 1, 2022 and subject to the terms B.(62) and C.(14) below
- Moore Edgar Lyster LLP—effective April 1, 2022 and subject to the terms B.(61) and C.(2) below
- 201. Scarborough Community Legal Services—effective April 1, 2022 and subject to the terms B.(21) and C.(2) below
- Vandergrift Legal Professional Corporation—effective April 1, 2022 and subject to the terms B.(1) and C.(14) below
- 203. Gardaworld Cash Services Canada Corporation—effective April 1, 2022 and subject to the terms B.(63) and C.(29) below
- Saskatchewan College of Pharmacy Professionals—effective May 1, 2022 and subject to the terms B.(64) and C.(2) below
- The Students' Association of the Algonquin College of Applied Arts and Technology Corporation—effective May 16, 2022 and subject to the terms B. (65) and C.(30)
- Snow Lawyers Professional Corporation—effective June 1, 2022 and subject to the terms B.(20) and C.(14) below
- Community Legal Services of Ottawa/Services Juridiques Communautaires d'Ottawa—effective June 6, 2022 and subject to the terms B.(21) and C.(2) below
- 208. Burford Law Professional Corporation—effective June 1, 2022 and subject to the terms B.(66) and C.(2) below
- Canadian Coalition for Good Governance—effective July 1, 2022 and subject to the terms B.(67) and C.(2) below
- 210. Reuters Canada Limited—effective July 1, 2022 and subject to the terms B.(68) and C.(31) below
- 211. ASC Aquaculture Services, Inc.—effective July 1, 2022 and subject to the terms B.(69) and C.(32) below
- 212. Cameron Koziskie Professional Corporation—effective July 1, 2022 and subject to the terms B.(1) and C.(14) below
- 213. Don Valley Community Legal Services—effective July 1, 2022 and subject to the terms B.(21) and C.(14) below
- 214. Ellen Low Professional Corporation—effective July 1, 2022 and subject to the terms B.(70) and C.(2) below
- 215. James J. Brink Professional Corporation—effective July 1, 2022 and subject to the terms B.(3) and C.(14) below
- 216. DeMelo Law Professional Corporation—effective July 1, 2022 and subject to the terms B.(70) and C.(2) below
- 217. Kendelle L. Pollitt Law Corporation—effective August 1, 2022 and subject to the terms B.(71) and C.(2)
- 218. Bukola Abioye Professional Corporation (operating as "Daphil Law Office")—effective August 1, 2022 and subject to the terms B.(1) and C.(2) below
- 219. The College of Family Physicians of Canada—effective August 1, 2022 and subject to the terms B.(12) and C.(2) below

- 220. The British Columbia College of Family Physicians (a chapter of the College of Family Physicians of Canada)—effective August 1, 2022 and subject to the terms B.(12) and C.(2) below
- 221. New Brunswick College of Family Physicians—effective August 1, 2022 and subject to the terms B.(12) and C.(2) below
- 222. The Nova Scotia College of Family Physicians—effective August 1, 2022 and subject to the terms B.(12) and C.(2) below
- 223. The Saskatchewan College of Family Physicians—a Chapter of the College of Family Physicians of Canada—effective August 1, 2022 and subject to the terms B.(12) and C.(2) below
- 224. Ontario Public Service Quarter Century Club—effective August 28, 2022 and subject to the terms B.(72) and C.(33) below
- 225. West Toronto Community Legal Services Inc.—effective September 1, 2022 and subject to the terms B.(73) and C.(34) below
- 226. First Peoples' Cultural Foundation—effective September 1, 2022 and subject to the terms B.(74) and C.(35) below
- 227. Everstone Law Professional Corporation—effective September 1, 2022 and subject to the terms B.(12) and C.(2) below
- 228. Kingston Community Legal Clinic—effective September 1, 2022 and subject to the terms B.(21) and C.(2) below
- 229. Healthcare Excellence Canada—effective October 1, 2022 and subject to the terms B.(1) and C.(36) below
- 230. H.H. Wright Law Services Corporation Incorporated—effective October 1, 2022 and subject to the terms B.(1) and C.(2) below
- 231. Angus Consulting Management Limited—effective September 1, 2022 and subject to the terms B.(75) and C.(37) below
- 232. ACML Management Western Limited—effective September 1, 2022 and subject to the terms B.(75) and C.(37) below
- 233. Parkdale Community Legal Services Inc.—effective October 1, 2022 and subject to the terms B.(76) and C.(38) below
- 234. Ontario Council on Articulation and Transfer—effective October 1, 2022 and subject to the terms B.(77) and C.(39) below
- 235. Rio Tinto Alcan Inc.—effective October 31, 2022 and subject to the terms B.(78) and C.(40) below
- 236. Collége Nordique Francophone—effective November 13, 2022 and subject to the terms B.(80) and C.(41) below
- 237. Jamie J. Feenan Professional Corporation (operating as "Feenan Law")—effective December 1, 2022 and subject to the terms B.(46) and C.(2) below
- 238. City of Whitehorse—effective December 14, 2022 and subject to the terms B.(79) and C.(42) below
- 239. iPolitics Limited—effective December 18, 2022 and subject to the terms and subject to the terms B.(12) and C.(2) below
- 240. C.D. Howe Institute—effective January 1, 2023 and subject to the terms B.(80) and C.(43) below
- 241. Elevate Export Finance Corp.—effective January 1, 2023 and subject to the terms B.(81) and C.(44) below

- 242. Global First Power Ltd.—effective January 1, 2023 and subject to the terms B.(82) and C.(47) below
- 243. Candu Energy Inc.—effective January 1, 2023 and subject to the terms B.(82) and C.(45) below
- 244. Community Living Thunder Bay—effective January 1, 2023 in respect of its unionized employees and October 14, 2023 in respect of its non-unionized employees and subject to the terms B.(84) and C.(49) below
- 245. Laurentis Energy Partners—effective January 1, 2023 and subject to the terms B.(85) and C.(50) below
- Koskie Minsky LLP—effective January 1, 2023 and subject to the terms B.(86) and C.(2) below
- 247. Ira Parghi Professional Corporation—effective January 1, 2023 and subject to the terms B.(1) and C.(14) below
- 248. Sotos LLP—effective January 1, 2023 and subject to the terms B.(12) and C.(51) below
- 249. MDC Law Limited—effective January 1, 2023 and subject to the terms B.(87) and C.(8) below
- Tomi Olutunfese Law Office, Professional Corporation—effective January 1,
 2023 and subject to the terms B.(46) and C.(2) below
- Journey Law Professional Corporation—effective January 1, 2023 and subject to the terms B.(88) and C.(2) below
- 252. Cooper Lord Law Professional Corporation—effective January 1, 2023 and subject to the terms B.(89) and C.(2)
- Themis Management Corporation—effective January 1, 2023 and subject to the terms B.(12) and C.(2) below
- 254. Sujit Choudhry Professional Corporation—effective January 1, 2023 and subject to the terms B.(1) and C.(14) below
- Glasscan Corporation—effective January 2, 2023 and subject to the terms B.(90) and C.(52) below
- Superior Greenstone Association for Community Living—effective January 22,
 2023 and subject to the terms B.(92) and C.(53) below
- 257. Canadian Pharmacists Association—effective January 22, 2023 and subject to the terms B.(93) and C.(35) below
- 258. Eric Grigg Professional Corporation—effective February 1, 2023 and subject to the terms B.(20) and C.(2) below
- 259. Grace Fellowship Canada—effective March 1, 2023 and subject to the terms B.(94) and C.(54) below
- 260. Catherine Wilde Prof. Corp—effective March 1, 2023 and subject to the terms B.(95) and C.(8) below
- 261. Jessica L. Burke Law Corporation—effective March 1, 2023 and subject to the terms B.(3) and C.(14) below
- 262. LiveWorkPlay—effective March 11, 2023 and subject to the terms B.(96) and C.(2) below
- 263. Christ The King Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 264. St. John/St. Peter Parish—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 265. Good Shepherd Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 266. Lutheran Church of Good Shepherd—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 267. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 268. Prince of Peace Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 269. Foothills Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 270. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 271. Lutheran Church Of The Good Shepherd—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 272. Prince of Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 273. St. Matthew Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 274. Grace Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 275. Our Saviours Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- St. John Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 278. Christ Our Hope—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 279. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 281. St. Paul Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 282. Saviour Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 283. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 284. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 285. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 286. Our Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 287. Trinity Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 288. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 289. St. Paul's & Emmanuel Lutheran Churches—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 290. All Saints Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 291. Bethlehem Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 292. Concordia Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- Concordia Lutheran Seminary—effective April 1, 2023 and subject to the terms
 B.(83) and C.(46) below
- 294. Cross of Christ Lutheran Church of the Deaf—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 295. Riverbend Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 296. St. Lukes/Zion Lutheran Churches—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 297. St. Paul Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 298. St. Peters Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- Our Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 300. Trinity Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 301. Concordia Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 302. Church Of The Lutheran Hour—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 303. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 304. Our Saviour Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 305. Bethany Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 306. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 307. Trail Peace-Trinity—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 308. Walnut Grove Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 309. Immanuel Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 310. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 311. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 312. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 313. Christ Our Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 314. Pilgrim Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 315. First St. Matthew Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 316. Good Shepherd Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 317. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 318. Holy Trinity & Hope Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 319. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 320. Lord Of Life Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 321. Christ Risen Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 322. First Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 323. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 324. Southampton Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 325. First Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 326. Bethel Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 327. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 328. Grace Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 329. Historic St. Pauls Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 330. Holy Cross Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 331. Hope Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 332. Luth Laymens League Of Canada—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 333. La Ronge Lutheran Fellowship—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 334. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 335. St. Peter Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 336. First Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 337. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 338. Good Shepherd Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 339. Our Saviour Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 340. St. Paul Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 341. Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 342. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 343. Christ Our King Luth Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 344. Risen Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 345. St. Marks Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 346. Grace Ev. Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 347. First Lutheran Church, Logan—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 348. Emmanuel Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 349. St. John's Evang.Lutheran Ch.—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 350. St. Paul's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 351. Christ Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 352. Mt. Calvary Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 353. Our Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 354. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 355. Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 356. Grace Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 357. Our Saviour Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 358. Our Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 359. Prince Of Peace Luth Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 360. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- Grace Lutheran Church-Locksley—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 362. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 363. Concordia Lutheran Church and School—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 364. Emmanuel Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 365. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 366. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 367. St. Paul's Luth.Church (Port Hope)—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 368. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 369. Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 370. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 371. Mt. Calvary Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 372. Christ Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 373. Good Shepherd Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 374. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 375. Mount Olive Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 376. New Beginnings Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 377. Prince Of Peace Luth. Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 378. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 379. Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 380. St. Matthew Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 381. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 382. Grace Lutheran Church, Inc—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 383. St. Pauls Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 384. St. Matthews Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 385. Bethel Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 386. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 387. St.Peter's Ev. Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 388. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 389. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 390. Concordia Luth.Theol. Seminary—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 391. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 392. Resurrection Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 393. Grace LC / Redeemer LC—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 394. King Of Kings Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 395. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 396. St. John Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 397. First St Johns Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 398. St. Peters Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 399. St. Pauls Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 400. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 401. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 402. Faith Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 403. Mt. Calvary Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 404. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 405. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 406. Epiphany Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 407. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 408. Christ Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 409. Toronto Korean Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 410. Trinity Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 411. The Church Of St. Luke—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 412. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 413. Bethlehem Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 414. Killarney Community Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 415. Trinity Luth Church Of Deaf—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 416. St. Johns Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 417. Hope Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 418. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 419. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 420. Messiah Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 421. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 422. Lutheran Church—Canada—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 423. First St. Paul Luth Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 424. St John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 425. Redeemer Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 426. Zion Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 427. Advent Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 428. St. John Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 429. First Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 430. Gethsemane Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 431. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 432. Trinity Ev Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 433. Beautiful Savior Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 434. Immanuel Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 435. Luth Church Of Good Shepherd—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 436. Luth. Church of The Redeemer—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 437. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 438. Saint James Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 439. Zion Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 440. Faith Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 441. Lutheran Bible Translators of Canada, Inc —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 442. St. Paul's Evangelical Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 443. Immanuel Lutheran Church —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 444. St. Luke Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 445. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 446. B.C. Mission Boat Society—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 447. LCC Financial Ministries—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 448. St. Johns Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 449. Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 450. Lord of All Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 451. LCC Worker Benefits Services Inc.—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 452. Hope Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 453. St. Matthew Evangelical Lutheran Church & Christian Academy—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 454. Grace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 455. Shepherd King Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 456. St. Peter Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 457. St. John's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 458. St. Albert Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 459. Peace Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 460. Our Saviours Community LC/Bethlehem Lutheran—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 461. Bethlehem Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 462. St. Peter's Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 463. Ascension Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 464. Kingman Lutheran Parish —effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 465. Hope Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 466. Faith Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 467. Holy Trinity Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 468. Shepherd of the Valley Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 469. Good Shepherd Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 470. St. Andrew's Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 471. Bethany Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below

- 472. Calgary Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 473. St. Luke Evangelical Lutheran Church—effective April 1, 2023 and subject to the terms B.(83) and C.(46) below
- 474. Pan-Canadian Pharmaceutical Alliance Corporation—effective April 1, 2023 and subject to the terms B.(1) and C.(53) below
- 475. York University Faculty Association—effective April 1, 2023 and subject to the terms B.(97 and C.(55) below
- 476. Tetyana İvanina Professional Corporation—effective April 1, 2023 and subject to the terms B.(12) and C.(14) below
- 477. Daniel Brown Law PC—effective April 1, 2023 and subject to the terms B.(1) and C.(2) below
- 478. Whistler Transit Ltd.—effective April 2, 2023 and subject to the terms B.(81) and C.(56) below
- 479. Diversified Transportation Ltd.—effective April 2, 2023 and subject to the terms B.(81) and C.(56) below
- 480. Serge Ettinger Law Office Professional Corporation—effective May 1, 2023 and subject to the terms B.(1) and C.(14) below
- 481. Janice Hansen Law Corporation—effective May 1, 2023 and subject to the terms B.(1) and C.(14) below
- 482. The Atkinson Charitable Foundation—effective May 6, 2023 and subject to the terms B.(98) and C.(57) below
- 483. Les Centres D'Acceuil Héritage—effective May 27, 2023 and subject to the terms B.(99) and C.(58) below
- 484. James Gage Professional Corporation—effective June 1, 2023 and subject to the terms B.(1) and C.(14) below
- 485. Ramirez-Joubert Law Professional Corporation—effective from June 1, 2023 to November 15, 2023 and subject to the terms B.(12) and C.(2) below
- 486. Sean D. Parker Professional Corporation—effective June 1, 2023 and subject to the terms B.(12) and C.(2) below
- 487. Toronto Star Children's Charities—effective June 5, 2023 and subject to the terms B.(12) and C.(2) below
- 488. Canadian Nuclear Laboratories—effective June 18, 2023 and subject to the terms B.(100) and C.(48) below
- 489. First Nations Education Authority—effective July 1, 2023 and subject to the terms B.(101) and C.(55) below
- 490. Christine Davies Professional Corporation—effective July 1, 2023 and subject to the terms B.(12) and C.(14) below
- 491. Andrew Loucks Professional Corporation—effective July 1, 2023 and subject to the terms B.(12) and C.(14) below
- 492. Meaghan Adams Legal Professional Corporation—effective July 1, 2023 and subject to the terms B.(102) and C.(14) below
- 493. Mitchell R.P. Barry Professional Corporation—effective August 1, 2023 and subject to the terms B.(103) and C.(14) below
- 494. Tyler Evyn Johnson Professional Corporation—effective August 1, 2023 and subject to the terms B.(104) and C.(14) below

- 495. Clinch Midwifery Professional Corporation—effective September 1, 2023 and subject B.(1) and C.(2) below
- 496. Steven T. Tiainen Law Corporation—effective September 1, 2023 and subject to the terms B.(1) and C.(14) below
- 497. Capacity Canada effective October 2, 2023 and subject to the terms B.(105) and C.(47) below
- 498. Brave New World Data Law Professional Corporation—effective November 1, 2023 and subject to the terms B.(1) and C.(2) below
- 499. Amandeep Hayer Law Corporation—effective November 1, 2023 and subject to the terms B.(106) and C.(2) below
- 500. Namerind Housing Corporation effective December 3, 2023 and subject to the terms B.(107) and C.(55) below
- 501. Aaron Anderson Law Professional Corporation effective January 1, 2024 and subject to the terms B.(108) and C.(2) below
- 502. Community Legal Clinic of York Region effective January 1, 2024 and subject to the terms B.(109) and C.(2) below
- 503. Kyle MacLean Professional Corporation effective January 1, 2024 and subject to the terms B.(20) and C.(14) below
- 504. Cristina Wendel Professional Corporation effective January 1, 2024 and subject to the terms B.(110) and C.(2) below
- 505. Black BurkeMayor LLP effective January 1, 2024 and subject to the terms B.(111) and C.(2) below
- 506. Canadian Bar Association Alberta Branch effective January 1, 2024 and subject to the terms B.(12) and C.(2) below
- 507. Jillian Swartz Professional Corporation effective January 1, 2024 and subject to the terms B.(1) and C.(14) below
- 508. Jonathan Wansbrough Professional Corporation effective January 1, 2024 and subject to the terms B.(1) and C.(14) below
- 509. B. Dawson Enterprises Limited effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 510. Bayers Westwood Daycare Society effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 511. Busy Bee Day Care Co. Ltd. effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 512. Halifax & Region Military Family Resource Centre effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 513. Explore and Discover Early Learning Centre Incorporated effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 514. Through The Years Early Learning Centre effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 515. D. Jean Enterprises Limited effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 516. Foundations Early Learning Centre Ltd. effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 517. Grace Note Child Care Centre (2010) Limited effective January 1, 2024 and subject to the terms B.(112) and C.(59) below

- 518. The Growing Place: Early Education Centre Ltd. effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 519. Hammonds Plains Children's Centre Inc. effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 520. Wentworth Learning Centre Cooperative Ltd. effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 521. Tallahassee Pre School and After School Program Ltd effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 522. Jewels and Gems Day Care Center effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 523. Sunny Valley Preschool effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 524. Windsor Nursery School Society effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 525. Magic Beings Child Care Centre Limited effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 526. Lunenburg Day Care Centre effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 527. Lower Sackville Nursery School effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 528. Splash of Color Children's Center Association effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 529. Lower Onslow Preschool Association effective January 1, 2024 and subject to the terms B.(112) and C.(59) below
- 530. Rhea P. Majewski Law Corporation effective January 1, 2024 and subject to the terms B.(1) and C.(14) below
- 531. Sims Law Corporation effective January 1, 2024 and subject to the terms B.(111) and C.(14) below
- 532. Victory Square Law Office LLP effective January 1, 2024 and subject to the terms B.(111) and C.(2) below
- 533. Alison Longmore Professional Corporation effective February 1, 2024 and subject to the terms B.(12) and C.(14) below
- 534. Centennial College Student Association Inc. effective January 5, 2024 and subject to the terms B.(114) and C.(2) below
- 535. Lutheran Community Care effective January 7, 2024 and subject to the terms B.(115) and C.(60) below
- 536. Culliton Inc. effective January 14, 2024 and subject to the terms B.(115) and C.(61) below
- 537. Wellesley Institute effective January 16, 2024 and subject to the terms B.(116) and C.(62) below
- 538. The Basilian Fathers of Toronto effective January 23, 2024 and subject to the terms B.(126) and C.(63) below
- 539. Goldenberg Law Professional Corporation effective February 1, 2024 and subject to the terms B.(1) and C.(2) below
- 540. J. André Daigle CPC Inc. effective February 1, 2024 and subject to the terms B.(1) and C.(14) below

- 541. John McLuckie Professional Corporation effective February 1, 2024 and subject to the terms B.(1) and C.(2) below
- 542. Kristaq Lala Professional Corporation effective February 1, 2024 and subject to the terms B.(1) and C.(14) below
- 543. The Legal Clinic effective February 1, 2024 and subject to the terms B.(118) and C.(2) below
- 544. Pratt Construction Inc. (operating as Bradley Homes) effective February 24, 2024 and subject to the terms B.123 and C.(64) below
- 545. Unifor Local 324 effective February 24, 2024 and subject to the terms B.(113) and C.(65) below
- 546. Fae Law effective March 1, 2024 and subject to the terms B.(3) and C.(14) below
- 547. Nicholas E. Burke Law Inc. effective March 1, 2024 and subject to the terms B.(12) and C.(14) below
- 548. Shea Coulson Law Corporation effective March 1, 2024 and subject to the terms B.(12) and C.(14) below
- 549. Slice Legal Inc. effective March 1, 2024 and subject to the terms B.(12) and C.(14) below
- 550. Mesiano-Crookston Law P.C. effective April 1, 2024 and subject to the terms B.(1) and C.(14) below
- 551. The Lincoln County Law Association effective May 1, 2024 and subject to the terms B.(12) and C.(8)
- 552. Jesse Harper Professional Corporation effective May 1, 2024 and subject to the terms B.(12) and C.(14) below
- 553. Julia Martin Law Professional Corporation effective May 1 2024 and subject to the terms B.(1) and C.(14) below

494.

Formatted: Normal, No bullets or numbering

B. MEMBER CONTRIBUTIONS OF DESIGN TWO EMPLOYEES

- (1) Effective June 1, 2018, every Member who is a Design Two Employee shall contribute to the Trust Fund in each calendar year 9% of the Member's Annual Earnings unless otherwise specified in this Appendix, provided that the Member contribution amount may not exceed 9% of the Member's Annual Earnings. Any Employer subject to this paragraph B.(1) shall contribute to the Trust Fund in each calendar year 9% of the Member's Annual Earnings.
- (2) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(2) shall, unless otherwise specified in

Appendix H, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	January 1, 2019 to December 31, 2019	January 1, 2020 to December 31, 2020	On and after January 1, 2021
Member Contribution Rate	4%	4.5%	5%
Employer Contribution Rate	4%	4.5%	5%

- (3) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(3) shall contribute to the Trust Fund in each calendar year 7% of the Member's Annual Earnings, with the Employer matching such contribution.
- (4) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(4) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Januar	y 1, January	January	January	On and
2019 to	1, 2020 to	1, 2021 to	1, 2022 to	after
Decemb	er December	December	December	January
31, 2019	31, 2020	31, 2021	31, 2022	1, 2023

Member	Choice of:	Choice of:	Choice of:	Choice of:	
Contribution	1% or 5%	2% or 5%	3% or 5%	4% or 5%	5%
Rate					
Employer	5%	5%	5%	5%	5%
Contribution					
Rate					

(5) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(5) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	July 1, 2019 to March	On and after April 1,
	31, 2020	2020
Member Contribution	4%	5%
Rate		
Employer	4%	5%
Contribution Rate		

(6) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(6) shall, unless otherwise specified in Appendix J, contribute to the Trust Fund in each calendar year the percentage of the Member's applicable earnings as set out in the table below:

Class of		July 1, 2019	July 1,	On and
Membership		to June 30,	2024 to	after July
		2024	June 30,	1, 2025
			2025	
Members	Member	3%	4%	5%
participating	Contribution			
pursuant to	Rate			

Paragraphs C.(4) I or (d) below	Employer Contribution	3%	4%	5%
	Rate			
Members	Member	5%	5%	5%
participating	Contribution			
pursuant to	Rate			
Paragraph C(4) (a)				
or (b) below and				
who were covered	Employer	5%	5%	5%
under Appendix C	Contribution			
of the Postmedia	Rate			
Plan				
Members	Member	4%	4%	5%
participating	Contribution			
pursuant to	Rate			
Paragraph C(4) (a)	Employer	4%	4%	5%
or (b) below and	Contribution			
who were covered	Rate			
under Appendix E				
of the Postmedia				
Plan				
Members	Member	5%	5%	5%
participating	Contribution			
pursuant to	Rate			
Paragraph C(4)(d)				
below, who, if				
hired prior to July	Employer	5%	5%	5%
1, 2019 would	Contribution			
have enrolled in	Rate			
an Employer				

sponsored pension plan on a defined benefit basis				
Members participating pursuant to Paragraph C(4)(a) below who were	Member Contribution Rate	5%	5%	5%
participating in a group registered retirement savings plan sponsored by the Employer	Employer Contribution Rate	5%	5%	5%

Any Member whose Employer is subject to this paragraph B.(6) who is hired on or after July 1, 2019, shall contribute at the member contribution rate corresponding to the class of membership such member would have belonged to had that member been hired prior to July 1, 2019, in accordance with the table above.

Notwithstanding the foregoing, any Member whose Employer is subject to this paragraph B.(6), who as of the July 1, 2019, was in receipt of income replacement pursuant to a long-term disability benefit sponsored by the Employer, and through an agreement between the Employer and Member's collective bargaining agent, was accruing defined contribution pension benefits on a waiver of contributions, shall not be required to contribute to the Plan, but such Employer shall make all employee and employer matching contributions to the Plan for so long as the Member continues to qualify for long-term disability income replacement under

- such Employer's disability benefit policy. Such contributions shall be made in accordance with the table above, as applicable.
- (7) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(7) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Class of membership		Employee Contributio ns	Employer Contributio ns
Licensed Electricians	September 1, 2019 – August 31, 2020	3.0%	3.0%
Apprentices	September 1, 2020 – August 31, 2021	4.0%	4.0%
Labourers Administrative Assistants	September 1, 2021 and thereafter	5.0%	5.0%
Corporate officers	September 1, 2019 and thereafter	5.0%	5.0%

(8) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(8) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which for the purposes of this paragraph B.(8) shall mean regular salary and wages, shift differential, additional earnings for holding a temporary position, cumulative sick leave payments and market adjustments received by the Member, but excluding overtime pay, cost of living bonuses, additional earnings for part-time Employees who are employed beyond their agreed to hiring status, unsociable hours differential or any other type of income received from the Employer) as set out in the table below:

	October 1, 2019 to December 31, 2020	On and after January 1, 2021
Member	7%	7.5%
Contribution Rate		
Employer	7%	7.5%
Contribution Rate		

- (9) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(9) shall, unless otherwise specified in Appendix K or L, contribute to the Trust Fund in each calendar year 6.75% of the Member's Annual Earnings (which for the purposes of this paragraph B.(9) shall mean basic salary or other supplementary income as determined by the Employer, but excluding overtime).
- (10) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(10) shall, unless otherwise specified in Appendix M, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Class of		January 1,	January 1,	On and after
Membership		2020 to	2021 to	January 1,
		December	December	2022
		31, 2020	31, 2021	
Members	Member	4%	5%	6%
employed by	Contribution			
the Employer	Rate			
prior to January	Employer	8%	7%	6%
1, 2020	Contribution			
	Rate			

Members	Member	6%	6%	6%
employed by	Contribution			
the Employer	Rate			
on and after				
on and arter	Employer	6%	6%	6%
January 1, 2020	Contribution			
	Rate			

(11) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(11) shall, unless otherwise specified in Appendix N, contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings (which for the purposes of this paragraph B.(11) shall mean either (a) in respect of non-bargaining Employees, their annual base salary, plus overtime, commission and bonuses, but excluding payments under the Employer's Management Incentive Plan and other fringe benefits and special payments or, (b) in respect of bargaining Employees, any amount paid as employment compensation including overtime, commissions and bonuses), and the Employer shall match such contributions.

Notwithstanding the foregoing, any Member whose Employer is subject to this paragraph B.(11), who as of January 1, 2020, was in receipt of income replacement pursuant to a long-term disability benefit sponsored by the Employer, and was accruing defined contribution pension benefits on a waiver of contributions under a plan sponsored by the Employer, shall not be required to contribute to the Plan. The Employer shall make all employee and employer matching contributions to the Plan for so long as the Member continues to qualify for long-term disability income replacement under such Employer's disability benefit policy, with such contributions being based on such member's Annual Earnings

immediately prior to commencement of such disability leave.

- (12) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(12) shall contribute to the Trust Fund in each calendar year at a rate of 5% of the Member's Annual Earnings, with the Employer matching such contributions.
- (13) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(13) shall, unless otherwise specified under Appendix O, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
January 1, 2020 to	4.0%	7.0%
December 31, 2020		
January 1, 2021 to	5.0%	7.0%
December 31, 2021		
January 1, 2022 to	6.0%	7.0%
December 31, 2022		
On and after	7.0%	7.0%
January 1, 2023		

Notwithstanding the foregoing, any Member whose Employer is subject to this paragraph B.(13) may opt to begin contributions at the maximum contribution rate in accordance with the table above and will receive the corresponding Employer Contribution; and

Any Member whose Employer is subject to this paragraph B.(13) who was participating in a retirement plan sponsored by such Employer shall, on January 1, 2020, begin contributing to the Plan an amount that is the greater of (a) the contribution rate required in accordance with the

applicable table above and (b) the required contribution rate at which the member was contributing to such retirement plan immediately prior to January 1, 2020; and

Any Member whose Employer is subject to this paragraph B.(13) who was hired on or after January 1, 2020 shall begin contributing at the rate of contribution in effect at the time of hire.

(14) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(14) shall, unless otherwise specified under Appendix O, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer
		Contributions
January 1, 2020 to	1.0%	5.0%
December 31, 2020		
January 1, 2021 to	2.0%	5.0%
December 31, 2021		
January 1, 2022 to	3.0%	5.0%
December 31, 2022		
January 1, 2023 to	4.0%	5.0%
December 31, 2023		
On and after	5.0%	5.0%
January 1, 2024		

Notwithstanding the foregoing, any Member whose Employer is subject to this paragraph B.(14) may opt to begin contributions at the maximum contribution rate in accordance with the table above and will receive the corresponding Employer Contribution; and

Formatted: Font: Bold

Any Member whose Employer is subject to this paragraph B.(14) who was participating in a retirement plan sponsored by such Employer shall, on January 1, 2020, begin contributing to the Plan an amount that is the greater of (a) the contribution rate required in accordance with the applicable table above and (b) the required contribution rate at which the member was contributing to such retirement plan immediately prior to January 1, 2020; and

Any Member whose Employer is subject to this paragraph B.(14) who was hired on or after January 1, 2020 shall begin contributing at the rate of contribution in effect at the time of hire.

(15) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(15) shall, unless otherwise specified under Appendix O, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
January 1, 2020 to	1.0%	3.0%
December 31, 2020		
January 1, 2021 to	2.0%	3.5%
December 31, 2021		
January 1, 2022 to	3.0%	4.0%
December 31, 2022		
January 1, 2023 to	4.0%	4.5%
December 31, 2023		
On and after	5.0%	5.0%
January 1, 2024		

Any Member whose Employer is subject to this paragraph B.(15) who was hired on or after January 1, 2020 shall begin contributing at the rate of

contribution in effect at the time of hire.

(16) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(16) shall, unless otherwise specified under Appendix O, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer
		Contributions
January 1, 2020 to	4.0%	4.0%
December 31, 2020		
January 1, 2021 to	4.0%	4.0%
December 31, 2021		
January 1, 2022 to	4.0%	4.0%
December 31, 2022		
January 1, 2023 to	4.5%	4.5%
December 31, 2023		
On and after	5.0%	5.0%
January 1, 2024		

Any Member whose Employer is subject to this paragraph B.(16) who was hired on or after January 1, 2020 shall begin contributing at the rate of contribution in effect at the time of hire.

(17) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(17) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
From March 1, 2020 to January 31, 2021	5.0%	5.0%

Formatted: Font: Bold
Formatted Table

Formatted: Font: Bold

On and after	7.0%	7.0%
February 1, 2021	7.070	7.070

(18) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(18) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
June 1, 2020 to December 31, 2020	5.0%	5.0%
January 1, 2021 to December 31, 2021	6.0%	6.0%
On and after January 1, 2022	6.5%	6.5%

Any Member whose Employer is subject to this paragraph B.(18) who was hired on or after June 1, 2020 shall begin contributing at the rate of contribution in effect at the time of hire.

(19) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(19) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
From June 1, 2020 to October 31, 2020	7.0%	7.0%
On and after November 1, 2020	9.0%	9.0%

(20) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(20) shall contribute to the Trust Fund in each

Formatted Table

- calendar year 6% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (21) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(21) shall contribute to the Trust Fund in each calendar year 5.5% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (22) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(22) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Class of membership	Employee Contributions	Employer Contributions
Non-executive Employees	5.0%	5.0%
Executive Employees	9.0%	9.0%

(23) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(23) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Class of		Member	Employer
Membership		Contributions	Contributions
Employees hired	October 1, 2020 to	2.0%	6.0%
prior to October 1,	September 30, 2021		
2020	October 1, 2021 to	3.0%	6.0%
	September 30, 2022		
	October 1, 2022 to	4.0%	6.0%
	September 30, 2023		

	October 1, 2023 to	5.0%	6.0%
	September 30, 2024		
	On and after	6.0%	6.0%
	October 1, 2024		
Employees hired on	On and after	6.0%	6.0%
or after October 1,	October 1, 2020		
2020			

Notwithstanding the foregoing, any Member whose Employer is subject to this paragraph B.(23) and who was hired prior to October 1, 2020 shall begin contributing to the Plan at the contribution rate in effect at the time that such employee commences participation in the Plan. Any Member whose Employer is subject to this paragraph B.(23) and who was hired prior to October 1, 2020 may opt to begin contributions at the maximum contribution rate in accordance with the table above.

(24) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(24) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
October 1, 2020 to	3.0%	3.0%
September 30, 2021	3.070	3.070
October 1, 2021 to	4.0%	4.0%
September 30, 2022	4.070	4.070
On and after October	5.0%	5.0%
1, 2022	3.076	3.070

(25) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(25) shall contribute to the Trust Fund in each

calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
January 1, 2021 to	3.0%	3.0%
December 31, 2021		
January 1, 2022 to	4.0%	4.0%
December 31, 2022		
January 1, 2023 to	5.0%	5.0%
December 31, 2023		
On and after January	6.0%	6.0%
1, 2024		

(26) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(26) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
January 1, 2021 to	3.0%	3.0%
December 31, 2021		
January 1, 2022 to December 31, 2022	3.5%	3.5%
January 1, 2023 to December 31, 2023	4.0%	4.0%
January 1, 2024 to December 31, 2024	4.5%	4.5%
On and after January 1, 2025	5.0%	5.0%

(27) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(27) shall contribute to the Trust Fund in each

Formatted: Font: Bold
Formatted Table

calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions	
January 1, 2021 to December 31, 2021	6.0%	6.0%	
January 1, 2022 to December 31, 2022	7.5%	7.5%	
On and after January 1, 2023	9.0%	9.0%	

(28) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(28) shall contribute to the Trust Fund in each calendar the percentage of the Member's Annual Earnings (which, for the purposes of this Section B.(28), is defined as base salary, retro pay, acting pay, bilingual bonus, maternity leave top-up, and instructor premium received by a Participating Member, and excludes any and all other monies which are received by a Participating Member from the Employer) as set out in the table below:

	Member Contributions	Employer Contributions
As of January 1, 2021	6.5%	6.5%
to December 31, 2022		
As of January 1, 2023	6.75%	6.75%
to December 31, 2023		
As of January 1, 2024	7.0%	7.0%

(29) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(29) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
February 1, 2021 to December 31, 2021	3.0%	3.0%

Formatted: Font: Bold
Formatted Table

Format	ted: Font: Bold
Format	ted Table
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman
Format	ted: Font: Times New Roman

January 1, 2022 to December 31, 2022	4.0%	4.0%
On and after January 1, 2023	5.0%	5.0%

(30) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(30) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
March 1, 2021 to	3.0%	3.0%
December 31, 2021	3.070	3.070
January 1, 2022 to	3.5%	3.5%
December 31, 2022	3.370	3.370
January 1, 2023 to	4.0%	4.0%
December 31, 2023	1.070	1.070
January 1, 2024 to	4.5%	4.5%
December 31, 2024	1.570	1.570
On and after	5.0%	5.0%
January 1, 2025	3.070	3.070

(31) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(31) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
March 1, 2021 to December 31, 2021	3.0%	3.0%
January 1, 2022 to December 31, 2022	4.0%	4.0%

Formatted: Font: Bold
Formatted Table

January 1, 2023 to	5.0%	5.0%
December 31, 2023	3.070	3.070
On and after	6.0%	6.0%
January 1, 2024	0.070	0.070

(32) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(32) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
March 1, 2021 to	5.0%	5.0%
December 31, 2021	3.070	3.070
January 1, 2022 to	6.0%	6.0%
December 31, 2022	0.070	0.070
January 1, 2023 to	7.0%	7.0%
December 31, 2023	7.070	7.070
January 1, 2024 to	8.0%	8.0%
December 31, 2024	0.070	0.070
On and after	9.0%	9.0%
January 1, 2025	2.070	2.070

(33) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(33) shall, unless otherwise specified under Appendix P, contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings, and the Employer shall match such contributions. Notwithstanding the foregoing, every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(33) who as of April 1, 2021, was in receipt of income replacement pursuant to a long-term disability benefit sponsored by the Employer, shall contribute 5% of the Annual Earnings (which for the purposes of this paragraph B.(33) shall mean the annual remuneration received by the

Member immediately prior to the commencement of such Member's leave).

(34) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(34) shall, unless otherwise specified in Appendix Q, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which for the purposes of this paragraph B.(34) shall mean base salary and does not include any overtime pay bonuses, or any other benefits received from Sanofi Pasteur) as set out in the table below:

	Member Contributions	Employer Contributions	Formatted: Font: Bold
April 1, 2021 to			Formatted Table
April 1, 2021 to	3.0%	15.0%	
December 31, 2021	2.2	22.0.1	
January 1, 2022 to	4.0%	14.0%	
December 31, 2022	4.070	14.070	
On and after	5.0%	13.0%	-
January 1, 2023	3.070	13.070	

Notwithstanding the foregoing, and pursuant to a collective bargaining agreement applicable to such Member, any Design Two Members whose Employer is subject to this paragraph B.(34) and is on income replacement pursuant to a long-term disability benefit shall not be required to contribute to the Plan, but the Employer shall make Employer and Member contributions in accordance with the applicable table above for so long as such member continues to qualify for long-term disability under the Employer's disability policy and the Annual Earnings in respect of such member on which such contributions are based shall be the Annual Earnings received by the Member immediately prior to the commencement of such Member's long-term disability leave.

Additionally, any Design Two Members whose Employer is subject to this paragraph B.(34) and is in receipt of benefits from the Ontario

Workplace Safety Insurance Board or from the Employer's workplace injury policy:

- a. Shall not contribute to the Plan for the first 12 consecutive months
 during which they are in receipt of such benefits, but shall continue
 to accrue a pension based on the Member's Annual Earnings
 immediately prior to commencing receipt of such benefits; and
- b. After such 12 month period, shall make Member contributions to the Plan, with the Employer making corresponding Employer contributions, based on the Member's Annual Earnings immediately prior to commencing receipt of benefits from the Ontario Workplace Safety Insurance Board or the Employer's workplace injury policy.
- (35) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(35) shall contribute to the Trust fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee	Date	Member	Employer
Class		Contributions	Contributions
Legal	April 1, 2021 to	5%	5%
Employees	December 31, 2021:		
	January 1, 2022 to	7%	7%
	December 31, 2022:		
	On and after January 1,	9%	9%
	2023:		
Non-legal	April 1, 2021 to	7%	7%
Employees	December 31, 2021:		
	On and after January 1,	9%	9%
	2022:		

Formatted: Font: Bold

(36) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(36) shall, unless otherwise specified in Appendix S, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which for the purposes of this paragraph B.(36) shall mean (i) for non-bargaining members, the total base salary received by the member from the Employer, including the annual incentives, overtime, vacation pay and commissions but excluding fringe benefits and special payments and (ii) for bargaining members, base salary plus vacation pay, payments under any merit award programs and earned commission only, but excludes any other bonuses, overtime, night differential or statutory holiday differential) as set out in the table below:

	Member Contributions	Employer Contributions	Formatted: Font: Bold
May 1, 2021 to April	4.0%		Formatted Table
30, 2022		6.0%	
On and after May 1, 2022	5.0%	5.0%	

Notwithstanding the foregoing, from the Transfer Date (as that term is defined in Appendix S), until a date that if forty-two months after the Transfer Date, the Employer shall contribute at a rate of 1%, but the Member shall accrue benefits as if both the Employer and Member were contributing at a rate of 5% for such period.

- (37) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(37) shall contribute to the Trust Fund in each calendar year at a rate of 7.5% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (38) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(38) shall contribute to the Trust Fund in each

calendar year the percentage of the Member's Annual Earnings as set out in the table below:

		Contribution Rate	
Employee Class	Date	Member Contributions	Employer Contributions
Management	On and after May 1, 2021	9.0%	9.0%
Supervisor	On and after May 1, 2021	7.0%	7.0%
Operator	May 1, 2021 to December 31, 2021	2.0%	2.0%
	January 1, 2022 to December 31, 2022	4.0%	4.0%
	On and after January 1, 2023	5.0%	5.0%

(39) Every Member who is a Design Two Employee employed prior to June 1, 2021 and whose Employer is subject to this paragraph B.(39) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From June 1, 2021 to	1%	1%
December 31, 2021:		
From January 1, 2022 to	1.5%	1.5%
December 31, 2022:		
From January 1, 2023:to	3%	3%
December 31, 2023		
From January 1, 2024 to	4%	4%
December 31, 2024		

On and after January 1,	5%	5%
2025		

Any other than regular full-time employee of an Employer subject to this B.(39) hired prior to the Effective Date shall begin making contributions to the Plan in accordance with the contribution rate in effect at the time of their enrolment in the Plan. Notwithstanding the foregoing, any employee of an Employer subject to this B.(39) may elect to begin contributing at the maximum contribution rate in accordance with the table above and shall receive the matching Employer Contribution Rate.

Every Member who is a Design Two Employee employed on or after June 1, 2021 and whose Employer is subject to this paragraph B.(39) shall contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings.

(40) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(40) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From June 1, 2021 to May	1.0%	8.25%
31, 2022:		
From June 1, 2022 to May	3.0%	8.25%
31, 2023:		
On and after June 1, 2023	5.0%	8.25%

(41) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(41) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From June 1, 2021 to	3.0%	3.0%
December 31, 2021:		
From January 1, 2022 to	3.5%	3.5%
December 31, 2022:		
January 1, 2023 to	4.0%	4.0%
December 31, 2023:		
January 1, 2024 to	4.5%	4.5%
December 31, 2024:		
On and after January 1,	5.0%	5.0%
2025:		

(42) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(42) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Employer Contributions	Member Contributions	
Executive Employees	7.5%	7.5%	
Non-Executive Employees	5.0%	5.0%	

Formatted: Font: Bold

(43) Every Member who is a Design Two Employee employed prior to June 1, 2021 and whose Employer is subject to this paragraph B.(43) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From June 1, 2021 to	3%	3%
December 31, 2021:		
From January 1, 2022 to	4%	4%
December 31, 2022:		
On and after January 1,	5%	5%
2023:		

Any other than regular full-time employee of an Employer subject to this B.(43) hired prior to June 1, 2021 that enrols after June 1, 2021 shall begin making contributions to the Plan in accordance with the contribution rate in effect at the time of their enrolment in the Plan. Notwithstanding the foregoing, any Design Two Employee of an Employer subject to this B.(43) hired prior to June 1, 2021 may elect to begin contributing at the maximum contribution rate in accordance with the table above and shall receive the matching Employer Contribution Rate.

Every Member who is a Design Two Employee employed on or after June 1, 2021 and whose Employer is subject to this paragraph B.(43) shall contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings.

(44) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(44) shall contribute to the Trust Fund in each calendar 7% of the Member's Annual Earnings, and any Employer subject

- to this paragraph B.(44) shall contribute 9% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (45) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(45) shall, unless otherwise specified in Appendix R, contribute to the Trust Fund in each calendar 7% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (46) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(46) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Employer Contributions	Member Contributions
Legal Employees	9.0%	9.0%
Non-Legal Employees	5.0%	5.0%

(47) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(47) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From July 1, 2021 to December 31, 2021:	7%	3%
From January 1, 2022 to December 31, 2022:	7%	4%

On and after January 1,	7%	5%
2023:		

(48) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(48) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee	Data	vee Contribution Rate		ibution Rate
Class	Date	Member	Employer	
Legal Employees	On and after August 1, 2021:	9.0%	9.0%	
Non-legal Employees	August 1, 2021 to December 31, 2021:	3.0%	3.0%	
	January 1, 2022 to December 31, 2022:	3.5%	3.5%	
	January 1, 2023 to December 31, 2023:	4.0%	4.0%	
	January 1, 2024 to December 31, 2024:	4.5%	4.5%	
	On and after January 1, 2025:	5.0%	5.0%	

(49) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(49) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Employer Contributions	Member Contributions

Legal Employees	7.0%	7.0%
Non-Legal Employees	5.0%	5.0%

(50) Every Member who is a Design Two Employee whose Employer is subject to this paragraph B.(50) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From August 1, 2021 to	5%	5%
December 31, 2021:		
From January 1, 2022 to	6%	6%
December 31, 2022:		
From January 1, 2023 to	7%	7%
December 31, 2023:		
From January 1, 2024 to	8%	8%
December 31, 2024:		
On and after January 1, 2025:	9%	9%

- (51) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(51) shall contribute to the Trust Fund in each calendar year-6% of the Member's Annual Earnings, and any Employer subject to this paragraph B.(51) shall contribute 12% of the Member's Annual Earnings.
- (52) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(52) shall contribute to the Trust Fund in each 150

calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Employer Contributions	Member Contributions
Management Employees	7.0%	7.0%
Non-Management Employees	5.0%	5.0%

- (53) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(53) shall contribute to the Trust Fund in each calendar 5% of the Member's Annual Earnings, and any Employer subject to this paragraph B.(53) shall contribute 13% of the Member's Annual Earnings.
- (54) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(54) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Date	Member	Employer
		Contributions	Contributions
Legal Employees	December 1, 2021 to	8.0%	8.0%
	December 31, 2021:	8.076	8.0%
	On and after January	9.0%	9.0%
	1, 2022:	9.076	9.076
Non-Legal Employees	December 1, 2021 to	4.0%	4.0%
	December 31, 2022:	4.070	4.070
	January 1, 2022 to	4.5%	4.5%
	December 31, 2022:	4.3%	7.570

Formatted: Font: Bold

On and after January 1, 2023:	5.0%	5.0%
-------------------------------	------	------

Formatted: Normal, Indent: Left: 0"

(55) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(55) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Formatted: Normal

Employee Class	Date	Member Contribution Rate	Employer Contribution Rate
Management	On and after January 1,	9.0%	9.0%
Employees	2022:	9.070	9.070
	January 1, 2022 to	3.0%	3.0%
	December 31, 2022:	5.070	3.070
Non-Management	January 1, 2023 to	4.0%	4.0%
Employees	December 31, 2023:	4.070	4.070
	On and after January 1,	5.0%	5.0%
	2024:	5.570	5.070

(56) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(56) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
January 1, 2022 to	3.0%	3.0%
December 31, 2022:		

From January 1, 2023 to	3.5%	3.5%
December 31, 2023:		
January 1, 2024 to	4.0%	4.0%
December 31, 2024:		
January 1, 2025 to	4.5%	4.5%
December 31, 2025:		
On and after January 1,	5.0%	5.0%
2026:		

(57) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(57) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
January 1, 2022 to	3.0%	3.0%
December 31, 2022:		
January 1, 2023 to	4.0%	4.0%
December 31, 2023:		
On and after January 1,	5.0%	5.0%
2024:		

(58) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(58) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which for the purposes of this paragraph B.(58) shall mean basic salary and wages for all hours worker (for greater certainty, "all hours worked" shall include overtime hours) based on regular straight time rate of pay, but shall not

include overtime premiums, bonuses or any other special remuneration) as set out in the table below:

•	Member Contributions	Employer Contributions	Additional Employer Contributions
January 30, 2022 to January 28, 2023:	1.0%	6.0%	0%
January 29, 2023 to February 3, 2024:	2.0%	6.0%	0%
February 4, 2024 to February 1, 2025:	2.5%	6.0%	0.5%
February 2, 2025 to January 31, 2026:	3.0%	6.0%	1.0%
February 1, 2026 to January 30, 2027:	3.5%	6.0%	1.5%
On and after January 31, 2027	5%	6.0%	0%

Notwithstanding the forgoing, at the date of enrolment any Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(58) may elect to begin making Member Contributions at the maximum contribution rate indicated in the table above. If such Member elects to contribute at the maximum contribution rate, the Employer Contribution Rate and the Additional Employer Contribution Rate shall

Formatted: Font: Bold

not be affected by such election and the Employer shall continue to make its contributions in accordance with the table above.

(59) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(59) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From February 1, 2022 to	3.0%	3.0%
December 31, 2022:		
From January 1, 2023 to	3.5%	3.5%
December 31, 2023:		
January 1, 2024 to	4.0%	4.0%
December 31, 2024:		
January 1, 2025 to	4.5%	4.5%
December 31, 2025:		
On and after January 1,	5.0%	5.0%
2026:		

(60) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(60) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

		Member	Employer
Employee Class	Date	Contribution	Contribution
		Rate	Rate
		Rate	Rate

Legal Employees	On and after the Effective Date:	6.0%	6.0%
	January 1, 2022 to December 31, 2022:	3.0%	3.0%
	January 1, 2023 to December 31, 2023:	3.5%	3.5%
Non-Legal Employees	January 1, 2024 to December 31, 2024:	4.0%	4.0%
	January 1, 2025 to December 31, 2025:	4.5%	4.5%
	On and after January 1, 2026:	5.0%	5.0%

(61) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(61) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From April 1, 2022 to December 31, 2022:	3.0%	3.0%
From January 1, 2023 to	3.5%	3.5%
December 31, 2023: January 1, 2024 to	4.0%	4.0%
December 31, 2024: January 1, 2025 to	4.5%	4.5%
December 31, 2025:		

On and after January 1,	5.0%	5.0%
2026:		

(62) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(62) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From April 1, 2022 to December 31, 2022:	3.0%	3.0%
From January 1, 2023 to December 31, 2023:	4.0%	4.0%
January 1, 2024 to December 31, 2024:	5.0%	5.0%

- (63) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(63) shall contribute to the Trust Fund in each calendar year 4% of the Member's Annual Earnings (which for the purposes of this paragraph B.(63) shall mean base salary only and excluding any other benefits, perquisites, and other remuneration). Any Employer who is subject to this paragraph B.(63) shall contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings, as defined in the preceding sentence of this paragraph B.(63).
- (64) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(64) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Member Contributions	Employer Contributions
Senior Executive Employees	6.25%	6.25%
Management Employees	5.0%	6.25%
Pharmacist Staff	5.0%	6.25%
Pharmacy Technicians	3.75%	6.25%
Administrative Staff	3.75%	6.25%

(65) Every Member who is a Design Two Employee who completed their probationary period on or prior to May 16, 2022 and whose Employer is subject to this paragraph B.(65) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(65) is defined as base salary only) as set out in the table below:

	Employer Contributions	Member Contributions
From May 16, 2022 to December 31, 2022:	5.0%	5.0%
From January 1, 2023 to December 31, 2023:	6.0%	6.0%
On and after January 1, 2023	7.0%	7.0%

Notwithstanding the forgoing, at the date of enrolment any Member who is a Design Two Employee who completed their probationary period h on or prior to May 16, 2022 and whose Employer is subject to this paragraph

B.(65) may elect to begin making Member Contributions at the maximum contribution rate indicated in the table above, and shall receive a corresponding Employer contribution.

Any Member who is a Design Two Employee completed their probationary period with the Employer after May 16, 2022 and whose Employer is subject to this paragraph B.(65) shall contribute to the Trust Fund in each calendar year 7% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(65) is defined as base salary only).

(66) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(66) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

<u> </u>	Member Contributions	Employer Contributions
June 1, 2022 to	3.0%	3.0%
December 31, 2022	3.070	3.070
January 1, 2023 to	3.5%	3.5%
December 31, 2023	3.370	3.370
January 1, 2024 to	4.0%	4.0%
December 31, 2024	1.070	1.070
January 1, 2025 to	4.5%	4.5%
December 31, 2025	1.370	1.570
On and after	5.0%	5.0%
January 1, 2026	3.070	5.070

(67) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(67) shall contribute to the Trust Fund in each calendar year 7% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(67), means base salary and bonuses but excluding benefits, perquisites, or any other remuneration), and the

Employer shall match such contributions.

- (68) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(68) shall contribute to the Trust Fund in each calendar year 6.5% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(68), means base salary only, and excluding any other benefits, perquisites, and other remuneration), and the Employer shall match such contributions.
- (69) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(69) may elect to contribute to the Trust Fund at one of the contributions rates set out below in Option 1, 2, 3 or 4, based on the Member's Annual Earnings. If the Member elects Option 1, 2, 3, or 4, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	6.00%	8.25%
Option 2:	7.00%	8.50%
Option 3:	8.00%	8.75%

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(69) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(69) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(70) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(70) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
July 1, 2022 to	3.0%	3.0%
December 31, 2022	3.070	3.070
January 1, 2023 to	3.5%	3.5%
December 31, 2023	3.370	3.370

January 1, 2024 to December 31, 2024	4.0%	4.0%
January 1, 2025 to December 31, 2025	4.5%	4.5%
On and after January 1, 2026	5.0%	5.0%

(71) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(71) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
August 1, 2022 to	3.0%	3.0%
December 31, 2022		
January 1, 2023 to	3.5%	3.5%
December 31, 2023		
January 1, 2024 to	4.0%	4.0%
December 31, 2024		
January 1, 2025 to	4.5%	4.5%
December 31, 2025		
On and after	5.0%	5.0%
January 1, 2026		

(72) Every Member who is a Design Two Employee whose Employer is subject to this paragraph B.(72) shall contribute to the Trust Fund in each calendar year 5% of the Member's Annual Earnings.

Notwithstanding the foregoing, every Member who is a Design Two Employee employed prior to August 28, 2022 and whose Employer is subject to this paragraph B.(72) may elect to contribute to the Trust Fund in each calendar the percentage of the Member's Annual Earnings as set

out in the table below:

	Employer Contributions	Member Contributions
From August 28, 2022 to	1%	1%
December 31, 2022:		
From January 1, 2023 to	2%	2%
December 31, 2023:		
From January 1, 2024 to	3%	3%
December 31, 2024:		
From January 1, 2025 to	4%	4%
December 31, 2025:		
On and after January 1,	5%	5%
2026:		

Any other than regular full-time employee of an Employer subject to this B.(72) hired prior to the August 28, 2022 may elect to contribute in accordance with the table above. If such Member elects to contribute in accordance with the table above, they shall begin making contributions to the Plan in accordance with the contribution rate in effect at the time of their enrolment in the Plan and shall receive the matching Employer Contribution Rate.

(73) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(73) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
September 1, 2022 to	1.5%	5.0%
December 31, 2023	1.3%	3.0%

January 1, 2024 to December 31, 2024	3.0%	6.0%
January 1, 2025 to December 31, 2025	4.5%	6.0%
On and after January 1, 2026	6.0%	6.0%

Any Employee whose Employer is subject to this paragraph B.(73) and who enrolled in the Plan after September 1, 2022 shall begin making contributions to the Plan in accordance with the contribution rate in effect at the time of their enrolment in the Plan.

Notwithstanding the foregoing, any Employee whose Employer is subject to this paragraph B.(73) may elect on their date of enrolment to begin contributing to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions	
September 1, 2022 to December 31, 2023	5.0%	5.0%	
On and after	6.0%	6.0%	
January 1, 2024	0.076	0.076	

(74) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(74) may elect to contribute to the Trust Fund at one of the contributions rates set out below in Option 1, 2, 3, 4 or 5, based on the Member's Annual Earnings (which, for the purposes of the paragraph B.74, is defined as base salary only). If the Member elects Option 1, 2, 3, 4 or 5, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate

shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	5%	5%
Option 2:	6%	6%
Option 3:	7%	7%
Option 4:	8%	8%
Option 5:	9%	9%

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(74) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(74) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(75) Every Member who is a Design Two Employee hired prior to February 24, 2022 and whose Employer is subject to this paragraph B.(75) shall, unless otherwise specified under Appendix V, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(75) is defined as the amount of remuneration paid to the Employees for services rendered by such Employee to the applicable Employer, including regular bonuses and special bonuses but excluding overtime, holiday premium pay (including shift premium and on-call), and any amounts payable by the applicable Employer on behalf of the Employees under any form of employee trust benefit or insurance plan maintained by the Employer under which the Employee participates) as set out in the table below:

A	Member Contributions	Employer Contributions
From September 1,	1.0%	1.0%
2022 to December		
31, 2022:		
From January 1,	2.0%	2.0%
2023 to December		
31, 2023:		
From January 1,	3.0%	3.0%
2024 to December		
31, 2024:		

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman, Bold

Formatted Table

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

From January 1,	4.0%	4.0%
2025 to December		
31, 2025:		
From January 1,	5.0%	5.0%
2026 to December		
31, 2026:		
On and after	6.0%	6.0%
January 1, 2027:		

Any other than regular full-time Member of an Employer who is subject to this paragraph B.(75) who was hired prior to February 24, 2022 but elects to enrol after September 1, 2022 shall contribute to the Plan in accordance with the applicable contribution rate above and shall receive the corresponding Employer contribution rate.

Notwithstanding the foregoing, any Member of an Employer who is subject to this paragraph B.(75) who was hired prior to February 24, 2022 may elect to contribute at the maximum contribution rate in accordance with the table above, and shall receive a corresponding Employer contribution rate.

Any Member of an Employer who is subject to this paragraph B.(75) who was hired on or after February 24, 2022 shall contribute to the Trust Fund in each calendar year 6% of the Member's Annual Earnings, as defined in this first paragraph of this paragraph B.(75).

(76) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(76) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(76) is defined as base salary, pay equity adjustments, and retroactive wage increases, but does not include

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

bonuses, vacation pay, overtime, or any other benefits, perquisites, and other remuneration) as set out in the table below:

Employee Class	Member Contributions	Employer Contributions
Management Employees	5.5%	5.5%
Staff Lawyers	5.5%	5.5%
OPSEU Employees		
covered by the Local 5118-5096 Collective	4.5%	5.5%
Agreement		

(77) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(77) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(77), is defined as base salary only) as set out in the table below:

Employee	Date	Member	Employer
Class		Contributions	Contributions
Non-	October 1, 2022 to	2%	6%
Management	December 31, 2023:		
Employees	From January 1, 2024	3%	6%
	to December 31, 2024:		
	On and after January 1,	4%	6%
	2025:		
Management	October 1, 2022 to	3%	6%
Employees	December 31, 2022:		
	From January 1, 2023	5%	6%
	to December 31, 2024:		
	On and after January 1,	6%	6%
	2025		

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Formatted: Font: Bold

Formatted: Font: Times New Roman
Formatted: Font: Times New Roman
Formatted: Font: Times New Roman

- (78) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(78) shall contribute to the Trust Fund in each calendar year 8% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(78), is defined as basic salary, vacation leave pay, statutory holiday pay, and regular hourly wages, and excluding bonuses paid as part of the Employer's incentive program, special payments, overtime, shift differential and other premium pay, stock options, other allowances and taxable employee benefits, indemnity or reimbursement for expenses, and salary continuance), and the Employer shall contribute 10% of the Member's Annual Earnings, as defined above in this paragraph B.(78).
- (79) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(79) shall contribute to the Trust Fund in each calendar year 3% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(79), is defined as any earnings received from the Employer, including overtime and shift premiums, but does not include any payouts including the Yukon Bonus, Long Service Bonus or Clothing Allowance), and the Employer shall contribute 7% of the Member's Annual Earnings, as defined above in this paragraph B.(79).

If a Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(79) is in receipt of income replacement benefits pursuant to a long-term disability policy sponsored by the Employer or from an applicable workers' compensation board, such Member shall not contribute to the Plan in accordance with Section 4.02(2) or 4.02(3) of Appendix G to the Plan, and the Member shall not receive a corresponding Employer contribution under such leave provisions. The Participating Member may elect to purchase such period on their return to work in accordance with Section 4.03(3) of Appendix G

to the Plan.

For greater clarity, in the event that a Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(79) is in receipt of income replacement benefits pursuant to applicable workers' compensation legislation, but is receiving payments from the Employer and not directly from the workers' compensation board, the Member shall, during such period that they are receiving benefits from the Employer, contribute to the Plan in accordance with Section 4.02(1) of Appendix G to the Plan.

- (80) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(80) shall contribute to the Trust Fund in each calendar year at a rate of 5% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(80), means base salary only, excluding bonuses and any other remuneration), with the Employer matching such contributions.
- (81) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(81) shall contribute to the Trust Fund in each calendar year at a rate of 3% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(81), means base salary only, excluding bonuses and any other remuneration), with the Employer matching such contributions.
- (82) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(82) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(82) is defined as base salary, including on-call pay and acting pay, but excluding other benefits and perquisites) as set out in the table below:

Formatted: Font: Bold

Employee Class	Member Contributions	Employer Contributions	
Members who have			
completed between 0			
to 3 years of service	6.5%	6.5%	
with an Employer	0.370	0.3%	
subject to this			
paragraph B.(82)			
Members who have			
completed between 3			
to 6 years of service	8.0%	8.0%	
with an Employer	8.070	0.070	
subject to this			
paragraph B.(82)			
Members who have			
completed 6 or more			
years of service with	8.0%	10.0%	
an Employer subject to			
this paragraph B.(82)			

Any Member whose Employer is subject to this paragraph B.(82) who, immediately prior to January 1, 2023, was in receipt of benefits from the Employer's long-term disability plan or applicable workers' compensation board, shall be given the option to contribute to the Plan each January 1 for so long as the Member continues to receive disability benefits. If the Member elects to contribute to the Plan, they shall receive a corresponding Employer contribution as set out in the table above. Such contributions shall be based on the Annual Earnings, as defined above in this paragraph B.(82), received by the Member immediately prior to the start of their leave.

(83) Every Member who is a Design Two Employee and whose Employer is subject to this B.(83) shall, unless otherwise specified in Appendix X, contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(83) is defined as base salary, and if applicable, cash utility and housing allowance, or the monetary value of housing furnished by the Employer subject to this B.(83), which shall be deemed to be 30% of the base salary, but does not include bonuses, car allowances, cash allowances and other remuneration) as set out in the table below:

Member Contributions	Employer Contributions
4%	6%

Any Member who is a Design Two Employee and whose Employer is subject to this B.(83) who, as of April 1, 2023, was in receipt of benefits from an Employer's long-term disability plan or from an applicable workers' compensation board and was accruing a benefit under a registered pension plan sponsored by an Employer subject to this B.(83) shall not be required to contribute to the Plan, but the Employer shall continue to make the Employer contribution in accordance with the table above for so long as the Member continues to be in receipt of long-term disability benefits. Contributions shall be based on the Annual Earnings the member received immediately prior to the start of their long-term disability leave.

Any Member who is a Design Two Employee and whose Employer is subject to this B.(83) who commences receipt of benefits from an Employer's long-term disability plan or from an applicable workers' compensation board on or after April 1, 2023 shall not be required to contribute to the Plan, but the Employer shall make Employer contributions and Member contributions in accordance with the table above on behalf of the Member. Such contributions shall be based on the

Formatted: Font: Bold

earnings received by the Member during their period of leave.

If a Member who is a Design Two Employee and whose Employer is subject to this B.(83) is absent on a statutory leave under employment standards legislation application to the Member, the Member's Employer shall continue to contribute Employer contributions to the Plan during such period of statutory leave regardless of whether the Member continues to contribute Member contributions to the Plan during such leave. During such period of statutory leave, the Member may elect to continue contributing Member contributions to the Plan in accordance with the applicable provisions of the Plan, provided that if the Member does not contribute such Member contributions within six months of their return from leave, they shall not be eligible to purchase this period of statutory leave.

For greater certainty, the terms of this paragraph B.(83) shall apply to any statutory leave under applicable employment standards legislation, regardless of whether the applicable employment standards legislation requires the Member to be allowed to continue to accrue a benefit in the Plan for the duration of such leave.

(84) Every Member who is a Design Two Employee of an Employer subject to this B.(84) as of January 1, 2023 and is represented by OPSEU Local 740 shall elect one of the three options set out in the table below and shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(84) is defined as base salary only) based on such election. Such election is a one-time irrevocable election. If a Member does not elect an option, the Member shall be deemed to have elected Option 3. The options are as follows:

A	Contribution Schedule	Member	Employer
		Contribution	Contribution
Option 1	From January 1, 2023 to	3%	5%
	December 31, 2023:		
	From January 1, 2024 to	4%	5%
	December 31, 2024:		
	On and after January 1,	5%	5%
	2025:		
Option 2	From January 1, 2023 to	4%	5%
	December 31, 2023:		
	On and after January 1,	5%	5%
	2024:		
Option 3	On and after January 1,	5%	5%
	2023:		

Any Member who is a Design Two Employee of an Employer subject to this B.(84) hired after January 1, 2023 and who is represented by OPSEU Local 740 or any Member who is a Design Two Employee of an Employer subject to this B.(84) who is non-unionized and is not a senior manager shall contribute 5% of Annual Earnings, as defined above in this paragraph B.(84), and the Employer shall match such contributions. Any Member who is a Design Two Employee of an Employer subject to this B.(84) who is non-unionized and is a senior manager shall contribute 3% of Annual Earnings, as defined above in this paragraph B.(84), and the Employer shall contribute 7.5% of Annual Earnings, as defined above in this paragraph B.(84).

(85) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(85) may elect within the timeline set out by the Employer to contribute to the Trust Fund at one of the contributions rates set out below in Option 1, 2, or 3, based on the Member's Annual

Formatted: Font: Bold
Formatted: Font: Times New Roman, Bold
Formatted: Font: Bold
Formatted: Font: Times New Roman, Bold
Formatted: Font: Times New Roman, Bold
Formatted Table
Formatted: Font: Bold
Formatted: Font: Bold
Formatted: Font: Times New Roman
Formatted: Font: Times New Roman

Earnings (which, for the purposes of this paragraph B.(85), is defined as base salary only). If the Member elects Option 1, 2, or 3, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	3%	3%
Option 2:	6%	6%
Option 3:	9%	9%

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(85) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or

after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(85) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(86) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(86) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

_	Member Contributions	Employer Contributions
January 1, 2023 to December 31, 2023	3.0%	3.0%
January 1, 2024 to December 31, 2024	4.0%	4.0%
On and after January 1, 2025	5.0%	5.0%

(87) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(87) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

A	Member Contributions	Employer Contributions
January 1, 2023 to	3.0%	3.0%
December 31, 2023	3.070	3.070
January 1, 2024 to	3.5%	3.5%
December 31, 2024	2.370	3.370
January 1, 2025 to	4.0%	4.0%
December 31, 2025	4.070	4.070
January 1, 2026 to	4.50/	4.50/
December 31, 2026	4.5%	4.5%

Formatted Table

Formatted: Font: Bold

Formatted: Font: Bold
Formatted Table

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman
Formatted: Font: Times New Roman

On and after	5.0%	5.0%
January 1, 2027	3.070	3.070

(88) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(88) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

A	Member Contributions	Employer Contributions
January 1, 2023 to	5.0%	5.0%
December 31, 2023	3.070	3.070
January 1, 2024 to	7.0%	7.0%
December 31, 2024	7.070	7.070
On and after	9.0%	9.0%
January 1, 2025	7.070	7.070

(89) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(89) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

A	Member Contributions	Employer Contributions
January 1, 2023 to December 31, 2023	3.0%	3.0%
January 1, 2024 to December 31, 2024	4.0%	4.0%
January 1, 2025 to December 31, 2025	5.0%	5.0%
January 1, 2026 to December 31, 2026	6.0%	6.0%
On and after January 1, 2027	7.0%	7.0%

Formatted: Font: Bold
Formatted Table

- (90) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(90) shall contribute to the Trust Fund in each calendar year 3% of the Member's Annual Earnings and the Employer shall match such contribution.
- (91) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(91) may elect to contribute to the Trust Fund at one of the contributions rates set out below in Option 1, 2, or 3, based on the Member's Annual Earnings. If the Member elects Option 1, 2, or 3, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	3%	3%
Option 2:	6%	6%
Option 3:	9%	9%

Effective January 1, 2025 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer

contribution. In the event that the Member fails to make an election under this paragraph B.(91) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(91) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(92)Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(92) may elect to contribute to the Trust Fund at one of the contributions rates set out below in Option 1, 2, or 3, based on the Member's Annual Earnings. If the Member elects Option 1, 2, or 3, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
--	-----------------------------	-------------------------------

Option 1:	6%	6%
Option 2:	7%	7%
Option 3:	8%	8%

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(92) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(92) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(93) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(93) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this B.(93), shall be defined as base salary only) as set out in the table below:

A	Member Contributions	Employer Contributions	5	Formatted: Font: Bold
				Formatted Table

Senior Executive Employees	8.0%	8.0%
Employees who are		
Not Senior Executive	6.0%	6.0%
Employees		

Every Member who is a Design Two Employee and whose Employer is (94)subject to this paragraph B.(94) may elect to contribute to the Trust Fund at one of the contribution rates set out below in Option 1, 2, or 3, based on the Member's Annual Earnings. If the Member elects Option 1, 2, or 3, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	3%	3%
Option 2:	4%	4%
Option 3:	5%	5%

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out

above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(94) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(94) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(95) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(95) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
March 1, 2023 to	3.0%	3.0%
December 31, 2023	3.070	3.070
January 1, 2024 to	3.5%	3.5%
December 31, 2024	3.370	3.570
January 1, 2025 to	4.0%	4.0%
December 31, 2025	4.070	4.070
January 1, 2026 to	4.5%	4.5%
December 31, 2026	4.370	4.570
On and after	5.0%	5.0%
January 1, 2027	3.070	3.070

Formatted: Font: Bold
Formatted Table

- (96) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(96) shall contribute to the Trust Fund in each calendar year 9% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(96), is defined as base salary only), with the Employer matching such contribution.
- (97)Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(97) may elect to contribute to the Trust Fund at one of the contribution rates set out in Option 1, 2 or 3 below, based on the Member's Annual Earnings. If the Member elects Option 1, 2 or 3, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer, the Member shall contribute at the rate set out in Option 3 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 3 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	1%	5%
Option 2:	3%	9%
Option 3:	5%	12%

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Effective January 1, 2024 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(97) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(97) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(98) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(98) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which, for the purposes of this paragraph B.(98) is defined as base salary only) as set out in the table below:

	Employer Contributions	Member Contributions
From May 6, 2023 to	5.0%	5.0%
December 31, 2023:		
From January 1, 2024 to	6.0%	6.0%
December 31, 2024:		

On and after January 1,	7.0%	7.0%
2025		

Notwithstanding the forgoing, at the date of enrolment any Member who is a Design Two Employee whose Employer is subject to this paragraph B.(98) may elect to begin making Member Contributions at the maximum contribution rate indicated in the table above, and shall receive a corresponding Employer contribution.

(99) Every Member who is a Design Two Employee and whose Employer is subject to this B.(99) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Member Contributions	Employer Contributions
ſ	3.0%	4.5%

(100) Every Member who is a Design Two Employee and whose Employer is subject to this B.(100) shall contribute to the Trust Fund in each calendar year 9% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(100) is defined as base pay, lump sum merit payments, continuous shift premium, 40/42 hour work premium, long-term acting pay, continuous on-call pay, responsibility premium, shift turnover premium, and other payments determined in the applicable collective bargaining agreement, but excludes pay out of unused vacation time, overtime pay, other bonuses, commission, allowances, other special remuneration of benefits and cash value of benefits), and the Employer shall match such contributions.

Any Member who commences receipt of long-term disability benefits from an Employer sponsored long-term disability plan shall have the option to contribute to the Plan in accordance with the paragraph above. If the Participating Member elects to contribute to the Plan they will receive a matching Employer contribution. Such contributions shall be made on the Member's Annual Earnings, as defined above in this paragraph B.(100), immediately prior to the start of their long-term disability leave.

(101) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(101) may elect to contribute to the Trust Fund at one of the contribution rates set out in Option 1, 2, 3 or 4 below, based on the Member's Annual Earnings. If the Member elects Option 1, 2, 3 or 4, the Employer shall make the corresponding Employer contribution. In the event that such election is made after enrolment and prior to sixty days thereafter, the elected contribution rate shall come into effect from the date upon which such election is received by the Employer. Until such election is received by the Employer shall contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution. If a Member does not make a valid election within sixty days of the date upon which they enrolled in the Plan, the Participating Member shall continue to contribute at the rate set out in Option 1 below, and the Employer shall make the corresponding Employer contribution.

	Member Contributions	Employer Contributions
Option 1:	3%	3%
Option 2:	5%	5%
Option 2:	7%	7%
Option 4:	9%	9%

Formatted: Font: Times New Roman

Formatted: Font: Times New Roman

Effective January 1, 2025 and each successive January 1 thereafter, the Member may re-elect to contribute at one of the contribution rates set out above. If the Member elects a new contribution rate or confirms an existing rate, the Employer shall make the corresponding Employer contribution. In the event that the Member fails to make an election under this paragraph B.(101) in accordance with the rules and timelines established for such election, the Member and Employer shall continue to contribute at the same rate as in effect immediately prior to the applicable January 1.

Notwithstanding the above, in the event a Member enrols in the Plan on or after September 1 of a given year, such Member shall not be entitled to reelect his or her contribution rate in accordance with this paragraph B.(101) effective as of the immediately following January 1. For greater certainty, the Participating Member will be entitled to elect a new contribution rate or confirm an existing rate effective the next January 1 and each successive January 1 thereafter.

(102) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(102) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From July 1, 2023 to	3.0%	3.0%
December 31, 2023:		
From January 1, 2024 to	4.0%	4.0%
December 31, 2024:		
On and after January 1,	5.0%	5.0%
2025		

- (103) Every Member who is a Design Two Employee and whose Employer is subject to this B.(103) shall contribute to the Trust Fund in each calendar year 8% of the Member's Annual Earnings, and the Employer shall match such contributions.
- (104) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(104) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From August 1, 2023 to December 31, 2023:	8.0%	8.0%
From January 1, 2024 to December 31, 2024:	8.5%	8.5%
On and after January 1, 2025	9.0%	9.0%

Formatted: Normal, Indent: Left: 0"

(105) Every Member who is a Design Two Employee and whose Employer is
subject to this paragraph B.(105) shall contribute to the Trust Fund in each
calendar year the percentage of the Member's Annual Earnings as set out
in the table below:

Contribution Schedule	Employer Contributions	Member
		Contributions
From October 2, 2023, to	<u>3%</u>	<u>3%</u>
<u>September 30, 2024</u>		
From October 1, 2024, to	<u>4%</u>	<u>4%</u>
<u>September 30, 2025</u>		
On and after October 1, 2025	<u>5%</u>	<u>5%</u>

(106) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(106) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From November 1, 2023	5.0%	<u>5.0%</u>
to December 31, 2023:		
From January 1, 2024 to	<u>5.5%</u>	<u>5.5%</u>
December 31, 2024:		
From January 1, 2025 to	<u>6.0%</u>	<u>6.0%</u>
December 31, 2025:		
From January 1, 2026 to	6.5%	<u>6.5%</u>
December 31, 2026:		
From January 1, 2027 to	<u>7.0%</u>	<u>7.0%</u>
<u>December 31, 2027:</u>		
From January 1, 2028 to	<u>7.5%</u>	<u>7.5%</u>
December 31, 2028:		
From January 1, 2029 to	8.0%	8.0%
December 31, 2029:		
From January 1, 2030 to	<u>8.5%</u>	<u>8.5%</u>
December 31, 2030:		
On or after January 1,	9.0%	9.0%
2031		

(107) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(107) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings (which,

for the purposes of this B.(107), is defined as base earnings only) as set out in the table below:

Employee Class	Contribution	Employer	<u>Member</u>
	<u>Schedule</u>	Contributions	Contributions
Executive employees	On and after	<u>9%</u>	<u>9%</u>
	<u>December 3, 2023</u>		
Non-executive employees	On and after	<u>6%</u>	<u>6%</u>
	<u>December 3, 2023</u>		

(108) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(108) shall contribute to the Trust Fund in each calendar year the percentage of the Member's Annual Earnings as set out in the table below:

Employee Class	Employer Contributions	Member Contributions
Legal Employees	6.0%	6.0%
Non-Legal Employees	9.0%	9.0%

- (109) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(109) shall contribute to the Trust Fund in each calendar year 2% of the Member's Annual Earnings, and any Employer subject to this paragraph B.(109) shall contribute 5% of the Member's Annual Earnings.
- (110) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(110) shall contribute to the Trust Fund in each calendar year the percentage of Member's Annual Earnings as set out in the table below:

Employer Contributions	Member Contributions

From January 1, 2024 to	3.0%	3.0%
December 31, 2024:		
From January 1, 2025 to	4.0%	4.0%
December 31, 2025:		
On or after January 1,	<u>5.0%</u>	<u>5.0%</u>
<u>2026:</u>		

(111) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(111) shall contribute to the Trust Fund in each calendar year the percentage of Member's Annual Earnings as set out in the table below:

	Employer Contributions	Member Contributions
From January 1, 2024 to December 31, 2024:	3.0%	3.0%
From January 1, 2025 to December 31, 2025:	3.5%	3.5%
From January 1, 2026 to	4.0%	4.0%
December 31, 2026:		4.70
From January 1, 2027 to December 31, 2027:	<u>4.5%</u>	4.5%
On and after January 1, 2028:	5.0%	5.0%

(112) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(112) shall contribute to the Trust Fund in each calendar year the percentage of Member's Annual Earnings (which, for the purposes of this paragraph B.(112), is defined as regular wages for all hours worked, including retroactive pay and vacation pay, but does not

include overtime pay, bonuses, pay in lieu of vacation, shift premiums or termination pay), as set out in the table below:

	Employer Contributions	Member Contributions
From January 1, 2024 to April 30, 2024:	2.5%	2.5%
On and after May 1, 2024:	5.0%	5.0%

- (113) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(113) shall contribute to the Trust Fund in each calendar year:
 - (a) if the Member is a non-executive employee, 5% of the Member's
 Annual Earnings (which, for the purposes of this paragraph
 B.(113), is defined as base earnings only), and the Employer shall match such contributions;
 - (b) if the Member is an executive employee, 9% of the Member's

 Annual Earnings (as defined in subparagraph (a) above), and the

 Employer shall match such contributions.
- (114) Every Member who is a Design Two Employee and whose Employer is

 subject to this paragraph B.(114) shall contribute to the Trust Fund in each
 calendar year the percentage of the Member's Annual Earnings as set out
 in the table below:

Employee Class	Employer Contributions	Member Contributions
Executive Employees	10.0%	5.0%
Management Employees	6.0%	4.0%
Employees other than	4.0%	1.0%
executive and management		
<u>employees</u>		

- (115) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(115) shall contribute to the Trust Fund in each calendar year 5.0% of the Member's Annual Earnings (which, for the purposes of this paragraph B.(115), is defined as base earnings only), and the Employer shall match such contributions.
- (116) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(116) shall contribute to the Trust Fund in each calendar year 3% of the Member's Annual Earnings and the Employer shall contribute 6% of the Member's Annual Earnings.
- (117) Every Member who is a Design Two Employee and whose Employer is subject to this paragraph B.(117) shall contribute to the Trust Fund in each calendar year 2% of the Member's Annual Earnings, and the Employer shall contribute 8% of the Member's Annual Earnings.
- (118) Every Member who is a Design Two Employee and whose Employer is

 subject to this paragraph B.(118) shall contribute to the Trust Fund in each
 calendar year the percentage of Member's Annual Earnings as set out in
 the table below:

Employee Class	Contribution	Contribution Rate	
	<u>Schedule</u>	Member	Employer
Legal Employees	From February 1, 2024 to December 31, 2024	4.5%	4.5%
	On and after January 1, 2025	<u>5.5%</u>	5.5%

Non-legal	From February 1,	3.0%	3.0%
employees	2024 to	<u>5.070</u>	<u>5.070</u>
employees			
	December 31,		
	2024		
	From January 1,	3.5%	3.5%
	<u>2025 to</u>		
	December 31,		
	<u>2025</u>		
	From January 1,	4.0%	4.0%
	<u>2026 to</u>		
	December 31,		
	<u>2026</u>		
	From January 1,	4.5%	4.5%
	<u>2027 to</u>		
	December 31,		
	<u>2027</u>		
	From January 1,	5.0%	<u>5.0%</u>
	2028 to		
	December 31,		
	<u>2028</u>		
	On and after	5.5%	5.5%
	January 1, 2029		

Formatted: List Paragraph, Indent: Left: 1"

Formatted: Line spacing: 1.5 lines, Don't keep with next

C. PARTICIPATION

(1) Any Design Two Employee of Employers that are subject to this paragraph C.(1) who:

- (a) as of December 31, 2018 was participating in any defined contribution pension plan or group registered retirement saving plan arrangement maintained by such Employers shall cease to participate in such arrangement as of December 31, 2018 and commence participation as a Member of the Plan as of January 1, 2019:
- (b) as of December 31, 2018 was not a participant in any defined contribution pension plan or group registered retirement saving plan arrangement maintained by such Employers will not be required commence participation as a Member of the Plan, but shall have the option on and after January 1, 2019 to join the Plan regardless of whether employed on a regular full-time basis or other than regular full-time basis; or,
- (c) become employed by such an Employer on and after January 1, 2019 shall be eligible to participate as a Member of the Plan provided that those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (2) Any Design Two Employee of Employers that are subject to this paragraph C.(2) shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (3) Any Design Two Employee of Employers that are subject to this paragraph C.(3) who:
 - (a) as of June 30, 2019 was participating in any defined contribution arrangement maintained by such Employer shall cease to participate in such arrangement as of June 30, 2019 and shall

- commence participation as a Member of the Plan as of July 1, 2019;
- (b) as of June 30, 2019 was not a participant in any pension plan maintained by such Employer shall (i) if employed on a regular full-time basis be required to join the Plan as of July 1, 2019 and (ii) if employed on an other than regular full-time basis have the option to join the Plan on and after July 1, 2019; or
- (c) become employed by such an Employer on and after July 1, 2019 shall be eligible to participate as a Member of the Plan provided that those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (4) Any Design Two Employee of Employers that are subject to this paragraph C.(4) who:
 - (a) as of June 30, 2019 was participating in any defined contribution pension plan or group registered retirement savings plan maintained by such Employer shall cease to participate in such arrangement as of June 30, 2019 and shall commence participation as a Member of the Plan as of July 1, 2019;
 - (b) as of June 30, 2019, has suspended active participation in a pension plan sponsored by such Employer shall not be required to join the Plan but shall be permitted to do so on an optional basis;
 - (c) as of June 30, 2019 was not a participant in any defined contribution pension plan maintained by such Employer will not be required commence participation as a Member of the Plan, but shall have the option on and after July 1, 2019 to join the Plan regardless of whether employed on a regular full-time basis or other than regular full-time basis; or

(d) become employed by such an Employer on and after July 1, 2019 shall be eligible to participate as a Member of the Plan provided that those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.

Notwithstanding the foregoing, no employee of the Employers subject to this paragraph C.(4) shall be eligible to join the Plan if any restriction existed as of July 1, 2019 under any applicable collective bargaining agreement or if he or she participates in a pension plan or savings arrangement not sponsored by the Employer.

- (5) Any Design Two Employee of Employers that are subject to this paragraph C.(5) who:
 - (a) as of immediately prior to September 1, 2019 was an active member of the University of Saskatchewan and Federated Colleges Non-Academic Pension Plan is required to join the Plan on the September 1, 2019;
 - (b) is hired by the Employer on and after September 1, 2019 who is a member of CUPE Local 1975 and is:
 - (i) employed in a permanent or seasonal position; or
 - (ii) hired on a term appointment for a period of greater than six months,

and is working at least on a 50% full-time basis, is required to join the Plan on their date of hire; or

- (c) was hired by the Employer prior to September 1, 2019 and who meets the eligibility requirements of (b) above as of September 1, 2019 shall be required to join the Plan on September 1, 2019; and
- (d) For the avoidance of doubt, a person required to join the Plan by virtue of (a), (b) or (c) above shall not cease to earn benefits under the Plan solely by virtue of working less than a 50% full-time basis unless otherwise required commence participation in a registered

- pension plan sponsored by the Employer pursuant to the terms of such other registered pension plan; and
- (e) For the further avoidance of doubt any person employed by the Employer who joins the Plan and subsequently has a change in employment position with the Employer to one which, by virtue of the terms of a registered pension plan sponsored by the Employer, requires that person to commence participation under such other registered pension plan shall cease to have contributions made to the Plan in respect of such person's continued <a href="https://doi.org/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.250/10.2
- (6) Any Design Two Employee of the Employer that is subject to this paragraph C.(6) who:
 - (a) as of October 1, 2019 was employed by the Employer, but was not employed in a full-time executive position, and was not a participant in the Transferring Plan as defined in Appendix K, shall be eligible to join the Plan on the earlier of (i) the date such person would have completed the waiting period under such Transferring Plan as defined in Appendix K and (ii) January 1, 2021, provided that those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan;
 - (b) became employed by the Employer on and after October 1, 2019 but before January 1, 2021, but is not employed in a full-time executive position, shall not be immediately eligible to join the Plan but shall be eligible to join the Plan consistent with the foregoing paragraph (a) on and after January 1, 2021, provided that those employed on a regular full-time basis are required to join the

- Plan and those employed on an other than regular full-time basis have the option to join the Plan;
- (c) became employed by the Employer on and after January 1, 2021, but is not employed in a full-time executive position, shall be eligible to join the Plan provided that those employed on a regular full-time basis are required to join the Plan on hire and those employed on an other than regular full-time basis have the option to join the Plan on and after hire;
- (d) is classified as Relief Staff (which for the purposes of this paragraph C.(6) means relief staff and persons employed for a definite term of task whose employment is, or would have been, covered by, and subject to, the Part-Time Collective Agreement and Letter of Understanding, both made between CUPE Local 2191 and the Employer as of December 16, 2015 and renewed on February 21, 2019) shall not be eligible to join the Plan; or
- (e) was employed with the Employer in a full-time executive position and participated in the Design One Provisions of the Plan, but who is no longer in a full-time executive position with the Employer shall commence participation in the Plan in accordance with Appendix K, paragraph B.(9) of this Appendix A.1 and this paragraph €, and in accordance with section 12.03 of Appendix G.
- (7) Any Design Two Employee of Employers that are subject to this paragraph C.(4) who:
 - (a) as of December 31, 2019 was participating in the defined contribution portion of the pension plan maintained by such Employer shall cease to participate in such arrangement as of December 31, 2019 and shall commence participation as a Member of the Plan as of January 1, 2020;

- (b) become employed by such an Employer on and after January 1,2020 on a full-time basis shall be required to join the Plan; or
- (c) become employed by such an Employer on and after January 1, 2020 on an other than regular full-time basis have the option to join the Plan, however, such employee will be required to join the Plan once the eligibility criteria for other than regular full-time employees set out in the *Pension Benefits Act* (Manitoba) have been met.

Notwithstanding the foregoing, no employee of the Employers subject to this paragraph C.(7) shall be eligible to join the Plan if he or she participates in another pension plan or savings arrangement in which the Employer is a participating Employer.

(1)(8) Any Design Two Employee of Employers that are subject to this paragraph C.(8) and that are employed by such Employer as non-legal staff shall be eligible to participate as Members of the Plan provided employees employed on a regular full-time basis are required to join the Plan and employees employed on an other than regular full-time basis have the option to join the Plan.

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(2)(9) Any Design Two Employee of Employers that are subject to this paragraph C.(9) shall be eligible to participate as Members of the Plan provided that:

(a)a. All regular full-time employees of the Employer who, as of June 1, 2020 are members of the Unifor Local 2002 bargaining unit shall be required to join the Plan;

(b)b. All other than regular full-time employees of the Employer who, as of May 31, 2020, were participating in the defined contribution pension plan maintained by such Employer and, as of June 1, 2020, are members of the Unifor Local 2002 bargaining unit shall be required to join the Plan;

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

- (e)c. All full-time employees of the Employer hired on and after June 1, 2020 who are members of the Unifor Local 2002 bargaining unit shall be required to join the Plan on their date of hire;
- (d)d. All other than regular full-time employees of the Employer who, as of May 31, 2020, were not participating in the defined contribution pension plan maintained by such Employer and, as of June 1, 2020 are members of the Unifor Local 2002 bargaining unit shall have the option to join the Plan; and
- (e)c. All other than regular full-time employees of the Employer hired on and after June 1, 2020 who are members of the Unifor Local 2002 bargaining unit shall have the option to join the Plan.
- (3)(10) Any Design Two Employee of Employers that are subject to this paragraph C.(10) shall be eligible to participate as Members of the Plan provided that:
 - (a)a. All regular full-time employees of the Employer who are members of a Unifor bargaining unit (excluding the Montreal Unifor bargaining unit) or are members of the Teamsters Local 927 bargaining unit and are employed in New Brunswick or Prince Edward Island shall be required to join the Plan;
 - (b)b. All other than regular full-time employees of the Employer who, as of June 30, 2020, were participating in the defined benefit pension plan maintained by such Employer and, as of July 1, 2020, are members of a Unifor bargaining unit (excluding the Montreal Unifor bargaining unit) shall be required to join the Plan;
 - (e)c. All other than regular full-time employees of the Employer who, as of December 30, 2020, were participating in the defined benefit pension plan maintained by such Employer and, as of January 1, 2021, are members of the Teamsters Local 927 and are employed in New Brunswick or Prince Edward Island bargaining unit shall be required to join the Plan;

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

- (d)d. All other than regular full-time employees of the Employer who, as of June 30, 2020, were not participating in the defined benefit pension plan maintained by such Employer and, as of July 1, 2020 are members of a Unifor bargaining unit (excluding the Montreal Unifor bargaining unit) shall have the option to join the Plan;
- (e)c. All other than regular full-time employees of the Employer who, as of December 30, 2020, were not participating in the defined benefit pension plan maintained by such Employer and, as of January 1, 2021 are members of the Teamsters Local 927 bargaining unit and are employed in New Brunswick or Prince Edward Island shall have the option to join the Plan;
- (f)f. All other than regular full-time employees of the Employer hired on and after July 1, 2020 who are members of a Unifor bargaining unit (excluding the Montreal Unifor bargaining unit) shall have the option to join the Plan on and after their date of hire;
- (g)g. All other than regular full-time employees of the Employer hired on and after January 1, 2021 who are members of the Teamsters Local 927 bargaining unit and are employed in New Brunswick or Prince Edward Island shall have the option to join the Plan on and after their date of hire;
- (h)h. All employees of the Employer represented by NAPE and employed in Newfoundland and Labrador who, immediately prior to February 1, 2022, were participating in a pension plan sponsored by the Employer but elected to cease participation in such pension plan shall be required to join the Plan on February 1, 2022;
- (i)i. All employees of the Employer employed in Nova Scotia and represented by Teamsters Local 927 who, immediately prior to February 1, 2022, were participating in the defined benefit component of a pension plan sponsored by the Employer but elected to cease participation in such pension plan shall be required to join the Plan on February 1, 2022;

- (i)j.All employees of the Employer employed in Nova Scotia and represented by Teamsters Local 927 who, immediately prior to February 1, 2022, were participating in the defined contribution component of a pension plan sponsored by the Employer shall be required to join the Plan on February 1, 2022;
- (k)k. All regular full-time employees of the Employer who, as of February 1, 2022, were represented by NAPE and were employed in Newfoundland and Labrador or were employed in Nova Scotia and represented by Teamsters Local 927 and were not participating in a pension plan sponsored by the Employer, shall be required to join the Plan on February 1, 2022;
- (1)LAll other than regular full-time employees of the Employer who, as of February 1, 2022, were represented by NAPE and were employed in Newfoundland and Labrador or were employed in Nova Scotia and represented by Teamsters Local 927 and were not participating in a pension plan sponsored by the Employer, shall have the option to join the Plan;
- (m)m. All regular full-time employees of the Employer represented by NAPE and were employed in Newfoundland and Labrador or were employed in Nova Scotia and represented by Teamsters Local 927 that were hired on or after February 1, 2022 shall be required to join the Plan on their date of hire; and
- (n)n. All other than regular full-time employees of the Employer represented by NAPE and employed in Newfoundland and Labrador or employed in Nova Scotia and represented by Teamsters Local 927 that were hired on or after February 1, 2022 shall have the option to join the Plan.
- (4)(11) Any Design Two Employee of Employers that are subject to this paragraph C.(11) shall be eligible to participate as Members of the Plan provided that:

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(a)a. All full-time employees of the Employer who are not required to participate in the Ontario Teachers' Pension Plan shall be required to join the Plan; and

(b)b. All other than regular full-time employees of the Employer who are not required to participate in the Ontario Teachers' Pension Plan shall have the option to join the Plan.

(5)(12) Any Design Two Employee of Employers that are subject to this paragraph C.(12) shall be eligible to participate as Members of the Plan provided that:

(a)a. All employees of the Employer who work at least twenty-one hours a week as per the terms of their employment agreement shall be required to join the Plan; and

(b)b. All employees of the Employer who work less than twenty-one hours a week as per the terms of their employment agreement shall have the option to join the Plan. Notwithstanding the foregoing, an employee of the Employer who is hired to work less than twenty-one hours a week as per the terms of their employment agreement and is subsequently reclassified by the Employer such that the employee works twenty-one hours or more per week, the employee shall be required to join the Plan upon such reclassification.

(e)c. No employees in a senior executive role immediately prior to October 1, 2020 shall be eligible to join the Plan.

(6)(13) Any Design Two Employee of Employers that are subject to this paragraph C.(13) shall be eligible to participate as Members of the Plan provided that:

(a)a. All employees of the Employer who work at least twenty-one hours a week as per the terms of their employment agreement shall be required to join the Plan; and

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

- (b)b. All employees of the Employer who work less than twenty-one hours a week as per the terms of their employment agreement shall have the option to join the Plan. Notwithstanding the foregoing, an employee of the Employer who is hired to work less than twenty-one hours a week as per the terms of their employment agreement and is subsequently reclassified by the Employer such that the employee works twenty-one hours or more per week, the employee shall be required to join the Plan upon such reclassification.
- (e)c. Any employees of the Employer that are required by legislation to participate in a registered pension plan sponsored by the government of British Columbia shall not be eligible to join the Plan and upon enrolment of such other registered pension plan, contributions to the Plan in respect of such Design Two Member shall cease.
- (7)(14) Any Design Two Employee of Employers that are subject to this paragraph C.(14) and that are employed by such employer as legal staff shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (8)(15) Any Design Two Employee of Employers that are subject to this paragraph C.(15) who are members of Unifor Local 2002 shall be required to participate in the Plan.
- (9)(16) Any Design Two Employee of Employers that are subject to this paragraph C.(16) shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on a part-time basis have the option to join the Plan. Notwithstanding the foregoing, temporary or contracts employees of Employers that are subject to this paragraph C.(16) are not eligible to join the

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Plan.

Any Design Two Employee of Employers that are subject to this paragraph C.(17) shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on a part-time basis have the option to join the Plan. Notwithstanding the foregoing, any employee required to participate in another registered pension plan shall not be eligible to join the Plan.

(11)(18) Any Design Two Employee of Employers that are subject to this paragraph C.(18) shall, provided that such Employee is not represented by a collective bargaining unit, be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on a part-time basis have the option to join the Plan.

(12)(19) Any Design Two Employee of Employers that are subject to this paragraph C.(19) shall, provided that such Employee is not employed in a senior executive role with an employment contract which provides for participation in another retirement savings arrangement, be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.

(13)(20) Any Design Two Employee of Employers that are subject to this paragraph C.(20) who:

(a)a. as of June 30, 2021 was participating in the registered retirement savings plan sponsored by such Employer, shall commence participation in the Plan as of July 1, 2021;

b)b. are employed on a full-time basis shall be required to join the Plan; or

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

- (e)c. __are employed on an other than regular full-time basis have the option to join the Plan.
- Any permanent Design Two Employee of Employers that are subject to this paragraph C.(21) that are represented by the Steelworkers Local 2009 bargaining unit shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on a other than regular full-time basis have the option to join the Plan.
- Any Design Two Employee of Employers that are subject to this paragraph C.(22) that are not members of such Employer's Executive Committee shall be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- Any Design Two Employee of Employers that are subject to this paragraph C.(23) shall, provided that such Employee is employed in the Employer's National Office, Vancouver Office or Ottawa Office, be eligible to participate as Members of the Plan provided those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (17)(24) Any Design Two Employee of Employers that are subject to this paragraph C.(24) and are represented by the Unifor Local 252 bargaining unit shall be required to join the Plan.
- Any Design Two Employee of Employers that are subject to this paragraph C.(25) shall, provided that such Employee is not employed in a senior executive role and participating in another registered pension plan, be eligible to participate as Members of the Plan provided that those employed on a regular full-time basis are required to join the Plan and those employed on an other than regular full-time basis have the option to join the Plan.
- (19)(26) Any Design Two Employee of Employers that are subject to this paragraph C.(26) who:

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(a)a. is represented by the Unifor Local 195 bargaining unit; and

(b)b. has completed their probationary period;

(e)c. shall be eligible to participate in the Plan provided that those employed on a permanent full-time basis shall be required to join the Plan and those employed on a permanent other than regular full-time basis have the option to join the Plan.

(20)(27) Any Design Two Employee of Employers that are subject to this paragraph C.(27) shall be eligible to participate as Members of the Plan provided that:

(a)a. All permanent full-time employees of the Employer shall be required to join the Plan;

(b)b. All permanent part-time employees of the Employer shall have the option to join the Plan; and

(e)c. All non-permanent or contract employees of the Employer shall have the option to join the Plan after such employee has complete two years of continuous employment with the Employer.

(21)(28) Unless otherwise specified in Appendix T, any Design Two Employee of an Employer that is subject to this paragraph C.(28) who:

(a)a. as of March 28, 2022 was employed by the Employer on a regular full-time basis, but was not a participant in the Transferring Plan as defined in Appendix T, shall be eligible to join the Plan on and after March 28, 2022, provided that after two years of employment with the Employer such employee shall be required to join the Plan;

(b)b. is hired by the Employer on or after March 28, 2022 on a regular full-time basis shall be required to join the Plan on their date of hire; or

(c)c. is employed by the Employer on an other than regular full-time basis, but was not a participant in the Transferring Plan as defined in Appendix T, shall have the option to join the Plan.

(22)(29) Any Design Two Employee of an Employer that is subject to this paragraph C.(29) who:

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(a)a. is employed on a full-time basis and has been employed with the employer for 24 months of continuous service shall be required to join the Plan; or

(b)b. is employed on an other than regular full-time basis and has completed:

(i)i.__24 months of continuous service; and

(ii) ii. has received earnings from the Employer in excess of 35% of the Year's Maximum Pensionable Earnings, as defined in the *Canada Pension Plan Act* in two consecutive calendar years shall be required to join the Plan.

For the purposes of this paragraph C.(29), a Design Two Employee shall mean an Employee of an Employer that:

(a) is represented by the Unifor Local 114 bargaining unit;

(e)c. was hired on or after July 1, 2018; and

(d)d. is not participating in another pension plan sponsored by the Employer.

(23)(30) Any Design Two Employee of Employers that are subject to this paragraph C.(30) who:

(a)a. is represented by the CUPE Local 5154 bargaining unit; (b)b. has completed their 120-day probationary period; and shall join the Plan provided that those employed on a permanent full-time basis shall be required to join the Plan and those employed on a permanent other than regular full-time basis have the option to join the Plan.

24)(31) Any Design Two Employee of Employers that are subject to this paragraph C.(31) who is represented by The Canadian Media Guild bargaining unit and is hired on or after July 1, 2022 and who:

(a)a. is employed on a full-time basis shall be required to join the Plan; or

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 6 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 1.63" + Indent at: 1.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

(b)b. is employed on an other than regular full-time basis and has completed:

(i)i. 24 months of continuous service; and

(ii) ii. has received earnings from the Employer in excess of 35% of the Year's Maximum Pensionable Earnings, as defined in the *Canada Pension Plan Act* in two consecutive years, shall be required to join the Plan.

(25)(32) Any Design Two Employee of Employers that are subject to this paragraph C.(32) who:

(a)a. is employed on a full-time basis and is hired on or prior to July 1, 2022 shall be required to join the Plan;

(b)b. is employed on a full-time basis and is hired after July 1, 2022 shall be required to join the Plan on completion of one year of continuous employment with the Employer;

(e)c. is employed on an other than regular full-time basis and is hired on or prior to July 1, 2022 has the option to join the Plan; or

(d)d. is employed on an other than regular full-time basis and is hired after July 1, 2022 shall have the option to join the Plan on completion of one year of continuous employment with the Employer.

(26)(33) Any Design Two Employee of Employers that are subject to this paragraph C.(33) who:

(a)a. is employed on a full-time basis shall be required to join the Plan; or

(b)b. is employed on an other than regular full-time basis shall have the option of joining the Plan; or

(e)c. is employed on a contract or temporary basis shall have the option of joining the Plan upon completion of two years of continuous service with the Employer.

(27)(34) Any Design Two Employee of Employers that are subject to this paragraph C.(34) and are not represented by OPSEU Local 5118 shall be eligible to participate in the Plan provided that:

Formatted: Numbered + Level: 6 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 1.63" + Indent at: 1.75"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(a)a. those employed on a regular full-time basis are required to join the Plan on the later of the Effective Date and their date of hire; and

(b)b. those employed on an other than regular full-time basis have the option to join the Plan upon the completion of twelve months of continuous service with the Employer.

(28)(35) Any Design Two Employee of Employers that are subject to this paragraph C.(35) shall be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis are required to join the Plan upon completion of their three-month probationary period; and

(b)b. those employed on an other than regular full-time basis have the option to join the Plan on completion of 24 months of continuous service with the Employer.

(29)(36) Any Design Two Employee of Employers that are subject to this paragraph C.(36) shall, unless otherwise specified in Appendix W, be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis are required to join the Plan;

(b)b. those employed on an other than regular full-time basis have the option to join the Plan; and

(e)c. those employed on a temporary basis as an intern have the option to join the Plan on completion of 24 months of continuous service with the Employer.

(30)(37) Any Design Two Employee of Employers that are subject to this paragraph C.(37) shall, unless otherwise specified in Appendix V, be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis are required to join the Plan;

(b)b. those employed on an other than regular full-time basis have the option to join the Plan; and

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

(e)c. those employed on a temporary or contract basis have the option to join the Plan on completion of 24 months of continuous service with the Employer.

(31)(38) Any Design Two Employee of the Employers that are subject to this paragraph C.(38) shall be eligible to participate in the Plan provided that:

(a)a. those employed on a regular-full-time basis as of October 1, 2022 shall be required to join the Plan on October 1, 2022;

(b)b. management employees or staff lawyers employed on a regular full-time basis that are hired by the Employer after October 1, 2022 shall be required to join their date of hire;

(e)c. management employees or staff lawyers employed on an other than regular full-time basis shall have the option to join the Plan;

(d)d. OPSEU employees employed on a full-time basis that are covered by the Local 5118-5096 Collective Agreement that are hired by the Employer after October 1, 2022 shall be required to join the Plan on completion of three months continuous employment with the Employer; and

(e)c. OPSEU employees employed on an other than regular full-time basis that are covered by the Local 5118-5096 Collective Agreement shall have the option to join the Plan on completion of three months continuous employment with the Employer.

(32)(39) Any Design Two Employee of Employers that are subject to this paragraph C.(39) shall be eligible to participate in the Plan provided that:

(a)a. __those employed on a regular full-time basis as of October 1, 2022 are required to join the Plan on October 1, 2022;

(b)b. those employed on a regular full-time basis hired after October 1, 2022 are required to join the Plan upon completion of their three-month probationary period; and

(e)c. those employed on an other than regular full-time basis have the option to join the Plan on completion of 24 months of continuous service with the Employer.

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Any Design Two Employee of the Employers that are subject to this paragraph C.(40) shall be eligible to participate in the Plan if such Employee of the Employer is represented by Unifor Local 2301 and was hired on or after January 1, 2020 provided that:

(a)a. those employed on a permanent basis shall be required to join the Plan on the later of October 31, 2022 and the first day of the month immediately following the completion of their probationary period; and

(b)b. those employed on a temporary basis shall be required to join the Plan on the first day of the month immediately following the completion of one of the following within one year of employment with the Employer:

(i)i. worked 700 hours in that year; or

(ii) earned at least 35% of the Year's Maximum Pensionable Earnings, as defined in the Canada Pension Plan Act.

(34)(41) Any Design Two Employee of Employers that are subject to this paragraph C.(41) shall be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis as of November 13, 2022 are required to join the Plan on November 13, 2022;

(b)b. those employed on a regular full-time basis hired after November 13, 2022 are required to join the Plan on the first day of the pay period immediately following the date of completion of their six-month probationary period; and

(e)c. those employed on an other than regular full-time basis have the option to join the Plan on the later of November 13, 2022 and the first day of the pay period immediately following the date of completion 12 months of continuous service with the Employer.

(35)(42) Any Design Two Employee (which, for the purposes of this paragraph C.(42), shall mean a permanent employee of the Employers that are subject to this paragraph C.(42) who is represented by the Public Service

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 6 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 1.63" + Indent at: 1.75"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Indent: First line: 0", Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Alliance of Canada Local Y046) shall be required to join the Plan on the later of December 14, 2022 and the date they become a Design Two Employee.

For greater certainty, if a Member no longer meets the definition of Design Two Employee as set out above in this paragraph C.(42), then such Member shall cease to have contributions made to the Plan during such period of employment with the Employer, provided that for so long as such person remains employed with an Employer who is subject to this paragraph C.(42), such person shall not be considered to have attained their Termination of Membership Date, and shall continue to have any AIW Benefit applied in accordance with the terms of the Plan.

(36)(43) Any Design Two Employee of Employers that are subject to this paragraph C.(43) shall be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis and who were hired prior to January 1, 2023 may elect to commence participation in the Design Two Provisions of the Plan on January 1, 2023 or any date thereafter until December 31, 2024, after which date, such Design Two Employees of Employers that are subject to this paragraph C.(43) shall not be eligible to join the Plan;

(b)b. those employed on a regular full-time basis and who are hired on or after January 1, 2023 shall be required to join the Plan on the date that their probationary period is completed; and

(e)c. those employed on an other than regular full-time basis shall have the option to join the Plan.

Any Design Two Employee of Employers that are subject to this paragraph C.(44) shall be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis shall be required to join the Plan on the later of January 1, 2023 and the date that they have completed their 12-month probationary period; and

(b)b. those employed on an other than regular full-time basis shall have the option to join the Plan on the date they complete 24-months of continuous employment with the Employer.

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Any Design Two Employee (which, for the purposes of this paragraph C.(45), shall mean a non-casual employee of the Employers that are subject to this paragraph C.(45) who is represented by the Society of Professional Engineers and Associates—Scientists and Engineers, the Society of Professional Engineers and Associates—Technicians and Technologists, or the Society of Professional Engineers and Associates—Designers) shall be required to join the Plan on the later of January 1, 2023 and their date of hire.

For greater certainty, if a Design Two Employee whose Employer is subject to this paragraph C.(45) transitions to a position with such Employer that requires them to commence participation in another registered pension plan sponsored by the Employer, then such Design Two Employee shall cease to have contributions made to the Plan, provided that for so long as such person remains employed with an Employer who is subject to this paragraph C.(45), such person shall not be considered to have attained their Termination of Membership Date, and shall continue to have any AIW Benefit applied in accordance with the terms of the Plan.

Any Design Two Employee of Employers that are subject to this paragraph C.(46) shall, unless otherwise specified in Appendix X, be eligible to participate in the Plan provided that:

(a)a. those employed on a regular full-time basis are required to join the Plan on the later of April 1, 2023 and their date of hire;

(b)b. those employed on an other than regular full-time basis whose jurisdiction of employment is British Columbia, Alberta, New Brunswick, or Newfoundland and Labrador shall be required to join the Plan on completion of two years of continuous service with one or more Employers, provided that in each year of service the Design Two Employee has earned not less than 35% of the Year's Maximum Pensionable Earnings, as defined in the *Canada Pension Plan Act*, from their Employer(s); and

(e)c. those employed on an other than regular full-time basis whose jurisdiction of employment is Manitoba, Nova Scotia, Ontario, Prince

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Edward Island, or Saskatchewan, shall be required to join the Plan on completion of two years of continuous service with one or more Employers, provided that in each year of service the Design Two Employee has either:

- (i)i. earned not less than 35% of the Year's Maximum Pensionable Earnings, as defined in the *Canada Pension Plan Act*, from their Employer(s); or
- (ii) ii. worked at least 700 hours with their Employer(s).
- (40)(47) Any Design Two Employee of Employers that are subject to this paragraph C.(47) shall be required to join the Plan.

(41)(48) Any Design Two Employee of Employers that are subject to this paragraph C.(48) and who are represented by the Professional Institute of the Public Service of Canada's Chalk River Professional Employees Group, Whiteshell Professional Employees Group, or Whiteshell Technical Employees Group shall be eligible to participate in the Plan provided that:

- (a)a. Those employed on a full-time basis are required to join the Plan on their date of hire;
- (b)b. Those employed on a permanent part-time basis are required to join the Plan on their date of hire; and
- (e)c. Those employed as a term employees for longer than six months are required to join the Plan on their date of hire.

(42)(49) Any Design Two Employee of Employers that are subject to this paragraph C.(49) shall be eligible to participate in the Plan provided that:

- (a)a. Permanent employees of the Employer represented by OPSEU Local 740 are required to join the Plan on the later of January 1, 2023 and the date that they become a permanent employee represented by OPSEU Local 740;
- (b)b. Full-time non-unionized employees of the Employer hired prior to October 14, 2023 may elect to commence participation in the Plan on

Formatted: Numbered + Level: 6 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 1.63" + Indent at: 1.75"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.75"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

October 14, 2023 or any date thereafter until October 13, 2024, after which date such employees will not be eligible to join the Plan;

- (e)c. Full-time non-unionized employees hired on or after October 14, 2023 shall be required to join the Plan on completion of their three-month probationary period; and
- (d)d. Other than regular full-time non-unionized employees shall have the option to join the Plan on completion of their three-month probationary period.
- (43)(50) Any Design Two Employee of Employers that are subject to this paragraph C.(50) and who are not represented by the Power Workers' Union shall be eligible to participate in the Plan provided that:
 - (a)a. Those employed on a full-time basis shall be required to join the Plan on the later of January 1, 2023 and their date of hire;
 - (b)b. Those employed on an other than regular full-time basis shall have the option of joining the Plan on completion of 24 months of continuous employment with the Employer; and
 - that they are represented by the Power Workers' Union shall cease to have contributions made to the Plan during such period of employment with the Employer; provided that for so long as such person remains employed with the Employer they shall not be considered to have attained their Termination of Membership Date, and shall continue to have any AIW Benefit applied in accordance with the terms of the Plan.
- (44)(51) Any Design Two Employee of Employers that are subject to this paragraph C.(51) shall be eligible to participate in the Plan provided that:
 - (a)a. Those employed on a full-time basis are required to join the Plan on the later of January 1, 2023 and the completion of their three-month probationary period;
 - (b)b. Those hired prior to June 1, 2023 employed on an other than regular full-time basis shall have the option to join the Plan; and

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

(e)c. Those hired on or after June 1, 2023 employed on an other than regular full-time basis shall have the option to join the Plan on completion of 24-months of continuous employment with the Employer.

(45)(52) Any Design Two Employee of Employers that are subject to this paragraph C.(52) shall be eligible to participate in the Plan provided that:

(a)a. Those employed on a full-time basis as of January 1, 2023 are required to join the Plan on January 1, 2023;

(b)b. Those employed on a full-time basis hired after January 1, 2023 are required to join the Plan on completion of their three-month probationary period;

(e)c. Those employed on an other than regular full-time basis shall have the option to join the Plan on completion of 12-months of continuous employment with the Employer.

(46)(53) Any Design Two Employee of Employers that are subject to this paragraph C.(53) and who are non-unionized shall be eligible to participate in the Plan provided that:

(a)a. Those employed on a full-time basis are required to join the Plan;

(b)b. Those employed on an other than regular full-time basis have the option to join the Plan.

(47)(54) Any Design Two Employee of Employers that are subject to this paragraph C.(54) shall be eligible to participate in the Plan provided that:

(a)a. __Those employed on a full-time basis are required to join the Plan on completion of their three-month probationary period;

(b)b. Those employed on an other than regular full-time basis shall have the option to join the Plan on completion of 12-months of continuous employment with the Employer.

(48)(55) Any Design Two Employee of Employers that are subject to this paragraph C.(55) shall be eligible to participate in the Plan provided that:

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

(a)a. Those employed on a full-time basis are required to join the Plan on completion of their three-month probationary period; and

(b)b. Those employed on an other than regular full-time basis shall have the option to join the Plan on completion of their three-month probationary period.

(49)(56) Any Design Two Employee of Employers that are subject to this paragraph C.(56) and who are represented by Unifor Local 114 shall be eligible to participate in the Plan provided that:

(a)a. Those employed on a full-time basis are required to join the Plan on completion of two years of continuous employment with the Employer;

(b)b. Those employed on an other than regular full-time basis shall have the option to join the Plan on completion of two years of continuous employment with the Employer; and

that they are no longer represented by the Unifor Local 114 shall cease to have contributions made to the Plan during such period of employment with the Employer; provided that for so long as such person remains employed with the Employer they shall not be considered to have attained their Termination of Membership Date, and shall continue to have any AIW Benefit applied in accordance with the terms of the Plan.

(50)(57) Any Design Two Employee of Employers that are subject to this paragraph C.(57) shall be eligible to participate in the Plan provided that:

(a)a. Those employed on a full-time basis as of May 6, 2023 are required to join the Plan on May 6, 2023;

(b)b. Those employed on a full-time basis hired after May 6, 2023 are required to join the Plan on completion of their three-month probationary period; and

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

(e)c. Those employed on an other than regular full-time basis shall have the option to join the Plan on completion of their three-month probationary period.

(51)(58) Any Design Two Employee of Employers that are subject to this paragraph C.(58) shall be eligible to participate in the Plan provided that:

(a)a. Full-time employees of the Employer as of May 27, 2023 may elect to commence participation in the Plan on the later of May 27, 2023 and the date they complete their probationary period, or any date thereafter until May27, 2024, after which date such employees will not be eligible to join the Plan;

(b)b. Full-time employees hired after May 27, 2023 shall be required to join the Plan on completion of their probationary period; and

(e)c. Other than regular full-time non-unionized employees shall have the option to join the Plan on completion of 24 months of continuous employment with the Employer.

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 8 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1" + Indent at: 1.25"

Formatted: Bullets and Numbering

- (59) Any Design Two Employee (which for the purposes of any Employers that are subject to this paragraph C.(59), shall mean any employees of the Employer that are early childhood educators or support staff of early childhood educators) shall be eligible to participate as Members of the Plan provided that:
 - a) All permanent and contract Employees as of January 1, 2024 shall commence participation in the Plan on January 1, 2024;
 - b) Any permanent or contract Employees hired by the Employer after

 January 1, 2024 shall commence participation in the Plan on their date
 of hire;
 - c) All casual Employees have the option to commence participation in the Plan on the later of January 1, 2024 and completion of either:
 - i. 700 hours of employment with the Employer; or
 - ii. Receiving earnings from the Employer of not less than 35% of the Year's Maximum Pensionable Earnings, as defined under the Canada Pension Plan;

In each of the two consecutive calendar years immediately prior to joining Design Two.

- (60) Any Design Two Employee of an Employer that is subject to this paragraph C.(60) who:
 - (a) is employed as of January 7, 2024 shall commence participation in the Plan on January 7, 2024;
 - (b) is employed as a full-time employee hired after January 7, 2024 shall commence participation in the Plan as of their date of hire.
 - or hired after January 7, 2024, shall commence participation in the Plan on the earlier of January 7, 2024 and their date of hire;
 - (d) is employed as a temporary, or short-term employee shall have the option to commence participation in the Plan immediately following either:
 - (i) 700 hours of service with the Employer; and

- (ii) receiving not less than 35% of the *Year's Maximum*Pensionable Earnings from the Employer, as defined in the Canada Pension Plan;
 - in each of the two consecutive calendar years prior to joining the Plan.
- (61) Any Employees (which, for the purposes of this paragraph C.(61), shall mean any employees of an Employer subject to this paragraph C.(61) that are:

 (a) not subject to a collective agreement between the Employer and a union;
 and (b) are not participating in a defined benefit pension plan sponsored by
 the Employer) shall be eligible to participate as Members of the Plan provided that:
 - a) All full-time Employees hired by the Employer prior to March 1,
 1993, shall commence participation in the Plan on January 14, 2024;
 - b) All full-time Employees hired by the Employer on or after March 1,
 1993, shall commence participation in the Plan on the later of January
 14, 2024 and the completion of 24 months of continuous employment with the Employer;
 - c) All other than regular full-time Employees shall have the option to commence participation in the Plan on the later of January 14, 2024 and the completion of 24 months of continuous employment with the Employer;
 - d) Notwithstanding a), b), or c) of this paragraph C.(61), an Employee who is a senior HVAC technician holding a G1 license hired by the Employer prior to January 14, 2024 shall have the option to join Design Two at any time on and after January 14, 2024 until January 14, 2025.
- (62) Any Employees (which for the purposes of this paragraph C.(62), shall mean any employees of an Employer subject to this paragraph C.(62) that are:
 (a) special project employees of the Employer hired prior to June 1, 2020 and
 (b) non-special project employees of the Employer) shall be eligible to participate as Members of the Plan provided that:

- a) All full-time Employees shall commence participation in the Plan on the later of January 16, 2024 and completion of their three-month probationary period;
- b) All other than regular full-time Employees shall have the option to commence participation in the Plan on the later of January 16, 2024, and completion of their three-month probationary period.
- (63) Any Employees (which for the purposes of this paragraph C.(63), shall mean any employees of an Employer subject to this paragraph C.(63) who are working in administrative, financial, or operation roles) shall be eligible to participate as Members of the Plan provided that:
 - a) All full-time Employees commence participation in the Plan on the later of January 23, 2024 and the completion of 24 months of continuous employment with the Employer;
 - b) All other than regular full-time Employees shall have the option to commence participation in the Plan on the later of January 23, 2024 and the completion of 24 months of continuous employment with the Employer.
- (64) Any Design Two Employee of Employers that are subject to this paragraph C.(64) shall be eligible to participate in the Plan provided that:
 - a) All full-time Employes hired prior to January 1, 2024 shall commence participation in the Plan as of February 24, 2024; and
 - b) Any full-time Employees hired by the Employer on or after January 1,
 2024, shall commence participation in the Plan on the completion of their 12-month probationary period;
 - c) All other than regular full-time Employees shall have the option to commence participation in the Plan on the later of February 24, 2024 and the completion of 24 months of continuous employment with the Employer.
- (65) Any Employees (which for the purposes of this paragraph C.(65), shall mean executive and non-executive employees of an Employer subject to this

paragraph C.(65)), shall be eligible to participate as Members of the Plan provided that:

- a) All full-time Employes shall commence participation in the Plan on the later of February 24, 2024 and their date of hire;
- b) All other than regular full-time Employees shall have the option to commence participation in the Plan on the later of February 24, 2024 and completion of 24 months of continuous employment with the Employer.

Formatted: Normal, Indent: Left: 0"

APPENDIX B

TRANSFER OF NURSES TO CAAT PENSION PLAN

Notwithstanding any other provisions of the Plan, effective September 1, 1990, hereinafter referred to as the "transfer date", a group of nurses as named by the then Ontario Council of Regents transferred from the Public Service Pension Plan (PSPP) to and became Members of the Plan (the "transferred nurses"). The following provisions shall apply to the transferred nurses:

- (1) Benefits accrued by the transferred nurses under the PSPP shall not be transferred to the Plan.
- (2) Benefits for each member of the transferred nurses group shall be calculated in accordance with the Plan as of the date of retirement, termination of membership or death, as the case may be. For purposes of determining the entitlement to and calculation of benefits for the transferred nurses under the Plan, continuous service and pensionable service recognized under the PSPP shall constitute Continuous Service and Pensionable Service under this Plan, subject to the Act and the Income Tax Act.
- (3) The benefit calculated for each member of the transferred nurses group in accordance with Section (2) above shall be reduced as provided in the Plan by any benefit accrued to the transfer date payable from the PSPP.

Formatted: Highlight

APPENDIX C

MEMBER CONTRIBUTIONS

- (1) Effective January 1, 2004 and prior to January 1, 2008 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 9.1% of the Member's Contributory Earnings below the Year's Basic Exemption for that calendar year, as that term is defined in the Canada Pension Plan, plus
 - (b) 7.3% of the Member's Contributory Earnings above the Year's Basic Exemption up to the YMPE, plus
 - (c) 9.1% of the Member's Contributory Earnings in excess of the YMPE
- (2) Effective January 1, 2008 and prior to January 1, 2009 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 10.1% of the Member's Contributory Earnings below the Year's Basic Exemption for that calendar year, as that term is defined in the Canada Pension Plan, plus
 - (b) 8.3% of the Member's Contributory Earnings above the Year's Basic Exemption up to the YMPE, plus
 - (c) 10.1% of the Member's Contributory Earnings in excess of the YMPE
- (3) Effective January 1, 2009 and prior to January 1, 2010 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 11.1% of the Member's Contributory Earnings below the Year's Basic Exemption for that calendar year, as that term is defined in the Canada Pension Plan, plus
 - (b) 9.3% of the Member's Contributory Earnings above the Year's Basic Exemption up to the YMPE, plus

- (c) 11.1% of the Member's Contributory Earnings in excess of the YMPE
- (4) Effective January 1, 2010 and prior to January 1, 2012 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 12.1% of the Member's Contributory Earnings below the Year's Basic Exemption for that calendar year, as that term is defined in the Canada Pension Plan, plus
 - (b) 10.3% of the Member's Contributory Earnings above the Year's Basic Exemption up to the YMPE, plus
 - (c) 12.1% of the Member's Contributory Earnings in excess of the YMPE
- (5) Effective January 1, 2012 and prior to January 1, 2013 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 11.1% of the Member's Contributory Earnings below the YMPE, plus
 - (b) 12.9% of the Member's Contributory Earnings in excess of the YMPE
- (6) Effective January 1, 2013 and prior to January 1, 2014 every Member shall contribute to the Trust Fund in each calendar year
 - (a) 10.8% of the Member's Contributory Earnings below the YMPE, plus
 - (b) 14.4% of the Member's Contributory Earnings in excess of the YMPE
- (7) Effective January 1, 2014 and prior to January 1, 2019, every Member who is employed by an Employer listed in Appendix A shall contribute to the Trust Fund in each calendar year
 - (a) 11.2% of Member's Contributory Earnings below the YMPE, plus

- (b) 14.8% of the Member's Contributory Earnings in excess of the YMPE.
- (8) Effective January 1, 2019 and prior to January 1, 2025, every Member who is a Full-time Employee shall contribute to the Trust Fund in each calendar year
 - (a) 11.2% of the Member's Contributory Earnings below the YMPE, plus
 - (b) 14.8% of the Member's Contributory Earnings in excess of the YMPE.
- (8.1) Effective January 1, 2025, every Member who is a Full-time Employee shall contribute to the Trust Fund in each calendar year
 - (a) 10.2% of the Member's Contributory Earnings below the YMPE, plus
 - (b) 13.8% of the Member's Contributory Earnings in excess of the YMPE.
- (9) Effective January 1, 2019, every Member who is an Other Than Regular Full-time Employee shall contribute to the Trust Fund in each calendar year 9% of the Member's Annual Earnings.

The following provisions applied for Member current service contributions under Section 4.01 of the Plan as in effect prior to January 1, 2004.

(1) Contributions Prior to July 1, 1992

Prior to July 1, 1992, every Member of the Plan shall contribute to the Trust Fund from the Member's Earnings for the calendar year:

(a) 6% of the part thereof which is below the year's basic exemption as prescribed by the Canada Pension Plan; plus

- (b) 4.2% of the part thereof which is between the year's basic exemption and the YMPE; plus
- (c) 6% of the part thereof which is in excess of the YMPE;

provided that every Member who does not contribute to the Canada Pension Plan, and every Member who continues to contribute to the Trust Fund after attaining 35 years of Pensionable Service, shall contribute to the Trust Fund 6% of Earnings.

(2) Contributions After July 1, 1992 and Prior to January 1, 2004

Effective July 1, 1992 and prior to January 1, 2004, every Member of the Plan shall contribute to the Trust Fund from the Member's Earnings for the calendar year:

- (a) 7.6% of the part thereof which is below the year's basic exemption as prescribed by the Canada Pension Plan; plus
- (b) 5.8% of the part thereof which is between the year's basic exemption and the YMPE as prescribed by the Canada Pension Plan; plus
- (c) 7.6% of the part thereof which is in excess of the YMPE;

provided that every Member who does not contribute to the Canada Pension Plan, and every Member who continues to contribute to the Trust Fund after attaining 35 years of Pensionable Service, shall contribute 7.6% of Earnings.

APPENDIX D

CONTRIBUTORY EARNINGS

- For all Members of the Plan, Contributory Earnings, in accordance with Section 2.11, received by a Member from an Employer, shall also include the elements listed below:
 - (1) Shift premiums
 - (2) Negotiated lump sum settlement arising out of collective bargaining or such equivalent process
 - (3) Recognition allowance or similar lump sum payment to Employees at the top of their salary range or grid
 - (4) Pay equity settlements
 - (5) Lump sum re-earnable merit pay
 - (6) If so elected by the Member, vacation payouts received in respect of the vacation entitlement year applicable to the Member in which the Member's Termination of Employment Date occurs, whether paid in a lump sum or as regular pay
- 2. Contributory Earnings shall include any retroactive pay, provided such retroactive pay would otherwise have met the definition of Contributory Earnings if it had been paid at the time for which such retroactive pay relates.
- 3. For members of the Plan who are employed by a College Employer, as listed in Appendix A of the Plan Text, Contributory Earnings shall include, in addition to the items mentioned above, the following elements:
 - (1) Co-ordinator allowance

- (2) Lead hand premium
- (3) Senior college master allowance.
- 4. For members of the Plan who are employed by a College Employer, as listed in Appendix A of the Plan Text, Contributory Earnings shall include, in addition to the items in 1.01 and 1.02, the following elements, if so elected by full-time employees who were members enrolled prior to January 1, 1989:
 - (1) College benefit plan premiums which are taxable income to the member
 - (2) Bonus for teaching the 11th month
 - (3) Pay for teaching days in excess of the maximum number of contact days.
- 5. Contributory Earnings excludes any payments or elements of remuneration not listed Sections 1, 2, 3 and 4 above unless approved by the Board of Trustees.

APPENDIX E

ROYAL ONTARIO MUSEUM AND FOUNDATION

Effective January 1, 2016, The Royal Ontario Museum (ROM) and The Royal Ontario Museum Foundation (ROM Foundation) became non-college participating employers in the Plan under Appendix A of the Plan following an agreement whereby the assets and liabilities of The Royal Ontario Museum Pension Plan (Registration No. 0469866) (the ROM Plan) in which both the ROM and the ROM Foundation participated, were to be transferred to the Plan effective January 1, 2016 (the Effective Date). This Appendix E and the schedules hereto shall apply with effect on and after the Effective Date in respect of the participation by the ROM and the ROM Foundation in the Plan and the transfer of assets and liabilities from the ROM Plan to the Plan.

- The following words and phrases shall, for the purposes of this Appendix E and the schedules hereto, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Accrued ROM Plan Benefits means the pension benefits accrued by each Active ROM Plan Member under the terms of the ROM Plan immediately prior to the Effective Date; (for ease of reference such accrued benefits have been listed in the attached Schedule E1).
 - b. Active ROM Plan Member means any individual that was an employee of the ROM or ROM Foundation who participated in the ROM Plan immediately prior to the Effective Date and who was not an Inactive ROM Plan Member.
 - Deferred ROM Plan Member means any person, including spouse or beneficiary, entitled to a deferred pension or any lump sum from the ROM Plan immediately prior to the Effective Date.
 - d. **Deferred Pension Entitlement** means the deferred pension entitlement or lump sum payable to each Deferred ROM Plan Member immediately prior to the Effective Date in accordance with the terms of the ROM Plan; (for ease of reference such amounts are listed in the attached Schedule E2).

- e. **Effective Date** has the meaning set forth in the introductory paragraph of this Appendix E.
- f. **Inactive ROM Plan Member** means any individual who is either a Retired ROM Plan Member or a Deferred ROM Plan Member.
- g. Retired ROM Plan Member means any person, including a spouse, in receipt of a pension from the ROM Plan immediately prior to the Effective Date.
- h. **Retired Pension Entitlement** means the monthly pension entitlement payable to each Retired ROM Plan Member in accordance with the terms of the ROM Plan immediately prior the Effective Date and in accordance with the form of pension so elected in respect of the Retired ROM Plan Member (e.g., joint and survivor, 60%, life only, etc.); (for ease of reference such amounts and form of pension are listed in the attached Schedule E3).
- ROM has the meaning set forth in the introductory paragraph of this Appendix E.
- j. **ROM Foundation** has the meaning set forth in the introductory paragraph of this Appendix E.
- k. **ROM Member** means at any given time any employee of the ROM or the ROM Foundation who is an active member of the Plan.
- l. **ROM Plan** has the meaning set forth in the introductory paragraph of this Appendix E and any incorporation by reference into this Appendix E of the terms of the ROM Plan shall be limited to the terms of the ROM Plan (and any amendments thereto) as filed by the Plan pursuant to the Act as a document supporting the Plan.
- m. ROM Non-Plan Member Employee means (i) any employee of the ROM or ROM Foundation as of the Effective Date, other than an Active ROM Plan Member or (ii) any individual who commences employment with the ROM or ROM Foundation after the Effective Date. However, ROM Non-Plan Member Employee excludes anyone in respect of whom the ROM or ROM Foundation is required to contribute to the Ontario Teachers' Pension Plan.
- n. **Transfer Date** means the date on which the transfer of assets from the ROM Plan to the Plan is completed following the approval of the Ontario Superintendent of Financial Services, being December 19, 2016.

Any capitalized terms used in this Appendix E and the schedules hereto that are not defined herein, shall have the meaning ascribed to them in Article 2 of the Plan.

ROM Members

- 2. On and after the Effective Date, each Active ROM Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of the Plan;
 and
 - b. have any service credited to the Active ROM Plan Member under the ROM Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan on and after the Effective Date.
- 3. On and after the Effective Date, each ROM Non-Plan Member Employee shall be eligible to participate in and accrue benefits under the Plan in accordance with the terms of the Plan.
- 4. On and after the Effective Date, ROM Members shall be required to contribute to the Plan in accordance with the terms of the Plan and the ROM, and the ROM Foundation, as applicable, shall match such contributions in accordance with the terms of the Plan.
- 5. Notwithstanding Section 4 of this Appendix E or any other provision of the Plan and in consideration for the transfer of assets to the Plan from the ROM Plan, for the period commencing with the Effective Date and ending December 31, 2018 the contribution rates applicable to:
 - a. ROM Members shall be as set out in the chart below,

	ROM Member Contribution Rates		
	Contributory		
	Earnings below	Contributory Earnings above YMPE	
Effective	YMPE		

January 1, 2016	7.3%	10.5%
January 1, 2017	8.6%	12.2%
January 1, 2018	9.9%	13.5%

and

b. the ROM and the ROM Foundation shall be in accordance with Section 4.07 of the Plan as if ROM Members were contributing to the Plan in accordance with the rates set out in Article 4 of the Plan, as may be amended from time to time.

Accrued ROM Plan Benefits

- 6. As of the Effective Date, the pension benefit entitlement in respect of each Active ROM Plan Member under the ROM Plan as of immediately prior the Effective Date shall be assumed by, and shall be payable from, the Plan in accordance with the terms of the Plan, including as to form of pension and pension payable on a Termination of Membership Date, Normal Retirement Date, Postponed Retirement Date or Early Retirement Date, except as provided for under this Appendix E.
- 7. On and after the Effective Date, the pension benefit entitlement in respect each Active ROM Plan Member shall be:
 - a. The pension benefit accrued to the Active ROM Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix E or otherwise credited to person under the terms of the Plan, plus
 - b. The higher of:
 - The Active ROM Plan Member's Accrued ROM Plan Benefits; and
 - A pension benefit determined under Section 8 of this Appendix E for service credited under the ROM Plan as of December 31, 2015.
- 8. For the purpose of Section 7.b.ii of this Appendix E, the pension benefit for service credited under the ROM Plan as of immediately prior to the Effective Date shall be determined as follows:

- a. For ROM Plan credited service prior to 1990
 - i. 1.0% of the Active ROM Plan Member's Highest Average Pensionable Earnings up to the Average YMPE, multiplied by the years, including part year, of service credited to the Active ROM Plan Member under the ROM Plan immediately prior to the Effective Date for service to December 31, 1989; plus
 - 2.0% of the Member's Highest Average Pensionable Earnings in excess of the Average YMPE, multiplied by the years, including part year, of service credited to the Active ROM Plan Member under the ROM Plan immediately prior to the Effective Date for service to December 31, 1989;

Plus.

- b. For ROM Plan credited service after 1989
 - i. 1.3% of the Active ROM Plan Member's Highest Average Pensionable Earnings up to the Average YMPE, multiplied by the years, including part year, of service credited to the Active ROM Plan Member under the ROM Plan immediately prior to the Effective Date for service on and after January 1, 1990 and prior to December 31, 2015; plus,
 - ii. 2.0% of the Active ROM Plan Member's Highest Average Pensionable Earnings in excess of the Average YMPE, multiplied by the years, including part year, of service credited to the Active ROM Plan Member under the ROM Plan immediately prior to the Effective Date for service on and after January 1, 1990 and prior to December 31, 2015;

Where the Highest Average Pensionable Earnings calculation includes, as applicable, pensionable earnings under the ROM Plan prior to January 1, 2016, and in no event shall it be less than the earnings used to determine the Active ROM Plan Member's Accrued ROM Plan Benefits.

 On the commencement of a pension as of an Early Retirement Date immediately following a Termination of Membership Date, the benefit determined under:

- a. Section 7.a. of this Appendix E shall be reduced in accordance with the terms of the Plan; and
- b. Section 7.b. of this Appendix E shall be reduced as follows:
 - i. accrued pension benefits earned prior to January 1, 2010 under the ROM Plan shall be reduced by 2% per year for each of the first 5 years that the Early Retirement Date precedes the Normal Retirement Date (or the retirement date in Schedule E1 if earlier) and 5% per year for each of the remaining years the Early Retirement Date precedes the Normal Retirement Date (or the retirement date in Schedule E1 if earlier); and
 - ii. accrued pension benefits earned after December 31, 2009 under the ROM Plan shall be reduced by 5% per year for each year early retirement date precedes the Normal Retirement Date (or the retirement date in Schedule E1 if earlier).
- 10. On the death of any Active ROM Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 7.a. of this Appendix E; and
 - b. be in accordance with the terms of the ROM Plan immediately prior to the Effective Date in respect of the benefit determined under Section 7.b. of this Appendix E.
- 11. The pension benefit entitlement determined:
 - a. in accordance with Section 7 of this Appendix E shall, following the commencement of such pension following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, as applicable, be subject only to increase pursuant to Section 14.03 of the Plan but without regard to when such benefit entitlement was accrued;
 - in accordance with Section 7.a. of this Appendix E shall, following a
 Termination of Membership Date and prior to the commencement of such
 pension following a Normal Retirement Date or Early Retirement Date, as
 applicable, be subject only to increase pursuant to Section14.03 of the

Plan but without regard to when such benefit entitlement was accrued; and

c. in accordance with Section 7.b. of this Appendix E shall following a Termination of Membership Date and prior to the commencement of such pension following a Normal Retirement Date or Early Retirement Date, as applicable, not be subject to increase pursuant to Article 14 of the Plan.

Inactive ROM Plan Members

- 12. As of the Effective Date, the Retired Pension Entitlement of each Retired ROM Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix E, and for the avoidance of doubt the provisions of the ROM Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix E.
- 13. As of the Effective Date, the Deferred Pension Entitlement of each Deferred ROM Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix E, and for the avoidance of doubt the provisions of the ROM Plan in effect as of immediately prior to the Effective Date shall apply to determine normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension, except as provided for under this Appendix E.
- 14. The pension payable under Sections 12 and 13 of this Appendix E, once in payment, shall be subject to increase pursuant to Section 14.06 of the Plan regardless of when such benefit entitlement was accrued.
- 15. Notwithstanding Section 14 of this Appendix E, in no event shall the cumulative amount of indexing, if any, provided pursuant to Section 14 of this Appendix E to Retired ROM Plan Members, commencing with the January 1, 2017 indexation (being in respect of calendar year 2016) be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the ROM Plan in effect immediately prior to the Effective Date and commencing on and after January 1, 2017 in respect of calendar year 2016.
- 16. Notwithstanding Section 14 of this Appendix E, in no event shall the cumulative amount of indexing, if any, provided pursuant to Section 14 of this Appendix E to any Deferred ROM Plan Members, commencing on the January 1 of the year following the commencement of pension by the Deferred ROM Plan Member (being in respect of the calendar year in which the pension commenced) be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the ROM Plan in effect immediately prior to the Effective Date

and commencing on and after January 1 of the year following the commencement of pension by a Deferred ROM Plan member in respect of the prior calendar year.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active ROM Plan Member:
 - i. to the extent to which payment of such benefit has been made from the ROM Plan prior the Transfer Date; or
 - ii. under sections 6 or 7.b. of this Schedule E in respect of whom the ROM Plan has ceased to have any liability prior to the Transfer Date;
 - b. in respect of any Retired ROM Plan Member:
 - to the extent to which payment has been made from the ROM Plan prior to the Transfer Date in respect of such Retired ROM Plan Member; or
 - ii. in respect of whom the ROM Plan has ceased to have any liability prior to the Transfer Date; or
 - c. in respect of any Deferred ROM Plan Member:
 - to the extent to which payment has been made from the ROM Plan prior to the Transfer Date in respect of such Deferred ROM Plan Member; or
 - ii. in respect of whom the ROM Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX F

YOUTH SERVICES BUREAU OF OTTAWA

Effective January 1, 2018, Youth Services Bureau of Ottawa ("YSB") became a non-college participating employer in the Plan under an agreement whereby the assets and liabilities of the Retirement Plan for Employees of Youth Services Bureau of Ottawa (Registration No. 0362558) (the "YSB Plan") are to be transferred to the Plan effective January 1, 2018. Pending approval of the transfer of such assets and liabilities, the following provisions shall apply in respect of the participation by YSB in the Plan:

- 2-1. The following words and phrases shall, for the purposes of this Appendix F, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active YSB Plan Member means any individual that is an employee of YSB who participated in the YSB Plan immediately prior to the Effective Date and who is not an Inactive YSB Plan Member.
 - **b. Inactive YSB Plan Member** means any individual who is either a Retired YSB Plan Member or a Deferred YSB Plan Member.
 - c. Deferred YSB Plan Member means any person entitled to a deferred pension or any lump sum from the YSB Plan as at the Effective Date.
 - d. Effective Date means January 1, 2018.
 - e. Retired YSB Plan Member means any person in receipt of a pension from the YSB Plan as at the Effective Date.
 - f. YSB Member means at any given time any employee of YSB who is an active member of the Plan.
 - g. YSB Non-Plan Member Employee means (i) any employee of YSB as of the Effective Date, other than an Active YSB Plan Member or (ii) any individual who commences employment with YSB after the Effective

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Date.

3.2.On and after the Effective Date, each Active YSB Plan Member shall:

- a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of the Plan; and
- b. have any service credited to the Active YSB Plan Member under the YSB Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan on and after the Effective Date.
- 4.3. On and after the Effective Date, each YSB Non-Plan Member Employee shall be eligible to participate in and accrue benefits under the Plan in accordance with the terms of the Plan.
- 5.4.On and after the Effective Date, YSB Members shall be required to contribute to the Plan in accordance with the terms of the Plan and YSB shall match such contributions in accordance with the terms of the Plan.
- 6-5. Notwithstanding Section 4 of this Appendix F or any other provision of the Plan and in consideration for the transfer of assets to the Plan from the YSB Plan, for the period commencing with the Effective Date and ending December 31, 2021 the contribution rates applicable to Contributory Earnings of YSB Members who were Active YSB Plan Members that were active members of the YSB Plan on March 31, 2017 shall:
 - a. during the 12 months commencing on the Effective Date, be equal to the contribution rates applicable during such time period as provided for in Article 4 of the Plan less 420 basis points on Contributory Earnings up to the YMPE and less 580 basis points on Contributory Earnings above the YMPE;
 - b. during the 12 months commencing on January 1, 2019, be equal to the contribution rates applicable during such time period as provided for in Article 4 of the Plan less 340 basis points on Contributory Earnings up to the YMPE and less 460 basis points on Contributory Earnings above the

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

YMPE;

- c. during the 12 months commencing on January 1, 2020, be equal to the contribution rates applicable during such time period as provided for in Article 4 of the Plan less 160 basis points on Contributory Earnings up to the YMPE and less 240 basis points on Contributory Earnings above the YMPE; and
- d. during the 12 months commencing on January 1, 2021, be equal to the contribution rates applicable during such time period as provided for in Article 4 of the Plan less 80 basis points on Contributory Earnings up to the YMPE and less 120 basis points on Contributory Earnings above the YMPE.

And YSB shall make contributions in respect of such YSB Members as identified in this Section 5 of Appendix F in accordance with Article 4 of the Plan as if such YSB Members identified in this Section 5 of Appendix F were contributing to the Plan in accordance with the rates provided for in Article 4 of the Plan.

APPENDIX G – DESIGN TWO PROVISIONS

Article 1. APPLICATION

Section 1.01 Application of Appendix G

This Appendix G is added to the Plan effective June 1, 2018.

This Appendix G applies to Design Two Employees who become Members on or after June 1, 2018. Effective June 1, 2018, the benefits, rights and obligations of Design Two Employees shall be in accordance with the Design Two Provisions.

Effective January 1, 2019, this Appendix G applies to Other Than Regular Full-time Employees who were Members on December 31, 2018. In addition, this Appendix G applies to Other Than Regular Full-time Employees who become Members on or after January 1, 2019.

Effective January 1, 2019, the benefits, rights and obligations of Other Than Regular Full-time Employees shall be in accordance with the Design Two Provisions.

Section 1.02 **Application of the Design One Provisions**

In addition to the provisions of this Appendix G, the following provisions of the Design One Provisions shall also apply to a Member who accrues benefits under this Appendix G:

- (a) Article 1;
- (b) Article 12; and
- (c) Articles 15, 16, 17, and 18, provided, however, that the reference to Section 4.05 in Article 18 of the Design One Provisions shall be read as a reference to Section 4.05 of this Appendix G.

Article 2. DEFINITIONS

Section 2.01 In this Appendix G, unless the context clearly indicates otherwise, the following terms shall have the following meanings:

- (a) "<u>AIW Benefit</u>" shall mean the greater of 0 and the Member's AIW Benefit determined in accordance with Schedule 1 to this Appendix G.
- (b) "Annual Earnings" shall mean all wages, salary and other amounts received by a Member from an Employer in respect of their employment with that Employer on and after June 1, 2018 as a Design Two Employee or on or after January 1, 2019 as an Other Than Regular Full-time Employee, but excluding:
 - any employee benefits, perquisites or allowances whether or not such benefits, perquisites or allowances are taxable under the Income Tax Act;
 - (2) such amounts as are determined by the Board of Trustees or such amounts as are specified in an Appendix.
- (c) "Base Benefit" shall mean, in relation to a Member, the sum of:
 - (1) the product of 8.5% multiplied by the sum of the Member contributions and Employer contributions remitted to the Plan under the Design Two Provisions on behalf of that Member on and after June 1, 2018, and in respect of a period prior to January 1, 2025, plus
 - (2) the product of 9.5% multiplied by the sum of the Member contributions and Employer contribution remitted to the Plan under the Design Two Provisions on behalf of that Member in respect of a period on and after January 1, 2025.

Notwithstanding paragraphs (1) and (2) above, in determining the Base Benefit, the following shall apply:

- (i) where contributions remitted to the Plan under Section 4.01, 4.02 or 4.05 on or after January 1, 2025 relate to a period prior to January 1, 2025, such contributions shall be multiplied by 9.5%;
- (ii) contributions in respect of any period of Purchasable Service referenced in Sections 4.03(1) and 4.03(4) shall be multiplied by 8.5%; and
- (iii) Member contributions in respect of Section 4.03 are subject to any adjustment determined by the Board of Trustees at the time of purchase, and shall not be included in determining a Member's Base Benefit until such contributions are paid to the Trust Fund in accordance with Section 4.03.
- (d) "Continuous Service" shall mean Continuous Service as defined in the Design One Provisions, provided that the references to Section 10.02 therein shall be replaced with references to Section 10.02 of this Appendix G.
- (e) "<u>Deemed Annual Earnings</u>" shall mean the amount of Annual Earnings a Member would have received during a leave, work stoppage or other absence from work, as applicable, which amount shall be based on the Member's rate of Annual Earnings in the period immediately preceding such leave, work stoppage or other absence, adjusted from time to time, if applicable to the Member, by any increases in those elements of Annual Earnings as a result of:
 - (1) negotiated increases under a collective bargaining agreement;
 - (2) increases resulting from normal progression through the earnings classification; or,
 - (3) increases obtained through the grievance arbitration procedure under a collective bargaining agreement or through a similar mechanism.

- (f) "<u>Deferred Pension</u>" shall mean the pension benefit determined pursuant to Article10 and as may be adjusted pursuant to Article 11.
- (g) "Early Retirement Date" has the meaning set forth in Section 7.01.
- (h) "<u>Early Retirement Pension</u>" shall mean the pension benefit determined pursuant to Section 7.02.
- (i) "Excess Contributions" shall mean the excess of (1) over (2) as follows:
 - the Member's contributions made in accordance with Article 4 plus
 Credited Interest to the date a determination is required;
 - (2) 50% of the Commuted Value of the Member's pension benefit in respect of all Pensionable Service, calculated as of the same date as in subsection (1);

provided that the Member contributions in (1) above and the Member's pension benefit in (2) above will not apply to benefits resulting from the election by a Member and to the Member's contributions in accordance with Section 4.03 for periods of Purchasable Service in respect of which the Employer makes no contributions pursuant to Section 4.05 in respect of the Member.

- (j) "Former Member" shall mean:
 - a person who has attained their Termination of Membership Date and who is entitled to a benefit under the Plan, excluding Retired Members; or
 - (2) except for the purposes of Article 11, a person who is a Former Member as defined under the Design One Provisions.

- (k) "<u>Normal Retirement Pension</u>" shall mean the pension benefit provided in Section 6.01 as of a Normal Retirement Date.
- (l) "Pensionable Service" shall be determined in accordance with the provisions of Article 5.
- (m) "Postponed Retirement Date" has the meaning set forth in Section 8.01.
- (n) "Postponed Retirement Pension" has the meaning set forth in Section 8.02.
- (o) "Retired Member" shall mean:
 - (1) a person who has either attained their Termination of Employment Date or Termination of Membership Date and who
 - (a) is receiving or has elected in accordance with the Plan to receive a Normal Retirement Pension; a Postponed Retirement Pension; an Early Retirement Pension; or a pension determined in accordance with Section 10.03; or
 - (b) has attained their Normal Retirement Date; or
 - (2) except for the purposes of Article 11, a person who is a Retired Member as defined under the Design One Provisions.
- (p) "Termination of Membership Date" shall be as defined in the Design One Provisions, provided that the reference to Section 10.02 therein shall be replaced with a reference to Section 10.02 of this Appendix G and the reference to Section 9.08 therein shall be replaced with a reference to Section 9.08 of the Design One Provisions as modified in accordance with this Appendix G.

In addition to the terms defined in this Appendix G, any terms which are defined in Article 2 of the Design One Provisions and which are used in capitalized form in this Appendix G shall have the same meaning as set out in Article 2 of the Design One Provisions. However, if such a definition in Article 2 of the Design One Provisions uses a

term defined in this Appendix G and uses that term in the capitalized form, in such definition such term shall have the meaning as set out in this Appendix G. Unless otherwise indicated, a reference in this Appendix G to a provision, Section, or Article is a reference to a provision, Section, or Article of Appendix G, provided, however, that a reference in this Appendix G to Article 9 or a Section of Article 9 is a reference to Article 9 or a Section of Article 9 of the Design One Provisions, as applicable, subject to any modifications thereof in accordance with the Design Two Provisions.

Article 3. ELIGIBILITY AND MEMBERSHIP IN THE PLAN

Section 3.01 Continuation of Membership

Each Other Than Regular Full-time Employee who was a Member of the Plan on December 31, 2018 shall remain a Member of the Plan until their Termination of Membership Date and, subject to Section 3.04, shall be entitled to accrue benefits under the Plan after December 31, 2018 only under this Appendix G.

Section 3.02 Eligibility for Membership

(1) Other Than Regular Full-time Employees

Each Other Than Regular Full-time Employee other than one referenced in Section 3.01 may elect to become a Member of the Plan at any time on or after the Employee's date of hire, provided the Employee's election date is prior to the end of the year in which the Employee reaches age 71, or such other age as is specified in the Income Tax Act (Canada).

(2) Design Two Employees

Unless otherwise specified in an Appendix, each Design Two Employee shall become a Member of the Plan upon the Employee's date of hire, provided the Employee's date of hire is prior to the end of the year in which the Employee reaches age 71, or such other age as is specified in the Income Tax Act (Canada).

(3) Excluded Employees

An Other Than Regular Full-time Employee or Design Two Employee who is participating and accruing benefits under any pension plan as designated by the Sponsors' Committee is not eligible to become a Member of the Plan.

Section 3.03 Continuation of Membership

A person who becomes a Member of the Plan shall remain a Member until their Termination of Membership Date.

Section 3.04 Full-time Employees

If a Member is an Other Than Regular Full-time Employee with one Employer (the "OTRFT Employer") or a Design Two Employee with one Employer (the "Design Two Employer") and is concurrently employed by another Employer as a Full-time Employee (the "Full-Time Employer"), no contributions will be made in respect of employment with any OTRFT Employer or Design Two Employer as long as the Employee continues to be recognized as a Full-time Employee with the Full-Time Employer.

Section 3.05 Membership and Benefit Determination Following Re-employment

(1) Membership on Re-employment Prior to Pension Commencement

(a) Prior Member With No Benefit Entitlement

A person who is neither a Former Member nor a Retired Member, but who was previously a Member of the Plan, and who is hired as an Other Than Regular Full-time Employee or as a Design Two Employee will be considered a new Employee for all purposes of the Plan and the provisions of Section 3.02 shall apply.

(b) Former Member

If a Former Member is hired as an Other Than Regular Full-time Employee or as a Design Two Employee, the Former Member shall become a Member immediately upon hire and any Deferred Pension in respect of such Member's prior period of membership under the Design Two Provisions will not be affected by such Member's subsequent membership. Except as provided herein, any pension benefit to which the Member may be entitled under the Design One Provisions at the Member's subsequent Termination of Membership Date will not be affected by such Member's subsequent membership.

(2) Membership on Re-employment After Pension Commencement

If a Retired Member is hired as an Other Than Regular Full-time Employee or a Design Two Employee, the Retired Member may elect either

- A. to continue to receive pension payments and not become a Member of the Plan, or
- B. to become a Member of the Plan, in which case pension payments will cease immediately and the Retired Member will become a Member and recommence contributions under Section 4.01 immediately on rehire and any pension in respect of such Member's prior period of membership under the Design Two Provisions will not be affected by his subsequent period of membership and except as provided herein, any pension benefit to which the Member may be entitled under the Design One Provisions at the Member's subsequent Termination

of Membership Date will not be affected by such Member's subsequent membership.

If the Retired Member is hired by an Employer after the end of the year in which the Retired Member reaches age 71, or such other age as is specified in the Income Tax Act (Canada), the Retired Member will continue to receive pension payments and shall not become a Member of the Plan.

(3) Determination of Pension Benefits for Re-employed Members

(a) Re-employed Former Members

If a Former Member is hired as an Other Than Regular Full-time Employee or a Design Two Employee, the benefit to which the Member will be entitled under the Design Two Provisions at the Member's subsequent Termination of Membership Date will be the sum of:

- the Deferred Pension to which the Member is entitled in respect of each prior period of membership under the Design Two Provisions, and
- (ii) the pension benefit determined under the Design Two Provisions based on the contributions made under Article4 for the Member's most recent period of membership under the Design Two Provisions.

(b) Re-Employed Retired Members

If a Retired Member is hired as an Other Than Regular Full-time Employee or a Design Two Employee and elects under Section 3.05(2) to become a Member of the Plan, the benefit to which the Member will be entitled under the Design Two Provisions at the Member's subsequent Termination of Membership Date will be the sum of:

- (i) the pension in pay at the time the Member was re-employed and rejoined the Plan, plus any inflation protection that would have been received during the period of reemployment, adjusted to what the Retired Member would have been receiving under the normal form of pension in the event they elected an optional form; and
- (ii) the pension benefit determined under the Design Two Provisions based on the contributions made under Article 4 for the Member's most recent period of membership under the Design Two Provisions.

Section 3.06 Re-employment Following Grievance

If a person who is neither a Former Member nor a Retired Member but who was previously a Member, has their employment as an Other Than Regular Full-time Employee or Design Two Employee reinstated as a result of a grievance or other legal proceeding, the Member shall recommence contributions under Article 4 as of the effective date of the reinstatement and, following any determination of a past service pension adjustment under the Income Tax Act, shall make contributions on any retroactive salary, wages, or other amounts received as a result of the reinstatement of employment, as determined by the Board of Trustees, and the Member may either:

(1) elect to repay any Commuted Value plus any Excess Contributions or contribution refund, with Credited Interest from the date of payment to the date of repayment in which case, the benefit to which the Member will be entitled at the Member's subsequent Termination of Membership Date will be determined without regard to the interruption in the Member's employment, provided, however, that if the amount to be repaid under this Section 3.06 is in respect of pre-1992 employment, the amount being repaid must be transferred directly from another registered pension plan on

- behalf of the Member or from a registered retirement savings arrangement of which the Member is the annuitant; or
- (2) elect not to repay the Commuted Value plus any Excess Contributions received in which case the Member will be considered a new Employee for all purposes of the Plan effective as of the date of the reinstatement of employment.

Article 4. CONTRIBUTIONS

Section 4.01 Member Contributions

- (1) Every Member shall contribute to the Trust Fund in accordance with:
 - (a) For a Member who is a Design Two Employee, the applicable rates detailed in Appendix A.1 in respect of the applicable periods.
 - (b) For a Member who is an Other Than Regular Full-time
 Employee, the rates detailed in Appendix C of the Plan in respect of the applicable periods.

Provided, however, for the avoidance of doubt, the rate of Member contributions shall not exceed 9%.

- (1.1) Effective January 1, 2022, an Employer may permit Members who are Design Two Employees to elect to make contributions to the Trust Fund from a list of available rate options as detailed in Appendix A.1 in accordance with the following:
 - (a) The option to make such contributions shall be offered to eligible Members upon Plan enrolment and periodically in the future, as detailed in Appendix A.1.

- (b) Each contribution rate option provided to eligible Members shall attract a concurrent, associated Employer contribution, as detailed in Appendix A.1.
- (c) Once elected, the contribution associated with the rate chosen shall be considered to be a required contribution for purposes of the Act.
- (d) In the event an eligible Member fails to make an election, the Employer may specify a contribution rate for such Members as detailed in Appendix A.1
- For greater certainty, Member contributions made pursuant to Section 4.01(1) and 4.01(1.1) together shall not exceed 9%.
- (2) The contributions of a Member in a calendar year may not exceed the maximum amount that may be permitted for registered pension plans under the provisions of the Act, the Income Tax Act or any other applicable legislation as may be in force from time to time. Notwithstanding the generality of the foregoing, unless a higher limit is approved under the Income Tax Act, the contributions of a Member for a calendar year shall not exceed the limit specified under Section 8503(4)(a) of the Regulations of the Income Tax Act.

Section 4.02 <u>Contributions During Leave of Absence, Workers Compensation</u> Benefits, or LTD Benefits

(1) Periods of Leave of Absence

Each Member who is absent on an authorized leave of absence for any reason where the Member's Employer has authorized such leave and who is receiving Annual Earnings during that absence shall continue to contribute to the Trust Fund in respect of such absence in accordance with

Section 4.01 as applicable, based on the Member's actual Annual Earnings received during the absence.

(2) Periods of Workers Compensation Benefits

If a Member is receiving regular and periodic workers compensation benefits in respect of an Employer, provided that the benefit is not paid to the Member in a lump sum:

- (i) the Member will not be required to contribute under Section 4.01 during the first 12 consecutive months during which he or she is in receipt of such regular and periodic benefits (provided this is a requirement under the legislation under which such workers compensation benefits are provided); and
- (ii) the Member will accrue Pensionable Service and be credited with member contributions based on the Member's Deemed Annual Earnings from the date as of which the Member first becomes entitled to such benefits to the earliest of:
 - (aa) 12 months from the date as of which such regular and periodic benefits first commenced; and
 - (bb) the date such regular and periodic benefits cease;
 - (cc) their Termination of Membership Date.
- (iii) within the deadline established by the Board of Trustees, the Member may elect to contribute in accordance with Section 4.01 for the period in which they are receiving regular and periodic benefits under the Workplace Safety and Insurance Act that is between the end of the period

referenced in (i) and their Termination of Membership Date.

(iv) The Member's contributions for the period referenced in (iii) will be determined in accordance with Section 4.01 based on the Member's regular and periodic benefits under the Workplace Safety and Insurance Act, as determined using evidence provided by the Member that is acceptable to the Board of Trustees. Such contributions shall be payable to the Trust Fund by the Member in accordance with the deadlines and directions established by the Board of Trustees.

(3) Periods of Receipt of Benefits under Long Term Disability Plan

Within the deadline established by the Board of Trustees, a Member who is entitled to a regular and periodic benefit under a long term disability plan of their Employer may, provided that the benefit is not paid to the Member in a lump sum, elect to contribute to the Plan for the period in which they are entitled to a regular and periodic benefit under a long term disability plan of an Employer that is before their Termination of Membership Date. The Member's contributions for the period described in this Section will be determined in accordance with Section 4.01 based on the Member's regular and periodic benefits under the long-term disability plan of their Employer, as determined using evidence provided by the Member that is acceptable to the Board of Trustees. Such contributions shall be payable to the Trust Fund by the Member in accordance with the deadlines and directions established by the Board of Trustees.

(4) Limit

Contributions under this Section 4.02 shall be subject to the terms and restrictions in Section 4.04 and Section 5.03 and the Act, the Income Tax Act, and any other applicable legislation.

Section 4.03 Contributions for Periods of Purchasable Service

A Member may elect, prior to attaining their Termination of Employment Date, to contribute to the Trust Fund as provided in this Section 4.03 for any period of Purchasable Service that is after 1990 in respect of which they have not been credited with Pensionable Service. All contributions made pursuant to this Section 4.03 shall be subject to the terms and restrictions in Section 4.04 and the Act, the Income Tax Act and any other applicable legislation.

- (1) Periods of Service with an Employer Prior to Becoming a Member
- (a) Each Member may elect to contribute to the Trust Fund in respect of periods of employment with an Employer before the Member became a contributor under Section 4.01.
- (b) The Member may contribute up to 18% of the maximum pensionable compensation permissible under the Income Tax Act received from their Employer during the prior period of employment.
- (c) Such contribution shall be payable to the Trust Fund through a transfer directly from a Registered Retirement Savings Arrangement of which the Member is the annuitant and paid within three months of the date the Member is advised of the contribution required. If the contribution required is not paid to the Trust Fund through a transfer from a Registered Retirement Savings Arrangement of which the Member is the annuitant within three months of the date the Member is advised of the amount of the contribution required, a new request to contribute for the prior period of employment will be required and the contribution required will be re-determined as of the date the new request to contribute for the prior period of employment is received.

- (2) Periods of Parental, Pregnancy or Emergency Leave
 - (a) Every Member who is absent on a statutory leave under employment standards legislation applicable to the Member and which requires such Member to be allowed to continue participation in the Plan for the duration of such leave from an Employer, shall, unless the Member specifically elects to not contribute to the Trust Fund in respect of such absence, contribute to the Trust Fund in respect of such absence. The Member's contributions will be determined in accordance with Section 4.01 as applicable during the leave and based on the Member's Deemed Annual Earnings.
 - (b) If the Member elected not to contribute in respect of such leave and subsequently elects, within six months after the end of the leave to contribute in respect of such leave, the Member's contribution will be an amount equal to the amount of contributions determined in accordance with Section 4.03(2)(a) on Deemed Annual Earnings and shall be payable to the Trust Fund in a lump sum within six months after the end of the leave.
 - (c) A Member who did not make an election under (b) above or did make an election but did not make the required payment within six months after the end of the absence, may elect to contribute in respect of such leave. In that case, the Member's contribution will be the sum of:
 - (i) the amount of contributions required by the Member determined in accordance with Section 4.01, as applicable during the period of leave; plus

(ii) the amount of contributions that would have been payable by the Employer pursuant to subsections4.05(1) and (2) had the Member not commenced a period of leave as described in subsection (a) above, as applicable during the period of the leave.

Such contribution shall be based on the Member's Deemed Annual Earnings for the applicable period. Such contribution shall be payable to the Trust Fund through a transfer directly from a Registered Retirement Savings Arrangement of which the Member is the annuitant and paid within three months of the date the Member is advised of the contribution required. If the contribution required is not paid to the Trust Fund through a transfer from a Registered Retirement Savings Arrangement of which the Member is the annuitant within three months of the date the Member is advised of the amount of the contribution required, a new request to contribute for the leave will be required and the contribution required will be re-determined as of the date the new request to contribute for the leave is received.

- (3) Periods of Leave of Absence Without Pay or Layoff or Work Stoppage
 - (a) If a Member is
 - (i) on a leave of absence without pay for any reason where the Member's Employer has authorized such leave, as provided in Section 2.39(1) of the Design One Provisions, or
 - (ii) absent on a period of layoff as provided in Section 2.39(2)
 of the Design One Provisions during which recall rights exist, or

(iii) a member of a bargaining unit and is absent from work due to work stoppage and has not attained their Termination of Employment Date,

the Member may elect to contribute to the Trust Fund in accordance with this Section for the period of such absence that constitutes Continuous Service.

- (b) If the Member so elects after the end of the absence or cessation of the work stoppage, as applicable, the Member may contribute the sum of:
 - the amount of contributions required by the Member determined in accordance with Section 4.01, as applicable during the period of leave; plus
 - (ii) the amount of contributions that would have been payable by the Employer pursuant to subsections 4.05 (1) and (2) had the Member not commenced a period of leave as described in subsection (a) above, as applicable during the period of the leave.

Such contributions shall be based on the Member's Deemed Annual Earnings for the applicable period. Such contribution shall be payable to the Trust Fund through a transfer directly from a Registered Retirement Savings Arrangement of which the Member is the annuitant and paid within three months of the date the Member is advised of the contribution required. If the contribution required is not paid to the Trust Fund through a transfer from a Registered Retirement Savings Arrangement of which the Member is the annuitant within three months of the date the Member is advised of the amount of the contribution required, a new request to contribute for the period of absence or work stoppage, as applicable, will be required and the contribution required will be

re-determined as of the date the new request to contribute for the period of absence or work stoppage, as applicable, is received.

- (4) Prior Service With An Employer, Predecessor Employer or Non-Participating Employer
 - (a) A Member may elect to contribute to the Trust Fund in respect of prior periods of employment with (i) an Employer as provided in Section 2.39(5) of the Design One Provisions, (ii) a Predecessor Employer, or (iii) a Non-Participating Employer in accordance with this Section.
 - (b) If the Member so elects, the Member may contribute up to 18% of the maximum pensionable compensation permissible under the Income Tax Act received from an Employer as provided in Section 2.39(5) of the Design One Provisions, a Predecessor Employer, or a Non-Participating Employer during the prior period of employment.

Such contribution shall be payable to the Trust Fund through a transfer directly from a Registered Retirement Savings

Arrangement of which the Member is the annuitant and shall be paid within three months of the date the Member is advised of the contribution required. If the contribution required is not paid to the Trust Fund through a transfer from a Registered Retirement Savings Arrangement of which the Member is the annuitant within three months of the date the Member is advised of the amount of the contribution required, a new request to contribute for the prior period of employment will be re-determined as of the date the new request to contribute for the prior period of employment is received.

Section 4.04 **Payment of Member Contributions**

- (1) The contributions required to be made by a Member under Sections 4.01 and 4.02(1) in any pay period shall be deducted from the Member's Annual Earnings in the pay period. Any contributions payable by a Member under Section 4.02(2) or (3) shall be payable within the periods prescribed in that Section. Any contributions payable by a Member under Section 4.03 shall be payable within the periods prescribed in that Section.
- (2) Notwithstanding any provisions of the Plan, in no event will a Member contribute for periods of absence in excess of five years of full-time equivalent remuneration, plus three years of full-time equivalent remuneration for periods of parenting, throughout such Member's period of employment with any Employer after December 31, 1990, in accordance with the Act and the Income Tax Act. Contributions made during a calendar year for any prior period of service shall be subject to the Income Tax Act.

Section 4.05 **Employer Contributions**

In order to fund the benefits under the Plan, each Employer, subject to Section 4.07, shall contribute to the Trust Fund:

- (1) an amount equal to the contributions made by Members who are employed by that Employer that are required by Sections 4.01(1) and 4.02(1), or another contribution rate if set out in Appendix A.1 in respect of the applicable periods;
- (2) an amount based on contribution rates elected pursuant to Section 4.01(1.1) as detailed in Appendix A.1; and
- (3) an amount equal to the contributions made by Members who are employed by that Employer that are payable under Sections

4.02(2) and (3) and Sections 4.03(2)(a) and (b), or another contribution rate if set out in Appendix A.1 in respect of the applicable periods.

Section 4.06 **Remittance of Contributions**

The contributions made by a Member under Sections 4.01 and 4.02(1) together with the amount required to be paid by the Employer in respect of such contributions and the amount required to be paid by the Employer under Section 4.05(2) shall be paid by the Employer to the financial institution designated by the Board of Trustees so that they shall be received by that financial institution on a business day that is no more than thirty days following the month in respect of which the contributions were made.

When an Employer fails to pay to the Board of Trustees the contributions within the time specified above, there shall be charged to the Employer and deposited to the Trust Fund one and one-half per cent of the amount of the unpaid contributions on the first day of each subsequent month, prorated for partial months, until the total amount due has been paid.

Notwithstanding other provisions of this Section 4.06, all Member contributions under Sections 4.01 and 4.02(1) shall be remitted by the Employers to the Board of Trustees in accordance with the Act and the Income Tax Act.

Section 4.07 **Overpayments**

Any overpayment of a contribution by an Employer or a Member in accordance with this Article 4 may be refunded to the Employer or the Member, as the case may be, provided it is in accordance with applicable legislation.

Section 4.08 **Deficiencies or Surplus While Plan is Ongoing**

Employer and Member current service contributions made in accordance with Section 4.01 and Section 4.05 shall be inclusive of contributions required to be made in respect of any going concern unfunded liability identified by the Actuary in a valuation filed pursuant to the Act. However, and notwithstanding any other terms of the Plan, in the event a going concern unfunded liability is identified by the Actuary in a valuation filed pursuant to the Act and the contributions made in accordance with Section 4.01 and Section 4.05 are not sufficient to cover contributions required to be made in respect of such going concern unfunded liability, then the Plan will be amended to reduce the Base Benefit in respect of future contributions to ensure that the contributions under Section 4.01 and Section 4.05 are sufficient to cover the contributions required to be made in respect of such going concern unfunded liability. Contributions shall not be required in respect of any solvency deficiency identified by the Actuary in a valuation filed pursuant to the Act. Any such contributions to the Plan shall not exceed the maximum contributions permitted, under the Act and the Income Tax Act. While the Plan remains ongoing any funding excess shall be applied in the manner determined by the Sponsors' Committee, subject to the provisions of Article 11.

Article 5. PENSIONABLE SERVICE

Section 5.01 Pensionable Service

The Pensionable Service of a Member shall mean the total number of years, each of which is a year in which contributions were made by or on behalf of the Member under Section 4.01,4.02, 4.03(2)(a), or 4.03(2)(b). Pensionable Service shall also include periods of Pensionable Service accrued by a Member in accordance with Section 4.02(2)(ii).

For greater certainty, periods of Purchasable Service in respect of which a Member elects to contribute to the Trust Fund in accordance with Section 4.03(1), 4.03(2)(c), 4.03(3), or 4.03(4) shall not be included in Pensionable Service.

Section 5.02 Maximum Pensionable Service

Notwithstanding Article 5.01 or any other provision of the Plan, the following limits to Pensionable Service apply:

- (a) In no event will a Member accrue more than the equivalent of 1 year of Pensionable Service under the Design One Provisions and Design Two Provisions combined in any one calendar year;
- (b) No Pensionable Service under the Design Two Provisions shall be credited in a year where Pensionable Service is credited under the Design One Provisions; and
- (c) In no event will an Employee with concurrent employment with more than one Employer in a calendar year accrue in aggregate more than one year of Pensionable Service in respect of that calendar year.

Section 5.03 Limitations on Pensionable Service

(1) In no event will a Member receive Pensionable Service during periods of absence, other than a period in which the Member is Disabled, in excess of five years of full-time equivalent remuneration, plus three years of full-time remuneration for periods of parenting, throughout the Member's periods of employment with any Employer after December 31, 1990 in accordance with the Act and the Income Tax Act.

(2) A person may not accrue Pensionable Service and concurrently receive a pension under the Plan payable from the Trust Fund in respect of their own Pensionable Service.

Section 6.01 Normal Retirement Pension

Each Member who has attained their Termination of Employment Date and who has attained their Termination of Membership Date coincident with their Normal Retirement Date will be entitled to receive from the Trust Fund an annual pension benefit, as of their Normal Retirement Date, determined in accordance with the following formula, and payable in accordance with Article 9:

Normal Retirement Pension = Base Benefit + AIW Benefit

Section 6.02 Maximum Pension Benefit

Notwithstanding any other provisions of the Plan, the amount of lifetime pension benefits provided under the Design Two Provisions, in respect of any Member at Termination of Membership Date or termination of the Plan, or otherwise, may not exceed the maximum lifetime retirement benefits permitted under the Income Tax Act.

Section 6.03 **Reduction of Benefits**

Notwithstanding any other provisions of the Plan, benefits under the Plan may be reduced or contributions made in accordance with the Plan may be refunded, as appropriate, if such benefits or contributions exceed the maximum limits as prescribed under the Income Tax Act in order to avoid revocation of the Plan, subject to the Act and with the consent of the Financial Services Regulatory Authority of Ontario.

Article 7. EARLY RETIREMENT

Section 7.01 Early Retirement Date

A Member who has attained their Termination of Employment Date prior to Normal Retirement Date and who coincident with or prior to their Termination of Membership Date has attained age 50 may elect an Early Retirement Pension as of the Member's Early Retirement Date, being the last day of any month coincident with or following the attainment of age 50, subject to the provisions of this Article 7. For the purposes of this Article 7, age is calculated in years and part years on the last day of the month in which a determination is required.

Section 7.02 Early Retirement Pension

- (1) Determination of Early Retirement Pension

 Each Member who has elected an Early Retirement Date, subject to the provisions of this Article 7, will be entitled to receive from the Trust Fund an annual pension benefit and payable in accordance with the provisions of Article 9, computed as the pension determined in accordance with Section 6.01 reduced by five percent for each full year and prorated for partial years that the Member's Early Retirement Date precedes the Member's Normal Retirement Date.
- (2) Early Retirement Dates before January 1, 2026

 Notwithstanding Section 7.02(1), each Member who elects an

 Early Retirement Date and Early Retirement Pension to commence
 before January 1, 2026, subject to the provisions of this Article 7,

 will be entitled to receive from the Trust Fund an annual pension
 benefit and payable in accordance with the provisions of Article 9,
 computed as the pension determined in accordance with Section

6.01 reduced by three percent for each full year and prorated for

partial years that the Member's Early Retirement Date precedes the Member's Normal Retirement Date.

(3) Pension Payments

In the event that:

- (a) the application to commence an Early Retirement Pension is not received by the Plan Manager within one month following the later of the Member's Termination of Employment Date and the date on which the Member attains age 50 co-incident with or following a Member's Termination of Employment Date, then the Member's Early Retirement Date shall be the end of the month in which the application is received.
- (b) if the application is received more than 12 months after the Member's Termination of Membership Date, the person is entitled to a Deferred Pension determined pursuant to Article 10 and the provisions of Section 10.03 applicable to early commencement of Deferred Pension will apply.

Section 7.03 Maximum Pension Benefit

The Member's pension benefit payable on early retirement will be subject to the provisions of Section 6.02 and any applicable maximum permitted benefit provisions of the Income Tax Act.

Article 8. POSTPONED RETIREMENT

Section 8.01 Postponed Retirement Date

If a Member has not attained their Termination of Employment Date coincident with or prior to their Normal Retirement Date, pension benefits under the Plan will not commence until the Member's Postponed Retirement Date which will be the date when the earlier of (a) and (b) below occurs:

- (a) November 30th of the year in which the Member attains age 71, or such other age as is specified in the Income Tax Act (Canada), and
- (b) the last day of the month in which the Member's Termination of Employment Date occurs,

and the Member will continue to make contributions under Section 4.01 until the month in which the Member's Postponed Retirement Date occurs.

Section 8.02 **Postponed Retirement Pension**

Each Member who attains a Postponed Retirement Date under Section 8.01 will be entitled to receive from the Trust Fund an annual pension benefit, computed pursuant to Section 6.01 and payable in accordance with the provisions of Article 9.

The Member's Postponed Retirement Pension under this Section 8.02 will be subject to the provisions of Section 6.02.

Article 9. PAYMENT OF PENSION

Section 9.01 Payment of Pension

The payment of pensions under the Design Two Provisions shall be in accordance with Article 9 of the Design One Provisions, provided, however, that, for the payment of pensions under the Design Two Provisions, Article 9 of the Design One Provisions shall be modified as follows:

- (a) references to Article 6 shall be read as referring to Article 6 of the Design Two Provisions;
- (b) references to Article 7 shall be read as referring to Article 7 of the Design Two Provisions;
- (c) references to Article 8 shall be read as referring to Article 8 of the Design Two Provisions;
- (d) references to Article 10 shall be read as referring to Article 10 of the Design Two Provisions;
- (e) the reference to Section 10.03 in Section 9.06(a) shall be read as referring to Section 10.03 of the Design Two Provisions;
- (f) references to Sections of Article 9 shall be read as referring to Sections of Article 9 of the Design One Provisions subject to any modifications thereof in accordance with this Section;
- (g) the last paragraph of Section 9.06(1)(b) shall be deleted;
- (h) the reference to Section 3.03(2) in Section 9.06(2) shall be replaced with a reference to Section 3.05(2) of the Design Two Provisions; and

(i) references to terms in Article 9 that are defined in this Appendix G and which are used in the capitalized form in Article 9 shall be read as referring to those terms as defined in this Appendix G.

Article 10. BENEFITS ON TERMINATION OF EMPLOYMENT

Section 10.01 **Termination with Deferred Pension**

If a Member attains their Termination of Membership Date and is not then entitled to make an election under Section 7.01 to commence an Early Retirement Pension and has not attained their Normal Retirement Date, the Member will be entitled to receive a Deferred Pension in the amount accrued to the Member under Section 6.01, and shall be payable at the Member's Normal Retirement Date in accordance with Article 9 or such other date as the Member may elect under Section 10.03.

Section 10.02 Portability Options

Portability options under the Design Two Provisions shall be in accordance with Section 10.02 of the Design One Provisions, provided, however, that, for portability options under the Design Two Provisions, Section 10.02 of the Design One Provisions shall be modified as follows:

- (a) the reference to Section 7.01 in Section 10.02 shall be read as referring to Section 7.01 of the Design Two Provisions;
- (b) references to Section 10.02 in Section 10.02 shall be read as referring to Section 10.02 as modified in accordance with this Section; and
- (c) references to terms in Section 10.02 that are defined in this Appendix G and which are used in the capitalized form in Section

10.02 shall be read as referring to those terms as defined in this Appendix G.

Section 10.03 Early Commencement of Deferred Pension

On or after the attainment of age 50, a Former Member may elect to have payment of the Deferred Pension commence prior to the Former Member's Normal Retirement Date on the first day of any month subsequent to the Former Member's election being received by the Plan Manager.

The Former Member's Deferred Pension will be the pension determined in accordance with Section 6.01 reduced by five per cent for each full year and prorated for partial years that the Former Member's age at the date pension benefits commence is less than 65 years and will be payable in the normal form under Section 9.01 or 9.02, whichever is applicable, or the optional form as may be elected under Section 9.03 and for the periods provided in Section 9.06.

Article 11. INDEXATION

Section 11.01 Inflation Protection Factor

In this Article, the term "inflation protection factor" shall have the same meaning as specified in Section 14.01(1) of the Design One Provisions and the rules specified in Section 14.01(2) of the Design One Provisions for the application of an inflation protection factor in a subsequent 12 month period shall also apply.

Section 11.02 Increases in respect of service after December 2019

Pensions payable to Retired Members and to the surviving Spouse of a deceased Member, Former Member or Retired Member and the annual amount of Deferred Pension payable to a Former Member and the deferred pension payable to a surviving Spouse under Section 12.01 of the Design One Provisions, shall be increased on an annual basis pursuant to Section 11.03 in respect of service attributable to each calendar year after December 31, 2019 but only if the most recent actuarial valuation of the Plan for funding purposes discloses prior unutilized funding excesses or cumulative gains, and in the following order of priority:

- (1) increases in respect of the immediately preceding calendar year, followed by,
- (2) increases in respect of previous years where the maximum increase permitted by Section 11.03 had not previously been effected in respect of such years, in chronological order commencing with the first calendar year following 2019 in respect of which the maximum increase permitted by Section 11.03 has not previously been effected.

Section 11.03 **Application of Increases**

Any increases pursuant to Section 11.02 will be made each January 1 in accordance with Section 14.04 of the Design One Provisions provided, however, that, for the purposes of increases pursuant to Section 11.02, references in Section 14.04 of the Design One Provisions to terms that are defined in this Appendix G and which are used in the capitalized form in Section 14.04 of the Design One Provisions shall be read as referring to those terms as defined in this Appendix G.

Section 11.04 Cap on Benefit Increases

- (1) The aggregate benefit increases effected pursuant to Section 11.02 in respect of the period covered by an actuarial valuation of the Plan for funding purposes shall not exceed the difference between: (a) the prior unutilized funding excesses plus cumulative gains that are disclosed in such actuarial valuation of the Plan and, (b) the amounts required to liquidate the going concern unfunded liability of the Plan, if any, disclosed in such valuation.
- (2) For greater certainty, in no circumstances shall the aggregate benefit increases effected pursuant to Section 11.02 in respect of a particular year exceed the inflation protection factor in respect of those benefits for that year.

For greater certainty, the cumulative increase provided under Section 11.02 to any person shall not at any time exceed the cumulative increase in the Consumer Protection Index from the year the first increase was so provided to the year of the most recent increase so provided under Section 11.02

Article 12. OTRFT EMPLOYEE TRANSITION AND INTER-DESIGN ISSUES

Section 12.01 **Definitions**

In this Article 12, unless the context clearly indicates otherwise, the following terms shall have the following meanings:

- (a) "Continuing Design One Member" shall mean:
 - i) a Member who ceased to be a Full-time Employee and became an OTRFT Employee or a Design Two Employee before attaining their Termination of Membership Date; or
 - ii) a Member who was concurrently employed as a Full-time Employee and as an OTRFT Employee or a Design Two Employee and who subsequently ceased to be a Full-time Employee but continued to be employed as an OTRFT Employee or a Design Two Employee.
- (b) "Continuing Design Two Member" shall mean a Member who either is or was a OTRFT Employee, including a Transferred OTRFT Employee, or a Design Two Employee and who becomes a Full-time Employee before attaining their Termination of Membership Date.
- (c) "Effective Date" shall mean January 1, 2019.
- (d) "<u>Transferred OTRFT Employee</u>" shall mean an OTRFT Employee who was a Member on the day immediately before the Effective Date and who is transferred

to the Design Two Provisions in accordance with Section 12.02(a) as of the Effective Date.

(e) "Transition Date" shall mean:

- for a Continuing Design One Member in accordance with Section 12.01(a)i), the date they became an OTRFT Employee or a Design Two Employee after ceasing to be a Full-time Employee; or
- ii) for a Continuing Design One Member in accordance with Section 12.01(a)ii), the date they ceased to be a Full-time Employee; or
- iii) for a Continuing Design Two Member, the date they became a Full-time Employee.

Section 12.02 OTRFT Employee Transition

- (a) Each OTRFT Employee who was a Member on the day immediately before the Effective Date shall be transferred to the Design Two Provisions as of the Effective Date.
- (b) The following rules apply to each Transferred OTRFT Employee:
 - i) Notwithstanding any other provision of the Plan, on and after the Effective Date, each Transferred OTRFT Employee shall be entitled to benefits accrued under the Plan before the Effective Date for their Pensionable Service under the Design One Provisions before the Effective Date using:
 - A. their Pensionable Service under the Design One Provisions as of the day immediately before the Effective Date;
 - B. their Highest Average Pensionable Earnings under the Design One Provisions as calculated in accordance with ii); and
 - C. the Average YMPE under the Design One Provisions as calculated in accordance with iii).

- ii) Notwithstanding any other provision of the Plan, the Highest Average Pensionable Earnings under the Design One Provisions of a Transferred OTRFT Employee shall be determined using their Pensionable Earnings under the Design One Provisions before the Effective Date and Pensionable Service under the Design One Provisions as of the day immediately before the Effective Date.
- iii) Notwithstanding any other provision of the Plan, in relation to a Transferred OTRFT Employee, Average YMPE shall mean the average of the YMPE for the earlier of 2018 and the year in which the Member's Termination of Employment Date occurred and for each of the four years preceding the earlier of 2018 and the year in which the Member' Termination of Employment occurred.
- iv) The pension entitlement of each Transferred OTRFT Employee for service as an OTRFT Employee or Design Two Employee on and after the Effective Date shall be in accordance with the Design Two Provisions.
- v) On and after the Effective Date, a Transferred OTRFT Employee's Pensionable Service under the Design Two Provisions on and after the Effective Date shall be recognized under the Design One Provisions for the purpose of determining eligibility or entitlement to early retirement benefits under the Design One Provisions.
- vi) On and after the Effective Date and while they are an OTRFT Employee or a Design Two Employee, a Transferred OTRFT Employee shall contribute to the Plan in accordance with the Design Two Provisions.
- vii) The pension benefit entitlement of each Transferred OTRFT Employee shall be:
 - (a) The amount specified in i); plus

(b) The pension benefit accrued by the Transferred OTRFT Employee under the Design Two Provisions on and after the Effective Date or as otherwise credited to them under the Plan.

Section 12.03 Design One to Design Two – Continuing Design One Member

The following rules apply to a Continuing Design One Member:

- (a) On and after their Transition Date and while they are an OTRFT Employee or a Design Two Employee, a Continuing Design One Member will accrue benefits in accordance with the Design Two Provisions.
- (b) Any Pensionable Service of a Continuing Design One Member on and after their Transition Date under the Design Two Provisions will be recognized under the Design One Provisions for the purpose of determining the Continuing Design One Member's eligibility or entitlement to benefits under the Design One Provisions.
- (c) On and after their Transition Date and while they are an OTRFT Employee or a Design Two Employee, a Continuing Design One Member shall contribute to the Plan in accordance with the Design Two Provisions.
- (d) A Continuing Design One Member shall be entitled to benefits accrued under the Design One Provisions before their Transition Date for Pensionable Service under the Design One Provisions before their Transition Date using:
 - A. their Pensionable Service under the Design One Provisions as of the day immediately before their Transition Date;
 - B. their Highest Average Pensionable Earnings under the Design One Provisions as calculated in accordance with (e); and
 - C. the Average YMPE under the Design One Provisions as calculated in accordance with (f).

- (e) Notwithstanding any other provision of the Plan, the Highest Average Pensionable Earnings under the Design One Provisions of a Continuing Design One Member shall be determined using their Pensionable Earnings under the Design One Provisions before their Transition Date and Pensionable Service under the Design One Provisions as of the day immediately before their Transition Date.
- (f) Notwithstanding any other provision of the Plan, in relation to a Continuing Design One Member, Average YMPE shall mean the average of the YMPE for the year in which the Transition Date occurred and for each of the four preceding years.
- (g) The pension benefit entitlement of each Continuing Design One Member shall be:
 - 1) The amount specified at (d); plus
 - 2) The pension benefit accrued by the Continuing Design One Member under the Design Two Provisions on and after their Transition Date or as otherwise credited to them under the Plan.

Section 12.04 Design Two to Design One - Continuing Design Two Members

The following rules apply to a Continuing Design Two Member:

- (a) On and after their Transition Date and while they are a Full-time Employee, a Continuing Design Two Member will accrue benefits in accordance with the Design One Provisions.
- (b) Any Pensionable Service of a Continuing Design Two Member under the Design Two Provisions before their Transition Date will be recognized under the Design One Provisions for the purpose of determining the Continuing Design Two Member's eligibility or entitlement to benefits under the Design One Provisions.
- (c) On and after their Transition Date and while they are a Full-time Employee, a Continuing Design Two Member shall contribute to the Plan in accordance with the Design One Provisions.

- (d) A Continuing Design Two Member shall continue to be entitled to benefits accrued under the Design Two Provisions before their Transition Date for Pensionable Service under the Design Two Provisions before their Transition Date.
- (e) The pension benefit entitlement of each Continuing Design Two Member shall be:
 - 1) The amount specified at (d); plus

The pension benefit accrued by the Continuing Design Two Member under the Design One Provisions on and after their Transition Date or as otherwise credited to them under the Plan.

Section 12.05 Coordination of Benefits – Income Tax Act

Notwithstanding anything to the contrary herein or in the Plan, for a Member with benefits under the Design One Provisions and the Design Two Provisions, the benefit payable under the Design One Provisions and the Design Two Provisions combined shall be subject to any applicable maximum permitted benefit provisions of the Income Tax Act.

Section 12.06 Coordination of Benefits – Continuation of Membership

Notwithstanding anything to the contrary herein or in the Plan:

- (a) In the case of a Continuing Design One Member or Continuing Design Two Member, Termination of Membership Date means the date they have attained their Termination of Membership Date under both the Design One Provisions and the Design Two Provisions.
- (b) In the case of a Continuing Design One Member or Continuing Design Two Member, Normal Retirement Date means the date they have attained their Normal

- Retirement Date under both the Design One Provisions and the Design Two Provisions.
- (c) In the case of a Continuing Design One Member or Continuing Design Two Member, Early Retirement Date means the date they have attained the Early Retirement Date under either the Design One Provisions and the Design Two Provisions, provided that if the Member has not attained the conditions set out in (1), (2) or (3) of Section 7.02 under the Design One Provisions, any pension payable in respect of such Member under the Design One Provisions shall be reduced by 5% per year from his or her Normal Retirement Date and no amount on account of 7.02(1)(6) under the Design One Provisions will be paid until the Member has attained age 55.
- (d) In the case of a Continuing Design One Member or Continuing Design Two Member, Postponed Retirement Date means the date they have attained the Postponed Retirement Date under both the Design One Provisions and the Design Two Provisions.
- (e) A Continuing Design One Member is not entitled to commence receiving pension payments under the Design Two Provisions unless they have elected, or are required, to commence receiving pension payments under the Design One Provisions.
- (f) A Continuing Design Two Member is not entitled to commence receiving pension payments under the Design One Provisions unless they have elected, or are required, to commence receiving pension payments under the Design Two Provisions.

Section 12.07 Coordination of Benefits – Excess Contributions

Notwithstanding anything to the contrary herein or in the Plan, for a Member with benefits under the Design One Provisions and the Design Two Provisions, Excess Contributions means the excess of (1) over (2) as follows:

- (1) the Member's contributions made in accordance with Article 4 of the Design One Provisions and Article 4 of the Design Two Provisions plus Credited Interest to the date a determination is required;
- (2) 50% of the Commuted Value of the Member's pension benefit in respect of Pensionable Service under the Design One Provisions and the Design Two Provisions, calculated as of the same date as in subsection (1);

provided that the Member contributions in (1) above and the Member's pension benefit in (2) above will not apply to benefits resulting from an election by a Member and the Member's contributions made to purchase Pensionable Service in respect of which the Employer makes no contributions pursuant to Section 4.05 of the Design One Provisions in respect of the Member or to benefits resulting from the election by a Member and to the Member's contributions in accordance with Section 4.03 for periods of Purchasable Service in respect of which the Employer makes no contributions pursuant to Section 4.05 in respect of the Member.

Section 12.08 Coordination of Benefits - Payment of Pension

Notwithstanding anything to the contrary herein or in the Plan, for a Member with benefits under the Design One Provisions and the Design Two Provisions, the benefit payable under the Design One Provisions and Design Two Provisions shall be combined and considered one benefit and shall be payable in accordance with Article 9.

Section 12.09 Coordination of Benefits - Death Prior to Pension Commencement

Notwithstanding anything to the contrary herein or in the Plan, for a Member, Retired Member, or Former Member with benefits under the Design One Provisions and the Design Two Provisions who dies before the first installment of their pension is due, the benefit payable under Article 12 of the Design One Provisions shall be in respect of the

benefits under the Design One Provisions and Design Two Provisions combined and shall be considered one benefit.

Schedule 1 to Appendix G

1. Definitions

In this Schedule 1, unless the context clearly indicates otherwise, the following terms shall have the following meanings:

- (a) "AIW Factor" shall mean the Increase in Average Industrial Wage as may be determined by the Sponsors' Committee as of a Date of Determination.
- (b) "AIW Adjusted Design One Benefit" shall mean:
 - a. For a Transferred OTRFT Employee, the value of the benefits they accrued under the Design One Provisions before the Effective Date, adjusted by all applicable AIW Factors while they are accruing benefits under the Design Two Provisions until the earlier of: the expiry after the Effective Date of a twenty-four (24) consecutive month period, which period may commence before or after the Effective Date, in which no contributions have been made by or on behalf of the Member and their first Termination of Membership Date after the Effective Date.
 - b. For a Continuing Design One Member, the value of the benefits they accrued under the Design One Provisions between their last Termination of Membership Date under Design One Provisions, if any, and their Transition Date, adjusted by all applicable AIW Factors while they are accruing benefits under the Design Two Provisions until the earlier of: the expiry after their Transition Date of a twenty-four (24) consecutive month period, which period may commence before or after their Transition Date, in which no contributions have been made by or on behalf of the Member and their first Termination of Membership Date after their Transition Date.
- (c) "AIW Adjusted Design Two Benefit" shall mean:
 - a. The value of a Member's Base Benefit as of a specific date as determined by the Sponsors' Committee, adjusted by all applicable AIW Factors.

- b. For a Continuing Design Two Employee, the value of the benefits they accrued under the Design Two Provisions between their last Termination of Membership Date under Design Two Provisions, if any, and their Transition Date, adjusted by all applicable AIW Factors while they are accruing benefits under the Design One Provisions until the earlier of: the expiry after their Transition Date of a twenty-four (24) consecutive month period, which period may commence before or after their Transition Date, in which no contributions have been made by or on behalf of the Member and their first Termination of Membership Date after their Transition Date.
- (d) "Average Industrial Wage" shall mean the average Canadian weekly wages and salaries, defined as the industrial aggregate, as reported by Statistics Canada or, in the event the industrial aggregate ceases to be published, such measurement that replaces the industrial aggregate as determined by the Sponsors' Committee.
- (e) "Increase in Average Industrial Wage" shall mean the percentage by which the Average Industrial Wage for the 12-month period ending June 30 immediately prior to the Date of Determination increases or decreases as compared to the Average Industrial Wage for the 12-month period ending on July 1 in the prior calendar year, calculated to the second decimal place. In the event that the Average Industrial Wage on a June 30 does not exceed the Average Industrial Wage on the immediately preceding July 1, the Increase in the Average Industrial Wage shall be 0%.
- (f) "Date of Determination" shall mean one or more dates determined by the Sponsors' Committee, as set out in Exhibit 1 to this Schedule 1.

In addition to the terms defined in this Schedule 1, any terms which are defined in Article 2 of the Design One Provisions and which are used in the capitalized form in this Schedule 1 shall have the same meaning as set out in Article 2 of the Design One Provisions, as modified, if applicable, in accordance with Appendix G and any terms which are defined in Appendix G and which are used in the capitalized form in this Schedule 1 shall have the same meaning as set out in Appendix G.

2. Establishment of Date of Determination

The Sponsors' Committee may, in its sole discretion and subject to applicable limitations under the Income Tax Act, establish a Date of Determination for the application of an AIW Factor through an amendment to Exhibit 1 to this Schedule 1.

3. Application of AIW Factor

Where the Sponsors Committee has established a Date of Determination for the application of an AIW Factor through an amendment to Exhibit 1 to this Schedule 1, each Member accruing benefits under the Design Two Provisions on the Date of Determination shall be credited with a benefit determined using the formula, A+B, where:

A is the product of the value of their AIW Adjusted Design One Benefit as at December 31 of the year immediately preceding the Date of Determination by the corresponding AIW Factor; and

B is the product of their AIW Adjusted Design Two Benefit as at December 31 of the year immediately preceding the Date of Determination by the corresponding AIW Factor.

4. Calculation of Member's AIW Benefit

An individual's AIW Benefit determined as of the Member's Termination of Membership Date shall be calculated in accordance with the following formula:

(AIW Adjusted Design One Benefit + AIW Adjusted Design Two Benefit) - Base Benefit

Exhibit 1 to Schedule 1 to Appendix G

Date of Determination		
January 1, 2020		
January 1, 2021		
January 1, 2022		
January 1, 2023		
January 1, 2024		
January 1, 2025		
January 1, 2026		

Formatted: Font color: Auto

APPENDIX GP—DESIGN THREE PROVISIONS

Article 1. APPLICATION

Section 1.01 Application of Appendix GP

This Appendix GP is added to the Plan effective August 1, 2024.

This Appendix GP applies to Members, Former Members and Retired Members who elect to participate in the GP Account under this Appendix GP by means of the AVC Account or Variable Benefit Account, as defined herein.

Section 1.02 Application of Design One and Design Two Provisions

In addition to the provisions of this Appendix GP, Articles 1, 15, 16, 17 and 18 of the Design One Provisions shall also apply to a GP Participant who participates under this Appendix GP.

In addition to the terms defined in this Appendix GP, any terms which are defined in Article 2 of the Design One Provisions or Article 2 of the Design Two Provisions, as the case may be, and which are used in capitalized form in this Appendix GP shall have the same meaning as set out in Article 2 of the Design One Provisions or Article 2 of the Design Two Provisions, as the case may be.

Article 2. DEFINITIONS

Section 2.01 Definitions

<u>In this Appendix GP</u>, unless the context clearly indicates otherwise, the following terms shall have the following meanings.

- (a) "Additional Voluntary Contributions" shall mean any voluntary contributions made to the Trust Fund by a GP Participant, in addition to any contributions under either the Design One Provisions or Design Two Provisions and Lump Sum Voluntary Transfers.
- (b) "AVC Account" shall mean the account established under the Trust

 Fund for a GP Participant to which Additional Voluntary

 Contributions and Lump Sum Voluntary Transfers are deposited, and from which transfers and withdrawals are made, up to the end of the year a member turns 71.
- (c) "GP Participant" shall mean a Member, Former Member or Retired

 Member who participates under this Appendix GP by means of their

 AVC Account or Variable Benefit Account.
- (d) "GP Account" shall mean the account established under the Trust
 Fund for a GP Participant comprised of an AVC Account and a
 Variable Benefit Account, as applicable.
- (e) "GP Spouse" shall mean, in respect of a GP Participant, the individual who is the "spouse", "common law partner" or equivalent, as defined in the Act, of the GP Participant at the time of the death of the GP Participant.
- (f) "Lump Sum Voluntary Transfer" means:
 - any amount transferred to the Trust Fund in accordance with
 Article 5 for deposit into the GP Participant's AVC Account, and held as an Additional Voluntary Contribution, including from a registered pension plan, Registered Retirement Savings
 Arrangement or any other plan or arrangement permitted under the Act and the Income Tax Act; or
 - any amount transferred to the Trust Fund in accordance with <u>Article 9 for deposit into the GP Participant's Variable Benefit</u> <u>Account, as permitted in accordance with paragraph 8506(2)(c.1)</u> <u>of the Income Tax Regulations.</u>
- (g) "Specified Beneficiary" means the "specified beneficiary", as defined in and subject to section 8506(8) of the *Income Tax Regulations*, of a GP Participant.

(h) "Variable Benefit Account" means the account established under the Trust Fund for a GP Participant on or after the beginning of the year they turn 72, to which funds are transferred from their AVC Account, or Lump Sum Voluntary Transfers are deposited, and from which variable benefit payments are made.

<u>Unless otherwise indicated, a reference in this Appendix GP to a</u> provision, Section or Article is a reference to a provision, Section or <u>Article of this Appendix GP.</u>

Article 3. ELIGIBILITY AND PARTICIPATION

Section 3.01 Eligibility

A Member participating under the Design One Provisions or the Design Two Provisions shall be eligible to participate under this Appendix GP by means of their AVC Account. A Former Member and Retired Member shall also be eligible to participate, subject to Article 5 herein. A Retired Member shall also be eligible to participate by means of their Variable Benefit Account, subject to Article 9 herein.

Section 3.02 Non-Residents

Notwithstanding section 3.01, a Member, Former Member or Retired Member who is a non-resident of Canada for purposes of the Income Tax Act shall not be eligible to commence participation, or make Additional Voluntary Contributions or Lump Sum Voluntary Transfers, under this Appendix GP.

Article 4. AVC ACCOUNT

Section 4.01 Establishment of AVC Account

For each GP Participant, an AVC Account shall be established under the Trust Fund for purposes of accepting Additional Voluntary Contributions and Lump Sum Voluntary Transfers. The balance of such AVC Account shall be credited with a rate of return as set out in Article 6 of this Appendix GP.

Article 5. DEPOSITS TO AVC ACCOUNT

Section 5.01 Deposits to AVC Account

A GP Participant may make deposits to an AVC Account in accordance with this Article 5, in such manner, and on such timing and conditions acceptable to the Plan Manager, and subject to any limitations under the Act and the Income Tax Act. For greater certainty, an Employer is not permitted to make contributions to the Trust Fund in connection with a GP Participant's GP Account.

Section 5.02 Additional Voluntary Contributions

A GP Participant, who is a Member participating in the Design One

Provisions or the Design Two Provisions, may make Additional Voluntary

Contributions which shall be credited to the GP Participant's AVC

Account.

It shall be the responsibility of the GP Participant to ensure that such

Additional Voluntary Contributions are permitted under the Income Tax

Act.

Section 5.03 Lump Sum Voluntary Transfers

A GP Participant may make Lump Sum Voluntary Transfers to the Trust Fund which shall be credited to the GP Participant's AVC Account from another registered pension plan, Registered Retirement Savings

Arrangement or any other transfer permitted by the Act and the Income Tax Act. In the event that any such transfer is locked-in under the Act, such funds shall continue to be treated as locked-in for purposes of the Act under this Appendix GP. Such Lump Sum Voluntary Transfers shall be held as an Additional Voluntary Contribution.

Article 6. RATE OF RETURN ON GP ACCOUNT BALANCE

Section 6.01 Return on Balance

The balance held to the credit of the GP Participant in their GP Account shall be credited at least once annually, with such rate of return as can reasonably be attributed to the operation of the Trust Fund net of any investment and administration fees and expenses.

The return shall be calculated and determined in a manner acceptable to the Plan Manager.

Article 7. OPTIONAL WITHDRAWALS AND TRANSFERS FROM AVC ACCOUNT

Section 7.01 Withdrawals and Transfers

A GP Participant may make optional withdrawals and transfers from an AVC Account in accordance with this Article 7, in such manner, and on such timing and conditions acceptable to the Plan Manager, and subject to any limitations under the Act and the Income Tax Act. For greater certainty, such withdrawals and transfers shall be credited with a rate of

return pursuant to Article 6 of this Appendix GP.

Section 7.02 Withdrawal from AVC Account

In the event that any portion of a GP Participant's AVC Account balance is composed of funds that are not locked-in under the Act, a GP Participant shall have the option to withdraw such portion from their AVC Account.

Section 7.03 Transfers from AVC Account

Funds held in a GP Participant's AVC Account may be transferred by a GP Participant to a registered pension plan, or a Registered Retirement Savings Arrangement. The balance may also, if such option is required to be offered under the Act, be transferred by the GP Participant to the provider of an annuity for the purchase of a life annuity.

Section 7.04 Transfers Within the Plan

Funds held in a GP Participant's AVC Account may be transferred in connection with the purchase of periods of Purchasable Service under the Design One Provisions or Design Two Provisions of the Plan in accordance with the following sections:

I. Schedule 1, Sections 4.03 (1), (2)(c), (3), (4) and (5); and, II. Appendix G, sections 4.03 (1), (2)(c), (3) and (4),

provided that the GP Participant has not reached their Termination of Employment Date.

Article 8. MANDATORY WITHDRAWALS FROM AVC ACCOUNT

Section 8.01 Mandatory Withdrawals from AVC Account

Notwithstanding Article 7, a GP Participant is required to withdrawal or transfer funds from their AVC Account in accordance with this Article 8, in such manner and subject to the conditions described in subsections 7.02 and 7.03, as applicable.

Section 8.02 No Plan Entitlement

In the event that a GP Participant ceases to be entitled to a benefit under the Design One Provisions, the Design Two Provisions or any other defined benefit entitlement under the Plan for any reason, the GP Participant shall be immediately required to withdraw or transfer the full balance of their AVC Account.

Section 8.03 Certain Locked-in Funds

In the event that any portion of a GP Participant's AVC Account balance is composed of funds locked-in under the Act and the Act prohibits the payment of variable benefits under the Plan, the GP Participant shall be required to transfer such portion to a locked-in arrangement permitted under the Act, no later than the end of the year the GP Participant attains 71 years of age or such other age as required by the Income Tax Act.

Article 9. VARIABLE BENEFIT ACCOUNT

Section 9.01 Establishment of and Transfer to the Variable Benefit Account

Where permitted by the Act, a Variable Benefit account shall be established under the Trust Fund for each GP Participant for the purposes of transferring any remaining balance in an AVC Account after the end of

the year in which the GP Participant attains 71 years of age, or such other age as required by the Income Tax Act, and for purposes of accepting Lump Sum Voluntary Transfers.

A GP Participant must begin making withdrawals from a VBA Account beginning the year they turn 72. Such withdrawals, and any transfers and Lump Sum Voluntary Transfers made in connection with the GP Participant's VBA Account must be made in accordance with this Article 9, in such manner, and on such timing and conditions acceptable to the Plan Manager, and subject to any limitations under the Act and the Income Tax Act. For greater certainty, such withdrawals and transfers shall be credited with a rate of return pursuant to Article 6 of this Appendix GP.

Section 9.02 **Deposits to Variable Benefit Account**

A GP Participant may make Lump Sum Voluntary Transfers to the Trust Fund which shall be credited to the GP Participant's Variable Benefit Account from another registered pension plan or registered retirement income fund, or any other transfer permitted by the Act and the Income Tax Act. In the event that any such transfer is locked-in under the Act, such funds shall continue to be treated as locked-in for the purposes of the Act under this Appendix GP.

Section 9.03 Minimum Amounts Withdrawn

Notwithstanding any other provision of this Appendix GP, a GP

Participant shall be required to withdraw in cash a minimum amount each
year from their Variable Benefit Account. For greater certainty, in no
event shall the amount withdrawn in a calendar year be less than the
amount determined under paragraph 8506(1)(e.1) and subsection 8506(5)

of the Income Tax Regulations.

Section 9.04 Maximum Amounts Withdrawn

Notwithstanding Section 9.03 of Appendix GP, in the event that any portion of funds in a Variable Benefit Account are locked in under the Act, then the annual amount withdrawn in respect of such funds shall respect any maximum withdrawal limitations under the Act.

Section 9.05 Tax Sheltered Transfers from Variable Benefit Account

In the event that a GP Participant is entitled to transfer funds from their Variable Benefit Account, such GP Participant shall be entitled to transfer such funds to a registered pension plan or a Registered Retirement Savings Arrangement in accordance with the Act and the Income Tax Act.

Article 10. DEATH BENEFITS

Section 10.01 Refund or Transfer of AVC Account Funds

Upon the death of a GP Participant prior to receipt of any payments from their Variable Benefit Account, the remaining balance in their AVC Account, credited with a rate of return pursuant to Article 6 of this Appendix GP, shall be payable to the GP Spouse. In the event that there is no GP Spouse or such entitlement is waived, the remaining balance shall be payable to the Beneficiary or if none, the estate of the GP Participant.

Notwithstanding the above and subject to any terms imposed by the Plan Manager, in the event the GP Spouse is entitled to the remaining balance and is entitled to an immediate or deferred survivor pension under the Design One Provisions, Design Two Provisions, or any other defined

benefit entitlement, the GP Spouse may defer receipt of the remaining balance, but in no event later than the end of the year the GP Spouse attains age 71, or such other age as required under the Income Tax Act.

All such payments shall be payable in cash where permitted by the Act.

However, subject to the Income Tax Act, the GP Spouse may elect to transfer their entitlement to a registered retirement savings arrangement in accordance with the Act.

Any refunds or transfers must be permitted pursuant to and in accordance with the Act and made in accordance with and subject to the Income Tax Act.

Section 10.02 Variable Benefit Account

- (a) Upon the death of a GP Participant who has a Variable Benefit

 Account and is entitled to immediately commence receipt of payments
 from the Variable Benefit Account, the remaining balance shall be
 payable in the order and manner as described in Section 10.01 above.
- (b) Notwithstanding the above, in the event that a GP Spouse who has not waived their entitlement under this Appendix GP has been designated a Specified Beneficiary, they shall be entitled to:
 - i. elect the option under Section 10.02(a); or,
 - ii. continue to receive benefits from the GP Participant's Variable

 Benefit Account, subject to any conditions imposed by the Plan

 Manager and the requirements of the Income Tax Act.
- (c) Upon the death of a Specified Beneficiary who elected option (ii) above, any balance remaining in the GP Participant's Variable Benefit Account shall be payable to the Beneficiary of the Specified Beneficiary, or if none, to the estate of the Specified Beneficiary.

Section 10.03 Locked-In Funds

Notwithstanding anything to the contrary in this Article 10 of Appendix GP, upon the death of a GP Participant, any funds in the GP Account which are locked-in under the Act shall be paid in accordance with the requirements of the Act.

APPENDIX H

TORSTAR CORPORATION AND SUBSIDIARIES

Effective October 1, 2018, Torstar Corporation, Toronto Star Newspapers Limited and Metroland Media Group Ltd. (the "Torstar DB Companies") became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Torstar Plans listed below ("Torstar Plans") are to be transferred to the Plan effective October 1, 2018.

- 1. Toronto Star Pension Plan (Registration No. 0346171)
- 2. Torstar Pension Plan (Registration No. 0969873)
- 3. Metroland Pension Plan (Registration No. 0394288)
- Pension Plan for Employees of Metroland West Media Group (Registration No. 1050871)
- Metroland Pension Plan for Senior Management Employees (Registration No. 0691295)
- 6. Metroland Employees' Retirement Savings Plan (Registration No. 0691287)
- 7. Retirement Plan for Employees of Metroland (Registration No. 0569939)
- The Pension Plan for Employees of the Guelph Mercury and the Cambridge Reporter,
 Divisions of the Metroland Media Group Ltd. (Registration No. 1050863)

This Appendix H shall apply with effect on and after the Effective Date in respect of the participation by the Torstar DB Companies in the Plan and the transfer of assets and liabilities from the Torstar Plans to the Plan, but shall not apply in respect of any employee of the Torstar DB Companies on and after the Effective Date who was not an Active Torstar Plan Member:

- The following words and phrases shall, for the purposes of this Appendix H, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Torstar Plan Benefits means the pension benefits accrued by or credited to each Active Torstar Plan Member under the terms of the applicable Torstar Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - Active Torstar Plan Member means any individual that is an employee of a
 Torstar DB Company who participates in a Torstar Plan or who otherwise
 participates in and is accruing benefits under a Torstar Plan immediately prior to

the Effective Date and who is not an Inactive Torstar Plan Member.

- c. Adjustment Factor means, in respect of an Active Torstar Plan Member who was a member of the Pension Plan for Employees of Metroland West Media Group (FSCO Registration No. 1050871), the most recent July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada.
- d. **Deferred Torstar Plan Member** means any person entitled to a deferred pension or any lump sum from a Torstar Plan (other than Active Torstar Plan Members entitled to a refund of OAC balances) as at the Effective Date.
- e. Effective Date means October 1, 2018.
- f. Inactive Accrued Torstar Plan Benefits means the pension benefits accrued by or credited to each Inactive Torstar Plan Member under the terms of a Torstar Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- g. Inactive Torstar Plan Member means any individual who is either a Retired Torstar Plan Member, a Deferred Torstar Plan Member or a Torstar Plan Beneficiary.
- h. OAC Balance means (i) as of the Transfer Date the amount credited to an Active Torstar Plan Member or Inactive Torstar Plan Member in respect of optional ancillary contributions under the provisions of the applicable Torstar Plan; and (ii) as of any point after the Transfer Date the amount so credited to an Active Torstar Plan Member or Inactive Torstar Plan Member under the Plan in respect of optional ancillary contributions as of the Transfer Date plus compound interest accruing at the Plan's annual discount rate in effect during the applicable period.
- Retired Torstar Plan Member means any person in receipt of a pension from a Torstar Plan as at the Effective Date.
- j. Transfer Date means the date determined by Torstar and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Torstar Plans are to be transferred from the Torstar Plans to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- k. **Torstar DB Companies** has the meaning set forth in the introductory paragraph of this Appendix H.
- Torstar Plan Beneficiary means any person who is entitled to a payment from
 or otherwise has an entitlement under a Torstar Plan other than Active Torstar
 Plan Members, Retired Torstar Plan Members or Deferred Torstar Plan
 Members.

m. Torstar Plans has the meaning set forth in the introductory paragraph of this Appendix H and any incorporation by reference into this Appendix H of the terms of a Torstar Plan shall mean the terms of the applicable Torstar Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Torstar Plan texts (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.

Any capitalized terms used in this Appendix H that are not defined herein, shall have the meaning ascribed to them under Appendix G of the Plan.

Active Torstar Plan Members

- 2. On and after the Effective Date, each Active Torstar Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix H; and
 - b. have any service credited to the Active Torstar Plan Member under the Torstar Plans recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Torstar Plan Member shall be required to contribute in respect of their participation under this Appendix H of the Plan based on the definition of Earnings under the applicable Torstar Plan in which such Active Torstar Plan Member was participating immediately before the Effective Date, but without any reference to a cap in contributions.
- 4. The contribution rates applicable in respect of Active Torstar Plan Members is as set out below:
 - a. Active Torstar Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix H in accordance with the following rates:,

Effective	Contribution Rate
For the period	
commencing on	
October 1, 2018	
through to and	7%

including December	
31, 2019	
For the period	
commencing on	
January 1, 2020	
through to and	
including December	
31, 2020	8%
On and after January	
1, 2021	9%

- b. The Torstar DB Companies shall match contributions in respect of such Active Torstar Plan Members; and,
- c. Notwithstanding the foregoing, any Active Torstar Plan Member who as of the Effective Date was on income replacement pursuant to a long-term disability benefit and was accruing benefits on a waiver of contributions under the terms of a Torstar Plan, shall, for so long as the member continues to qualify for long long-term disability under their Torstar DB Company's' benefit policy, have earnings correspond to the earnings used to determine the benefit accruing to such member immediately before the Effective Date under the applicable Torstar Plan and such individual shall not contribute to the Plan but instead the applicable Torstar DB Company shall pay double the applicable contribution rates provided in a. above.

Accrued Torstar Plan Benefits

5. As of the Effective Date, and subject to the provisions of this Appendix H, the Active Accrued Torstar Plan Benefit of each Active Torstar Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Torstar Plan Member his or her Active Accrued Torstar Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix H, the form of payment, optional form of payment and reductions, as provided for under the terms of

- the Torstar Plan in which such Active Torstar Plan Member participated immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Torstar Plan Member shall be:
 - a. The pension benefit accrued to the Active Torstar Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix H or otherwise credited to the person under the terms of the Plan other than any Active Accrued Torstar Plan Benefit credited to the Member pursuant to paragraphs 5 and 7 of this Appendix H, plus
 - b. The Active Torstar Plan Member's Active Accrued Torstar Plan Benefit credited to the Active Torstar Plan Member under the Plan pursuant to paragraphs 5 and 7 of this Appendix H.
- 7. For the purposes of this Appendix H, the Active Accrued Torstar Plan Benefit so credited in respect of an Active Torstar Plan Member who was immediately prior to the Effective Date a member of the Pension Plan for Employees of Metroland West Media Group (FSCO Registration No. 1050871) shall be enhanced each January 1 following January 1, 2019 and prior to his or her Termination of Membership Date by the Adjustment Factor.
- 8. Any portability options in respect of an Active Accrued Torstar Plan Benefit credited to an Active Torstar Plan Member under the Plan shall be in accordance with the terms of the Plan and not any Torstar Plan.
- 9. The determination of ancillary benefits in respect of an Active Torstar Plan Member's Active Accrued Torstar Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Torstar Plan Member under the applicable Torstar Plan immediately prior to the Effective Date
- 10. On and after the Transfer Date, the pension so payable to an Active Torstar Plan Member under the Plan following a Termination of Membership Date shall be adjusted on account of the application of any OAC Balance then credited to the Active Torstar Plan Member to purchase ancillary benefits in a manner consistent with the applicable Torstar Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the applicable Torstar Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Active Torstar Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfeited.

- 11. On the death of any Active Torstar Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a. of this Appendix H; and
 - b. be in accordance with the terms of the applicable Torstar Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b. of this Appendix H.
- 12. On the commencement of pension by an Active Torstar Member, the Active Accrued Torstar Plan Benefit, as adjusted pursuant to this Appendix H, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2019 and that until January 1, 2027 the inflation protection factor so determined and applied pursuant to Article 11 of Appendix G of the Plan as if such inflation protection factor was determined using 25% of the change in the Consumer Price Index.

Inactive Torstar Plan Members

- 13. As of the Effective Date, the Inactive Accrued Torstar Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the applicable Torstar Plan) in respect of each Inactive Torstar Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix H, and for the avoidance of doubt the provisions of the applicable Torstar Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix H. And for the avoidance of doubt, such provisions of the applicable Torstar Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, preretirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 14. Any portability options in respect of an Inactive Accrued Torstar Plan Benefit credited to an Inactive Torstar Plan Member under the Plan shall be in accordance with the terms of the Plan and not any Torstar Plan.
- 15. Any pension payable under Section 13 of this Appendix H, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2019 and that until January 1, 2027 the inflation protection factor so determined and applied pursuant to Article 11 of Appendix G of the Plan as if such inflation protection factor was determined using 25% of the change in the Consumer Price Index.
- 16. Any pension payable under Sections 13 of this Appendix H shall be adjusted on account of the application of any OAC Balance then credited to the Inactive Torstar Plan Member

to purchase ancillary benefits in a manner consistent with the applicable Torstar Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the applicable Torstar Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Inactive Torstar Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfeited.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 16 of this Appendix H prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Torstar Plan Member:
 - to the extent to which payment of such benefit has been made from a Torstar Plan prior to the Transfer Date; or
 - ii. under Sections 5, 6.b. or 7 of this Appendix H in respect of whom a Torstar Plan has ceased to have any liability prior to the Transfer Date;
 - b. in respect of any Inactive Torstar Plan Member:
 - i. to the extent to which payment has been made from a Torstar Plan prior to the Transfer Date in respect of such Inactive Torstar Plan Member; or
 - ii. in respect of whom a Torstar Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX I

CANADIAN PRESS ENTERPRISES INC.

Effective July 1, 2019, Canadian Press Enterprises Inc. (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Transferring Plans listed below (the "Transferring Plans") are to be transferred to the Plan effective July 1, 2019.

- Canadian Press Enterprises Inc. Pension Plan for Employees Represented by the Canadian Media Guild (Registration No. 1031848)
- 2. The Pension Plan of Canadian Press Enterprises Inc. (Registration No. 0237537)

This Appendix I shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plans to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix I, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the defined benefit pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the applicable Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the defined benefit provisions of a Transferring Plan and is accruing defined benefit benefits under a Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from a Transferring Plan under its defined benefit provisions as at the Effective Date.

- d. Effective Date means July 1, 2019.
- e. **EPP Adjustment** means, in respect of an Active Transferring Plan Member or an Inactive Transferring Plan Member, the pension amount determined by applying the EPP Balance, if any, maintained in respect of such Active Transferring Plan Member or an Inactive Transferring Plan Member under Section 2.01(c) of Appendix G as if such EPP Amount was a contribution made as of April 1, 2020 in accordance with Section 4.03 of Appendix G for periods of Purchasable Service. Provided further that:
 - i. any such EPP Adjustment shall be applied as if accrued by the Active Transferring Plan Member or Inactive Transferring Plan Member, as applicable, in the years 2009 through 2011, as applicable; and,
 - ii. any such EPP Adjustment shall only be applied if the past service pension adjustment in respect of such EPP Adjustment is exempt from certification under the Income Tax Act.
- f. **EPP Balance** means, in respect of an Active Transferring Plan Member or an Inactive Transferring Plan Member, the balance owing to the individual by the Transferring Employer, as at December 31, 2018, pursuant to its EPP program with such amounts adjusted with interest at 2% per annum to April 1, 2020 and which have not been paid to such member prior to April 1, 2020. For the avoidance of doubt, the EPP program maintained by the Transferring Employer is separate from and does not form part of either Transferring Plan.
- g. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the defined benefit terms of the applicable Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- h. **Inactive Transferring Plan Member** means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- i. OAC Balance means (i) as of the Transfer Date the amount credited to an Active Transferring Plan Member or Inactive Transferring Plan Member in respect of optional ancillary contributions under the provisions of the applicable Transferring Plan; and (ii) as of any point after the Transfer Date the amount so credited to an Active Transferring Plan Member or Inactive Transferring Plan Member under the Plan in respect of optional ancillary contributions as of the Transfer Date plus compound interest accruing at the Plan's annual discount rate in effect during the applicable period.
- j. Retired Transferring Plan Member means any person in receipt of a pension from a Transferring Plan under its defined benefit provisions as at the Effective

Date.

- k. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix I.

m. Transferring Plans

has the meaning set forth in the introductory paragraph of this Appendix I and any incorporation by reference into this Appendix I of the terms of a Transferring Plan shall mean the terms of the applicable Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan texts (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan, provided however that the terms of the Transferring Plans shall read as if:

- i. Section B4.03 of each Transferring Plan was amended to add "On and after July 1, 2019, Employer consent shall be deemed given for the purposes of the section "Early Retirement Pension for Pre-2008 Service with Employer Consent" that follows." to the end of the first paragraph.
- ii. Section B4.03(a) of each Transferring Plan was deleted and replaced with:
 - a)1. If the Member, at the termination of his
 Continuous Service, has completed 2 years
 of Continuous Service and has attained at
 least age 50, the pension shall be reduced
 by 1/4 of one percent for each complete
 month by which the date that is the later of
 the date that the pension commences or the
 date the member would attain age 55
 precedes the earlier of:

i)a. the Normal Retirement Date of the Member; or ii)b. the earliest date on which, had the Member remained in the service of the Employer, he would have been eligible for a pension calculated in accordance with subsection (b) below.

And in the event such Member commences his pension prior to the attainment of age 55, the pension so paid shall be reduced such that

Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 2.38" + Indent at: 2.5"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 2.75" + Indent at: 3"

Formatted: Indent: Left: 2.25", Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 3.25" + Indent at: 3.5"

the value, as of the date on which the pension starts, is the Actuarial Equivalent of the pension, reduced as described above, but commencing as of the end of the month in which the member would attain age 55."

- **iii.** Section B4.04(a) of each Transferring Plan was deleted and replaced by:
 - a)1. A Member who has elected to retire early and meets the conditions described in Section B4.03(a) shall, with the consent of the Employer, be entitled to a bridge benefit in respect of Pre-2008 Credited Service. A Member who has elected to retire early and meets the conditions described in Section B4.03(a) shall be entitled to a bridge benefit in respect of Post-2007 Credited Service and no Employer consent is required. Such bridge benefit shall equal \$20 per month per year of applicable Credited Service (Post-2007 Credited Service minus any Credited Service for 2009, 2010 and 2011 while an Active Member, plus Pre-2008 Credited Service if the Employer has given consent) to a maximum of \$700 per month. The bridge benefit shall be reduced 1/4 of 1% percent for each complete month by which the date that is the later of the date that the pension commences or the date the Member would attain age 55 precedes the earlier of:
- i)a. the date on which the Member would have attained age 60 had the Member remained in service with the Employer, and where the Member's age (to the last completed month) plus Continuous Service (to the last completed month) would have totalled 85 years or more, or

ii)b. ___age 65.

And in the event such Member commences his pension prior to the attainment of age 55, the bridge benefit so paid shall be reduced such that the value, as of the date on which the pension starts, is the Actuarial Equivalent of the bridge benefit, reduced as described above, but commencing as of the end of the month in which the member would attain age 55."; and,

Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 2.38" + Indent at: 2.5"

Formatted: Numbered + Level: 4 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 2.75" + Indent at: 3"

Formatted: Indent: Left: 2.25", Numbered + Level: 5 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 3.25" + Indent at: 3.5"

iv. Section B4.04 of each Transferring Plan was amended by adding the following: Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 2.38" + Indent at: 2.5"

- "(f) On and after July 1, 2019, Employer consent for the purposes of (a) and (b) above shall be deemed given."
- n. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the defined benefit provisions of a Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix I that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix I; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plans for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix I of the Plan based on the definition of Annual Earnings under Appendix G of the Plan.
- 4. The contribution rate applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix I in accordance with the following rate:

Effective	Contribution Rate

For the period	
commencing on July 1,	
2019	7%

and

 The Transferring Employer shall match contributions in respect of such Active Transferring Plan Members.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix I, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit, including any EPP Adjustment effective as of April 1, 2020, shall be payable from the Plan subject to, unless otherwise provided for under this Appendix I, the normal form of payment, optional forms of payment and reductions, as provided for under the terms of the applicable Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date, provided that no bridge benefit is payable in respect of the EPP Adjustment. For the further avoidance of doubt, the Active Accrued Transferring Plan Benefit, including any EPP Adjustment effective April 1, 2020, of each Active Transferring Plan Member shall not be subject to any adjustment under the Schedule 1 to Appendix G of the Plan.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix I or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section 5 of this Appendix I, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section 5 of this Appendix I.
- 7. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance

with the terms of the Plan and not any Transferring Plan.

- 8. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the applicable Transferring Plan immediately prior to the Effective Date.
- 9. On and after the Transfer Date, the pension so payable to an Active Transferring Plan Member under the Plan following a Termination of Membership Date shall be adjusted on account of the application of any OAC Balance then credited to the Active Transferring Plan Member to purchase ancillary benefits in a manner consistent with the applicable Transferring Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the applicable Transferring Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Active Transferring Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfaited.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a of this Appendix I; and
 - b. be in accordance with the terms of the applicable Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix I.
- 11. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, earned on and after January 1, 2008 but prior to the Effective Date, including the application of the EPP Adjustment, if any, as adjusted pursuant to this Appendix I, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase by a rate of 1.5% pursuant to the terms of the applicable Transferring Plan and shall not be subject to the indexation provisions under Article 11 of Appendix G to the Plan. For the avoidance of doubt, any pension benefit accrued under Section 2 of this Appendix I will be subject to the indexation provisions under Article 11 of Appendix G.

Inactive Transferring Plan Members

- 12. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the applicable Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix I, and for the avoidance of doubt the provisions of the applicable Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix I. And for the further avoidance of doubt, such provisions of the applicable Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 13. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not any Transferring Plan.
- 14. Any pension payable under Section 12 of this Appendix I earned on and after January 1, 2008 but prior to the Effective Date, once in payment after the Effective Date, after taking in to account any adjustments pursuant to Section 15 of this Appendix I, shall increase by a rate of 1.5% pursuant to the terms of the applicable Transferring Plan. For the avoidance of doubt, such pension payable under Section 12 of this Appendix I, including as adjusted pursuant to Section 15 of this Appendix I, shall not be subject to the indexation provisions under Article 11 of Appendix G to the Plan.
- 15. Any pension payable under Sections 12 of this Appendix I shall be adjusted on account of:
 - a. any EPP Adjustment effective as of April 1, 2020, as may be applicable to the Inactive Transferring Plan Member subject to, unless otherwise provided for under this Appendix I, the form of payment, optional form of payment and reductions, as provided for under the terms of the applicable Transferring Plan in which such Inactive Transferring Plan Member participated, provided that no bridge benefit is payable in respect of the EPP Adjustment and that no early retirement reduction shall be applied if the Inactive Transferring Member is over age 65 as of April 1, 2020; and,
 - b. the application of any OAC Balance then credited to the Inactive Transferring Plan Member to purchase ancillary benefits in a manner consistent with the applicable Transferring Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the Transferring Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Inactive Transferring Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfeited.

Transition

- 16. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 16 of this Appendix I prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - c. in respect of any Active Transferring Plan Member:
 - i. to the extent to which payment of such benefit has been made from a Transferring Plan prior to the Transfer Date; or
 - ii. under Sections 5 and 6.b of this Appendix I in respect of whom a Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - d. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from a Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom a Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX J

POSTMEDIA NETWORK INC.

Effective July 1, 2019, Postmedia Network Inc. (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Transferring Plans listed below (the "Transferring Plans") are to be transferred to the Plan effective July 1, 2019.

- 1. Postmedia Network Inc. DB Plan A (Registration No.: 1279223)
- 2. Postmedia Network Inc. DB Plan B (Formerly the Union LFP Plan) (Registration No.: 1278225)
- 3. Postmedia Network Inc. DB Plan for Employees of Osprey (Registration No.: 1278357)
- 4. Postmedia Network Inc. DB Plan for Employees of the Stratford Beacon Herald (Registration No.: 1278340)
- 5. Postmedia Network Inc Retirement Plan (Registration No.: 1077049); and
- 6. The Windsor Star Retirement Plan (Registration No.: 1077064).

This Appendix J shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of defined benefit related assets and liabilities from the Transferring Plans to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix J, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the defined benefit terms of the applicable Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual who as of immediately prior to the Effective Date was participating and accruing benefits under the defined benefit provisions of a Transferring Plan and who is not an Inactive Transferring Plan Member.

- c. Deferred Transferring Plan Member means any individual as at the Effective Date entitled to a deferred pension or any lump sum from a Transferring Plan in respect of the defined benefit provisions thereunder, but excluding any individual whose benefits under a Transferring Plan are subject to a Transfer to another pension plan pursuant to a the sale of a portion of the Transferring Employer's business to a Metroland Media Group Ltd.
- d. Effective Date means July 1, 2019.
- e. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the applicable Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- f. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- g. OAC Balance means (i) as of the Transfer Date the amount credited to an Active Transferring Plan Member or Inactive Transferring Plan Member in respect of optional ancillary contributions under the provisions of the applicable Transferring Plan; and (ii) as of any point after the Transfer Date the amount so credited to an Active Transferring Plan Member or Inactive Transferring Plan Member under the Plan in respect of optional ancillary contributions as of the Transfer Date plus compound interest accruing at the Plan's annual discount rate in effect during the applicable period.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from a Transferring Plan as at the Effective Date in respect of the defined benefit provisions of a Transferring Plan.
- i. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plans are to be transferred from the Transferring Plans to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- j. Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix J.
- k. Transferring Plans has the meaning set forth in the introductory paragraph of this Appendix J and any incorporation by reference into this Appendix J of the terms of the Transferring Plans shall mean the terms of the applicable Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plans texts (and any amendments thereto) are filed by the Plan pursuant to the Act as documents

supporting the Plan.

 Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the defined benefit provisions of a Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix J that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix J; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plans for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member, including an Active Transferring Plan Member who, following the Effective Date, commences receipt of income replacement pursuant to a long-term disability policy, shall be required to contribute in respect of their participation under this Appendix J, based on the definition of Earnings under the applicable Transferring Plan in which such Active Transferring Plan Member was participating immediately before the Effective Date; and
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix J in accordance with the following rate(s):

Effective	Contribution Rate
For the period	
commencing on and	
after July 1, 2019	5%

- The Transferring Employer shall match contributions in respect of such Active Transferring Plan Members.
- c. Any Active Transferring Plan Member who as of the Effective Date was on income replacement pursuant to a long-term disability benefit and was accruing benefits on a waiver of contributions under the terms of the applicable Transferring Plan shall not be required to contribute to the Plan, but the Transferring Employer shall contribute to the Plan for so long as the member continues to qualify for long-term disability under the Transferring Employer's disability benefit policy at a rate of 10% of the earnings used to determine the benefit accruing to such member under the applicable Transferring Plan immediately prior to the Effective Date.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix J, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix J, the form of payment, optional form of payment and reductions, as provided for under the terms of the applicable Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix J or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section 5 of this Appendix J, plus
 - The Active Transferring Plan Member's Active Accrued Transferring Plan
 Benefit credited to the Active Transferring Plan Member under the Plan pursuant
 to Sections 5 of this Appendix J.

- Any portability options in respect of an Active Accrued Transferring Plan Benefit
 credited to an Active Transferring Plan Member under the Plan shall be in accordance
 with the terms of the Plan and not any Transferring Plan.
- 8. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the applicable Transferring Plan immediately prior to the Effective Date.
- 9. On and after the Transfer Date, the pension so payable to an Active Transferring Plan Member under the Plan following a Termination of Membership Date shall be adjusted on account of the application of any OAC Balance then credited to the Active Transferring Plan Member to purchase ancillary benefits in a manner consistent with the applicable Transferring Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the applicable Transferring Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Active Transferring Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfeited.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a of this Appendix J; and
 - be in accordance with the terms of the applicable Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix J.
- 11. No indexation will be applied to the Active Accrued Transferring Plan Benefit upon commencement of a pension by an Active Transferring Plan Member other than any indexation applicable under the terms of the applicable Transferring Plan, and, for the avoidance of doubt, any pension benefit accrued under Section 2 of this Appendix J will be subject to the indexation provisions under Article 11 of Appendix G.

Inactive Transferring Plan Members

12. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the applicable Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix J, and for the avoidance of doubt the provisions of the applicable Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix J. And for the avoidance of doubt, such provisions of the applicable Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 13. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not any Transferring Plan.
- 14. No indexation will be applied to pensions payable under Section 12 of this Appendix J once in payment after the Effective Date other than any indexation applicable under the terms of the applicable Transferring Plan.
- 15. Any pension payable under Section 12 of this Appendix J shall be adjusted on account of the application of any OAC Balance then credited to the Inactive Transferring Plan Member to purchase ancillary benefits in a manner consistent with the applicable Transferring Plan as at the Effective Date (but without applying any restrictions on post-retirement indexation of benefits that may have existed under the Transferring Plan) and within the maximum limits permitted under the Income Tax Act (Canada) in respect of any and all benefits payable to such Inactive Transferring Plan Member under the Plan using then applicable CIA commuted value standards. Any amount that cannot be so applied shall be forfeited under the Plan unless it is then permitted under Applicable Laws for such amount to be paid from the Plan to such person in which case such amount shall be paid to the person and shall not be considered forfeited.

Transition

- 16. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5-15 of this Appendix J prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from a Transferring Plan prior to the Transfer Date; or
 - under Sections 5 and 6.b of this Appendix J in respect of whom a Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:

- i. to the extent to which payment has been made from a Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
- ii. in respect of whom a Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX K

COMMUNITY LIVING TORONTO (NON-EXECUTIVE MEMBERS)

Effective October 1, 2019, Community Living Toronto (the "Transferring Employer") became a participating employer in the Plan for the purposes of Appendix A.1 in respect of certain of its employees under an agreement whereby the assets and liabilities of the Pension Plan for Employees of Community Living Toronto (the "Transferring Plan") (Registration No. 0279513) are to be transferred to the Plan effective October 1, 2019.

Subject to the approval of the transfer of assets and liabilities from the Transferring Plan to the Plan, this Appendix K shall apply with effect on and after the Effective Date (as defined below) in respect of the participation by the Transferring Employer in the Plan in respect of certain of its employees and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member (as defined below):

- The following words and phrases shall, for the purposes of this Appendix K, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the *Income Tax Act* (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor.
 - d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
 - e. Effective Date means October 1, 2019.
 - f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the

terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the *Income Tax Act* (Canada).

- g. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- i. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary approvals required from the CEO of the Financial Services Regulatory Authority of Ontario in order to enable the transfer of pension assets and liabilities, in accordance with the Act and applicable regulations thereunder, on which the assets and liabilities of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by O. Reg. 311/15.
- j. **Transferring Employer** has the meaning set forth in the introductory paragraph of this Appendix K.
- k. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix K and any incorporation by reference into this Appendix K of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the Effective Date. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.
- m. Transition Date means the date an Active Transferring Plan Member becomes employed with the Transferring Employer in a full-time executive position.

Any capitalized terms used in this Appendix K that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix K; and

- b. have any service credited to the Active Transferring Plan Member under the Transferring Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix K, based on the definition of Plan Earnings, as such term is defined under the Transferring Plan in effect immediately before the Effective Date.
- 4. The contribution rates applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members who are subject to Part A of the Transferring Plan shall contribute based on their Plan Earnings as set out in Section 3 of this Appendix K in accordance with the following rates:

Effective	Contribution Rate
For the period	
commencing on	
October 1, 2019	9%

and

b. Active Transferring Plan Members who are subject to Part B of the Transferring Plan shall contribute based on their Plan Earnings as set out in Section 3 of this Appendix K in accordance with the following rate(s):

Effective	Contribution Rate
For the period	
commencing on	
October 1, 2019	6.75%

and

c. The Transferring Employer shall match contributions in respect of such Active Transferring Plan Members in accordance with Plan terms.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix K, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix K, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in effect immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix K or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Sections 5 and 7 of this Appendix K, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 5 and 7 of this Appendix K.
- 7. For the purposes of this Appendix K, the Active Accrued Transferring Plan Benefit shall be enhanced by the Adjustment Factor on and after January 1, 2021 and every January 1 thereafter, prior to his or her Termination of Membership Date.
- 8. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 9. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:

- a. be in accordance with the terms of the Plan in respect of the benefit determined under Sections 6.a of this Appendix K; and
- b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix K.
- 11. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix K, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Inactive Transferring Plan Members

- 12. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix K, and for the avoidance of doubt: (i) the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix K; and (ii)such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 13. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 14. Any pension payable under Section 12 of this Appendix K, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Transition

- 15. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 14 of this Appendix K prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - ii. under Sections 5, 6.b and 7 of this Appendix K in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer

Date; and,

- b. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.
- 16. In the event that an Active Transferring Plan Member ceases to be employed in a non-executive position but continues employment in a full-time executive position with the Transferring Employer, such Active Transferring Plan Member shall cease participating in and accruing a benefit under the Design Two Provisions of the Plan and commence participating in and accruing a benefit under the Design One Provisions of the Plan, subject to the provisions of Appendix L, on and after the Transition Date in accordance with Section 12.04 of Appendix G of the Plan.
- 17. For the avoidance of doubt, on and after the Transition Date, the total pension benefit entitlement under the Plan in respect of such Active Transferring Plan Member described in Section 16 of this Appendix K shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Design Two Provisions of the Plan on and after the Effective Date and up to the Transition Date, pursuant to Section 2 of this Appendix K, plus
 - b. The pension benefit accrued to the Active Transferring Plan Member under the Design One Provisions of the Plan on and after the Transition Date, pursuant to Section 16 of this Appendix K, plus
 - c. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 5 and 7 of this Appendix K.

APPENDIX L

COMMUNITY LIVING TORONTO (EXECUTIVE MEMBERS)

Effective October 1, 2019, Community Living Toronto (the "Transferring Employer") became a participating employer in the Plan for the purposes of Appendix A in respect of certain of its employees under an agreement whereby the assets and liabilities of the Pension Plan for Executive Employees of Community Living Toronto (the "Transferring Plan") (Registration No. 0691873) are to be transferred to the Plan effective October 1, 2019.

Subject to the approval of the transfer of assets and liabilities from the Transferring Plan to the Plan, this Appendix L shall apply with effect on and after the Effective Date (as defined below) in respect of the participation by the Transferring Employer in the Plan in respect of certain of its employees and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date (as defined below) who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix L, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the *Income Tax Act* (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor.
 - d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
 - e. Effective Date means October 1, 2019.
 - f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to

the maximum pension benefit provisions of the *Income Tax Act* (Canada).

- g. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- i. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary approvals required from the CEO of the Financial Services Regulatory Authority of Ontario in order to enable the transfer of pension assets and liabilities, in accordance with the Act and applicable regulations thereunder, on which the assets and liabilities of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by O. Reg. 311/15.
- j. Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix L.
- k. **Transferring Plan** has the meaning set forth in the introductory paragraph of this Appendix L and any incorporation by reference into this Appendix L of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the Effective Date. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.
- m. Transition Date means the date an Active Transferring Plan Member becomes employed with the Transferring Employer in an other than regular full-time executive position or in a non-executive position.

Any capitalized terms used in this Appendix L that are not defined herein, shall have the meaning ascribed to them under Article 2 of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Design One Provisions
 of the Plan for service on and after the Effective Date in accordance with the
 terms of the Plan, subject to the provisions of this Appendix L; and

- b. have any service credited to the Active Transferring Plan Member under the Transferring Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, Active Transferring Plan Members shall be required to contribute to the Design One Provisions of the Plan in accordance with the terms of the Plan and the Transferring Employer shall match such contributions in accordance with the terms of the Plan, as may be amended from time to time.

Accrued Transferring Plan Benefits

- 4. As of the Effective Date, and subject to the provisions of this Appendix L, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix L, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in effect immediately prior to the Effective Date.
- 5. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix L or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Sections 4 and 6 of this Appendix L, plus
 - The Active Transferring Plan Member's Active Accrued Transferring Plan
 Benefit credited to the Active Transferring Plan Member under the Plan pursuant
 to Sections 4 and 6 of this Appendix L.
- 6. For the purposes of this Appendix L, the Active Accrued Transferring Plan Benefit shall be enhanced by the Adjustment Factor on and after January 1, 2021 and each January 1 thereafter, prior to his or her Termination of Membership Date.
- 7. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take

into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.

- On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 5.a of this Appendix L; and
 - be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 5.b of this Appendix L.
- 10. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix L, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 14 of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Inactive Transferring Plan Members

- 11. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix L, and for the avoidance of doubt: (i) the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix L; and (ii) such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 12. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 13. Any pension payable under Section 11 of this Appendix L, once in payment after the Effective Date, shall increase pursuant to Article 14 of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Transition

14. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 4 to 13 of this Appendix L prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:

- c. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - under Sections 4, 5.b and 6 of this Appendix L in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
- d. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.
- 15. In the event that an Active Transferring Plan Member ceases to be employed in a full-time executive position but continues as an employee of the Transferring Employer eligible for membership under the Design Two Provisions, such Active Transferring Plan Member shall cease participating in and accruing a benefit under the Design One Provisions of the Plan and commence participating in and accruing a benefit under the Design Two Provisions of the Plan, subject to the provisions of Appendix K, on and after the Transition Date, in accordance with Section 12.03 of Appendix G of the Plan, and in accordance with Section B.(9) of Appendix A.1 of the Plan.
- 16. For the avoidance of doubt, on and after the Transition Date, the total pension benefit entitlement under the Plan in respect of such Active Transferring Plan Member described in Section 15 of this Appendix L shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Design One Provisions of the Plan on and after the Effective Date and up to the Transition Date, pursuant to Section 2 of this Appendix L, plus
 - The pension benefit accrued to the Active Transferring Plan Member under the Design Two Provisions of the Plan on and after the Transition Date, pursuant to Section 15 of this Appendix L, plus
 - c. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 4 and 6 of this Appendix L.

APPENDIX M

UNITED WAY OF GREATER TORONTO

Effective January 1, 2020, United Way of Greater Toronto (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Retirement Plan for Employees of United Way of Greater Toronto (the "Transferring Plan") (Registration Number 0316240) are to be transferred to the Plan effective January 1, 2020.

This Appendix M shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of defined benefit assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix M, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the defined benefit pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor.
 - d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
 - e. **Effective Date** means January 1, 2020.

- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- g. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- i. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- j. Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix M.
- k. **Transferring Plan** has the meaning set forth in the introductory paragraph of this Appendix M and any incorporation by reference into this Appendix M of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix M that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix M; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to

benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.

- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix M of the Plan based on the definition of Annual Earnings under Appendix G of the Plan.
- 4. The contribution rates applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix M in accordance with the following rates:

Effective	Contribution Rate
From January 1, 2020 to	
December 31, 2020	4%
From January 1, 2021 to	
December 31, 2021	5%
On and after January 1,	
2022	6%

and

 b. The Transferring Employer shall contribute on Active Transferring Plan Members' Annual Earnings as set out in Section 3 of this Appendix M in accordance with the following rates:

Effective	Contribution Rate
From January 1, 2020 to	
December 31, 2020	8%
From January 1, 2021 to	
December 31, 2021	7%

On and after January 1,	
2022	6%

c. Any Active Transferring Plan Member who, as of the Effective Date, was on income replacement pursuant to a long-term disability benefit and was accruing benefits on a waiver of contributions under the terms of the Transferring Plan shall not be required to contribute to the Plan. The Transferring Employer shall make the required Employer and Employee contributions in accordance with the rates above for so long as the member continues to qualify for long-term disability under the Transferring Employer's disability benefit policy. Earnings used to determine such contributions shall be the earnings used to determine the benefit accruing to such member under the Transferring Plan immediately prior to the Effective Date.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix M, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix M, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix M or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Sections 5 and 7 of this Appendix M, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 5 and 7 of this Appendix M.
- 7. For the purposes of this Appendix M, the Active Accrued Transferring Plan Benefit shall be enhanced on and after the first January following the Effective Date, and each January

- 1st thereafter, prior to his or her Termination of Membership Date by the Adjustment Factor.
- 8. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 9. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a of this Appendix M; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix M.
- 11. On the commencement of a pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix M, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to the indexation provision under Article 11 of Appendix G of the Plan, provided that no such increase shall be made in respect of the period prior to the first January 1st following the Effective Date.
- 12. Notwithstanding Section 11 of this Appendix M, in no event shall the cumulative amount of indexing, if any, provided pursuant to Section 11 of this Appendix M to Active Transferring Plan Members be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the Transferring Plan in effect immediately prior to the Effective Date.

Inactive Transferring Plan Members

13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix M, and for the avoidance of doubt the provisions of the Transferring Plan in effect immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix M. For the further avoidance of

- doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, preretirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 14. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 15. Any pension payable under Section 13 of this Appendix M, once in payment after the Effective date, shall increase pursuant to the indexation provision under Article 11 of Appendix G of the Plan, provided that no such increase shall be made in respect of the period prior to the first January 1st following the Effective Date.
- 16. Notwithstanding Section 15 of this Appendix M, in no event shall the cumulative amount of indexing, if any, provided pursuant to Section 15 of this Appendix M to Active Transferring Plan Members be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the Transferring Plan in effect immediately prior to the Effective Date.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 16 of this Appendix M prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - under Sections 5, 6.b and 7 of this Appendix M in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX N

FP CANADIAN NEWSPAPERS LIMITED PARTNERSHIP

Effective January 1, 2020, FP Canadian Newspapers Limited Partnership (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Retirement Plan for Employees of Winnipeg Free Press and Canstar Community News (the "Transferring Plan") (CRA Registration Number 1076264) are to be transferred to the Plan effective January 1, 2020.

This Appendix N shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

The following words and phrases shall, for the purposes of this Appendix N, have the following meanings respectively, unless a different meaning is plainly required by the context:

- a. Active Accrued Transferring Plan Benefits means the pension of benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
- c. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
- d. Effective Date means January 1, 2020.
- e. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

(Canada).

- f. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- g. Retired Transferring Plan Member means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- h. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix N.
- j. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix N and any incorporation by reference into this Appendix N of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- k. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix N that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

19.7. On and after the Effective Date, each Active Transferring Plan Member shall:

a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix N; and

Formatted: Indent: Left: 0.69", Hanging: 0.3", Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.

20.8. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix N, based on the definition of Earnings under the Transferring Plan for both classes of members, but without any reference to a cap in contributions.

 $\begin{array}{lll} \textbf{Formatted:} & \textbf{Indent: Left:} & 0.25", \textbf{Numbered + Level:} & 1 + \\ \textbf{Numbering Style:} & 1, 2, 3, ... + \textbf{Start at:} & 1 + \textbf{Alignment: Left +} \\ \textbf{Aligned at:} & 1.25" + \textbf{Indent at:} & 1.5" \end{array}$

21.9. The contribution rates applicable in respect of Active Transferring Plan Members is as set out below:

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

 a. Active Transferring Plan Members shall contribute on their Earnings as set out in Section 3 of this Appendix N in accordance with the following rates: Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Effective	Contribution	Deemed
	Rate	Contribution Rate
From January 1, 2020	4%	5%
to December 31, 2021	470	
On and after January 1,	5%	5%
2022	3%	270

and

The Transferring Employer shall contribute on Active
 Transferring Plan Members' Earnings as set out in Section 3 of
 this Appendix N in accordance with the following rates:

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Effective	Contribution Rate	Deemed Contribution
		Rate
From January 1, 2020	0%	5%
to December 31, 2020		3,3

On and after January 1,	5%	5%
2021		3%

- c. For the avoidance of doubt, Active Transferring Plan Members shall contribute an amount in accordance with the Active Transferring Plan Member Contribution Rate column above, and the Transferring Employer shall contribute based on the Transferring Employer Contribution Rate column above but such Active Transferring Plan Members shall accrue benefits under the Plan based on the combined Active Transferring Plan Member Deemed Contribution Rate and the Transferring Employer Deemed Contribution Rate.
- d. All Active Transferring Plan Members who, as of the Effective Date, were on income replacement pursuant to a long-term disability benefit and were accruing benefit on a waiver of contributions under the terms of the Transferring Plan shall not be required to contribute to the Plan, but the Employer shall make Employer and Active Transferring Plan Member contributions in accordance with the applicable table above for so long as such member continues to qualify for long-term disability under the Employer's disability policy and the Earnings in respect of such member on which such contributions are based shall correspond to the earnings used to determine the benefit accruing to such member under the Transferring Plan immediately before the Effective Date.

Accrued Transferring Plan Benefits

As of the Effective Date, and subject to the provisions of this Appendix N, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix N, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.

23-11. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:

 $\begin{array}{lll} \textbf{Formatted:} & \textbf{Indent: Left:} & 0.25", \textbf{Numbered + Level:} & 1 + \\ \textbf{Numbering Style:} & 1, 2, 3, ... + \textbf{Start at:} & 1 + \textbf{Alignment: Left +} \\ \textbf{Aligned at:} & 1.25" + \textbf{Indent at:} & 1.5" \end{array}$

- a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix N or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section 5 of this Appendix N, plus
- Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"
- b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section 5 of this Appendix N.
- 24.12. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

- 25.13. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 26.14. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a of this Appendix N; and
 - be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix N.
- 27.15. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix N, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Inactive Transferring Plan Members

28-16. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix N, and for the avoidance of doubt the provisions of the Transferring

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix N. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, preretirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 29.17. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 30.18. Any pension payable under Section 11 of this Appendix N, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2021.

Transition

- 31.19. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 13 of this Appendix N prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - e. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - under Sections 5 and 6.b of this Appendix N in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - f. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

APPENDIX O

CATHOLIC CHARITIES OF THE ARCHDIOCESE OF TORONTO AND PARTICIPATING MEMBER AGENCIES

Effective January 1, 2020, the Transferring Employers listed below became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Pension Plan for Employees of Catholic Charities of the Archdiocese of Toronto and Participating Member Agencies (the "Transferring Plan") (Registration Number 238279) are to be transferred to the Plan effective January 1, 2020.

- 1. Catholic Charities of the Archdiocese of Toronto;
- 2. Catholic Cross Cultural Services;
- 3. Catholic Family Services of Durham;
- 4. Catholic Family Services of Peel-Dufferin;
- 5. Catholic Family Services of Simcoe County;
- 6. Catholic Family Services of Toronto;
- 7. Catholic Settlement House Day Nursery;
- 8. Mary Centre of the Archdiocese of Toronto;
- 9. Our Place Community of Hope Centre (Metropolitan Toronto);
- 10. Rosalie Hall;
- 11. Sancta Maria House;
- 12. Silent Voice Canada Inc.;
- 13. St. Michael's Homes;
- 14. LA Centre for Active Living; and
- 15. Rose of Durham.

This Appendix O shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employers in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employers on and after the Effective Date who was not an Active Transferring Plan Member:

1. The following words and phrases shall, for the purposes of this Appendix O, have the following meanings respectively, unless a different meaning is plainly required by the

context:

- a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- b. Active Transferring Plan Member means any individual that is an employee of any Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
- c. Adjustment Factor means an annual increase of 2.25% per year from the first January 1st following or coincident with the first anniversary of the Effective Date provided that such increase would not result in a past service pension adjustment under, or be restricted by, the Income Tax Act (Canada). If in any year a past service pension adjustment and/or a restriction would occur the amount of the increase will be reduced to avoid that adjustment and/or restriction and the amount of the foregone increase is to be carried forward to the next year or any subsequent year when it would be permitted without an adjustment.
- d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
- e. Effective Date means January 1, 2020.
- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- g. **Inactive Transferring Plan Member** means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- i. Transfer Date means the date determined by the Transferring Employers and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- Transferring Employers has the meaning set forth in the introductory paragraph of this Appendix O.

- k. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix O and any incorporation by reference into this Appendix O of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix O that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix O; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- On and after the Effective Date, any Active Transferring Plan Member shall be required
 to contribute in respect of their participation under this Appendix O of the Plan based on
 the definition of Annual Earnings under Appendix G.
- 4. The contribution rates applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their Annual Earnings as set out in Section 3 of this Appendix O in accordance with the following rates:

Effective	Contribution	Deemed
	Rate	Contribution Rate
From January 1, 2020	5.0%	7.0%
to December 31, 2020		1.376

5.5%	7.0%
6.0%	7.0%
6.5%	7.0%
7.0%	7.0%
	6.0%

and

- The Transferring Employer shall contribute on Active Transferring Plan
 Members' Annual Earnings as set out in Section 3 of this Appendix O at a rate of 7%.
- c. For the avoidance of doubt, Active Transferring Plan Members shall contribute an amount in accordance with the Active Transferring Plan Member Contribution Rate column above, but such Active Transferring Plan Members shall accrue benefits under the Plan based on the combined Active Transferring Plan Member Deemed Contribution Rate and the Transferring Employer Contribution Rate.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix O, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix O, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:

- a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix O or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Sections 5 and 7 of this Appendix O, plus
- b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 5 and 7 of this Appendix O.
- 7. For the purposes of this Appendix O, the Active Accrued Transferring Plan Benefit shall be enhanced on January 1, 2021, and each January 1st thereafter, prior to his or her Termination of Membership Date by the Adjustment Factor.
- 8. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 9. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a of this Appendix O; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b of this Appendix O.
- 11. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix O, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2021.

Inactive Transferring Plan Members

12. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix O, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix O. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 13. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 14. Any pension payable under Section 12 of this Appendix O, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2021.

Transition

- 15. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 14 of this Appendix O prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - under Sections 5, 6.b and 7 of this Appendix O in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX P

THE PRIORY OF CANADA OF THE MOST VENERABLE ORDER OF THE HOSPITAL OF ST. JOHN OF JERUSALEM AND PARTICIPATING COUNCILS

Effective April 1, 2021, the Transferring Employers listed below (the "Transferring Employers") became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the defined benefit assets and liabilities of the Pension Plan for Employees of the Priory of Canada of the Most Venerable Order of the Hospital of St. John of Jerusalem (the "Transferring Plan") (Registration Number 0299289) are to be transferred to the Plan effective April 1, 2021.

- 1. L'Ordre de Saint-Jean, Conseil du Quebec;
- 2. The Priory of Canada of the Most Venerable Order of the Hospital of St. John of Jerusalem;
- 3. St. John Council for Alberta;
- 4. St. John Council for Federal District;
- 5. St. John Council for Manitoba, Northwest Territories and Nunavut;
- 6. St. John Council for New Brunswick Inc.;
- 7. St. John Council for Newfoundland Properties Inc.;
- 8. St. John Council for Nova Scotia/Prince Edward Island;
- 9. St. John Council for Ontario;
- 10. St. John Council for Saskatchewan Inc.; and
- 11. St. John Society (British Columbia & Yukon).

This Appendix P shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employers in the Plan and the transfer of defined benefit assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employers on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix P, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the

Formatted: French (France)

maximum pension benefit provisions of the Income Tax Act (Canada).

- b. Active Transferring Plan Member means any individual that is an employee of a Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member. Notwithstanding the foregoing, Active Transferring Plan Member does not include Quebec employees who were participating in the Transferring Plan immediately prior to the Effective Date.
- c. Annuitized Transferring Plan Member means those individuals who ceased to be in active membership in the Transferring Plan and whose pension benefit accrued under the Transferring Plan is being, or will be prior to the Transfer Date, paid through an annuity from an insurance company purchased by the Transferring Plan other than in connection with the exercise of portability rights by such individual, or, where applicable, a beneficiary of such individual, but does not include Quebec retirees.
- d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date, but does not include Quebec deferred members of the Transferring Plan.
- e. Effective Date means April 1, 2021.
- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Inactive Accrued Transferring Plan Benefits does not include any benefit or portion thereof that has been annuitized or otherwise settled under the Transferring Plan as of the Transfer Date.
- g. Inactive Transferring Plan Member means any individual whose pension benefit is not subject to Quebec pension legislation and is either an Annuitized Transferring Plan Member, a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Offset Commencement Date** means the date determined by the Transferring Employer and the Plan Manager on which the contribution rate set out in Section 4.b. of this Appendix P shall become effective.
- i. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- j. Transfer Date means the date determined by the Transferring Employers and the Plan Manager following receipt of all necessary Regulatory Approvals on which the defined benefit assets of the Transferring Plan are to be transferred from the

Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.

- k. **Transferring Employers** has the meaning set forth in the introductory paragraph of this Appendix P.
- Transferring Plan has the meaning set forth in the introductory paragraph of
 this Appendix P and any incorporation by reference into this Appendix P of the
 terms of the Transferring Plan shall mean the terms of the Transferring Plan (and
 any amendments thereto) in effect immediately prior to the effective date of the
 asset transfer. The Transferring Plan text (and any amendments thereto) are filed
 by the Plan pursuant to the Act as documents supporting the Plan.
- m. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Annuitized Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix P that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix P; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- On and after the Effective Date, any Active Transferring Plan Member shall be required
 to contribute in respect of their participation under this Appendix P of the Plan based on
 the definition of Annual Earnings under Appendix G.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Transferring Employers and Active Transferring Plan Members shall contribute on Active Transferring Plan Members' Annual Earnings as set out in Section 3 of this Appendix P in accordance with the following rate(s):

Contribution Rate Schedule	Transferring Employers Contribution Rate	Active Transferring Plan Member Contribution Rate
For the period commencing on April 1, 2021	5%	5%

- b. Notwithstanding the above, as of the Offset Commencement Date, the
 Transferring Employers and Active Transferring Plan Members shall contribute
 1% of Annual Earnings to the Plan for a period of 53 months, but such Active
 Transferring Plan Members shall accrue benefits under the Plan based on the combined Active Transferring Plan Member Contribution Rate and the
 Transferring Employer Contribution Rate set out in Section 4.a. of this Appendix P.
- c. After the 53-month period set out in Section 4.b. of this Appendix P, the Transferring Employers and Active Transferring Plan Members shall resume making contributions to the Plan in accordance with Section 4.a. of this Appendix P.
- 5. All Active Transferring Plan Members who, as of the Effective Date, were on income replacement pursuant to a long-term disability policy and were accruing a benefit under the terms of the Transferring Plan shall contribute to the Plan and the applicable Employer shall make matching contributions for so long as such Active Transferring Plan Member continues to qualify for long-term disability benefits. The Annual Earnings in respect of such member on which such contributions are based shall correspond to the earnings used to determine the benefit accruing to such member under the Transferring Plan immediately before the Effective Date. For greater certainty, during the 53-month period after the Offset Commencement Date such Active Transferring Plan Member shall contribute to the Plan in accordance with the contribution rate set out in Section 4.b of this Appendix P, but shall accrue benefits under the Plan based on the combined Active Transferring Plan Member Contribution Rate and the Transferring Employer Contribution Rate set out in Section 4.a. of this Appendix P.

Accrued Transferring Plan Benefits

- 6. As of the Effective Date, and subject to the provisions of this Appendix P, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix P, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 7. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix P or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section(s) 6 and 8 of this Appendix P, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) 6 and 8 of this Appendix P.
- 8. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 9. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 10. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 7.a. of this Appendix P; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 7.b. of this Appendix P.

- 11. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix P, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2022.
- 12. Notwithstanding Section 11 of this Appendix P, in no event shall the cumulative amount of indexing, if any, provided to Active Transferring Plan Members pursuant to Section 11 of this Appendix P be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the Transferring Plan in effect immediately prior to the Effective Date.

Inactive Transferring Plan Members

- 13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix P, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix P. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 14. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 15. Any pension payable under Section 13 of this Appendix P, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to January 1, 2022.
- 16. Notwithstanding Section 15 of this Appendix P, in no event shall the cumulative amount of indexing, if any, provided to Inactive Transferring Plan Members pursuant to Section 15 of this Appendix P be less than the cumulative amount of indexing, if any, that would be provided for under the provisions of the Transferring Plan in effect immediately prior to the Effective Date.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 6-16 of this Appendix P prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - g. in respect of any Active Transferring Plan Member:

- i. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
- ii. under Sections 6, 7.b., and 8 of this Appendix P in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
- h. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or

in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX Q

SANOFI PASTEUR LIMITED

Effective April 1, 2021, Sanofi Pasteur Limited (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the defined benefit assets and liabilities of The Sanofi Pasteur Limited Pension Plan for Bargaining Employees (the "Transferring Plan") (Registration Number 0491316) are to be transferred to the Plan effective April 1, 2021.

This Appendix Q shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of defined benefit assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix Q, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor.
 - d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
 - e. Effective Date means April 1, 2021.
 - f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to

the maximum pension benefit provisions of the Income Tax Act (Canada).

- g. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. Offset Commencement Date means the later of January 1, 2023 and the first day of the month after the Transfer Date, or such other date within three months immediately following the Transfer Date, to be determined by the Transferring Employer and the Plan Manager.
- i. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- j. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the defined benefit assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- Transferring Employer has the meaning set forth in the introductory paragraph of this Appendix Q.
- Transferring Plan has the meaning set forth in the introductory paragraph of
 this Appendix Q and any incorporation by reference into this Appendix Q of the
 terms of the Transferring Plan shall mean the terms of the Transferring Plan (and
 any amendments thereto) in effect immediately prior to the effective date of the
 asset transfer. The Transferring Plan text (and any amendments thereto) are filed
 by the Plan pursuant to the Act as documents supporting the Plan.
- m. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix Q that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix Q; and

have any service credited to the Active Transferring Plan Member under the Transferring Plan, and, where applicable, under The Sanofi Pasteur Limited Pension Plan for Salaried Employees, for the purposes of determining eligibility or entitlement to benefits recognized under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.

- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix Q, based on the definition of Earnings under the Transferring Plan in which such Active Transferring Plan Member was participating immediately before the Effective Date.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. The Transferring Employer and Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix Q in accordance with the following rate(s):

Contribution Rate Schedule	Transferring Employer Contribution Rate	Active Transferring Plan Member Contribution Rate
From April 1, 2021to December 31, 2021	15%	3%
From January 1, 2022 to December 31, 2022	14%	4%
On and after January 1, 2022	13%	5%

and

Notwithstanding the above, as of the Offset Commencement Date, the
 Transferring Employer and Active Transferring Plan Members shall contribute to
 the Plan in accordance with the table below:

Months following the Offset Commencement Date	Transferring Employer Contribution Rate	Active Transferring Plan Member Contribution Rate
First 84 months	10%	2%
Months 85-96	11%	3%
Months 97-99	12%	4%

- c. For the avoidance of doubt, Active Transferring Plan Members shall contribute in accordance with the Active Transferring Plan Member Contribution Rate column above, and the Transferring Employer shall contribute in accordance with the Transferring Employer Contribution Rate column above, but for this 99-month period such Active Transferring Plan Members shall accrue benefits under the Plan as if both Active Transferring Plan Members and the Transferring Employer were contributing to the Plan based on the applicable contribution rate set out in Section 4.a. of this Appendix Q.
- d. After the 99-month period set out in Section 4.b. of this Appendix Q, the Transferring Employer and Active Transferring Plan Members shall resume making contributions to the Plan in accordance Section 4.a. of this Appendix Q.
- 5. Unless otherwise agreed to pursuant to a collective bargaining agreement applicable to Active Transferring Plan Members, all Active Transferring Plan Members who, on or after the Effective Date, were on income replacement pursuant to a long-term disability benefit shall not be required to contribute to the Plan, but the Employer shall make Employer and Active Transferring Plan Member contributions in accordance with the applicable table above for so long as such member continues to qualify for long-term disability under the Employer's disability policy and the Earnings in respect of such member on which such contributions are based shall correspond to the earnings used to determine the benefit accruing to such member under the Transferring Plan immediately before the Effective Date. For greater certainty, during the 99-month period after the Offset Commencement Date the Transferring Employer shall contribute to the Plan in accordance with the contribution rates set out in Section 4.b of this Appendix Q, but the Active Transferring Plan Member shall accrue benefits under the Plan based on the contribution rates set out in Section 4.a. of this Appendix Q.

- 6. Any Active Transferring Plan Member in receipt of benefits from the Ontario Workplace Safety Insurance Board or from the Employer's workplace injury policy:
 - a. Shall not contribute to the Plan for the first 12 consecutive months during which they are in receipt of such benefits, but shall continue to accrue a pension based on the Active Transferring Plan Member's Earnings immediately prior to commencing receipt of such benefits; and
 - b. After such 12 month period, shall make Active Transferring Plan Member contributions to the Plan, with the Transferring Employer making corresponding Transferring Employer contributions, based on the Active Transferring Plan Member's Earnings immediately prior to commencing receipt of benefits from the Ontario Workplace Safety Insurance Board or the Employer's workplace injury policy.

For greater certainty, during the 99-month period after the Offset Commencement Date the Employer and such Active Transferring Plan Member shall contribute to the Plan in accordance with the contribution rate set out in Section 4.b of this Appendix Q, but shall accrue benefits under the Plan based on the combined Active Transferring Plan Member Contribution Rate and the Transferring Employer Contribution Rate set out in Section 4.a. of this Appendix Q.

Accrued Transferring Plan Benefits

- 7. As of the Effective Date, and subject to the provisions of this Appendix Q, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix Q, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 8. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix Q or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section(s) 7 and 9 of this Appendix Q, plus

- b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) 7 and 9 of this Appendix Q.
- 9. For the purposes of this Appendix Q, the Active Accrued Transferring Plan Benefit shall be enhanced on and after January 1, 2022 each January 1 thereafter, prior to his or her Termination of Membership Date by the Adjustment Factor.
- 10. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 11. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 12. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 8.a. of this Appendix Q; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 8.b. of this Appendix Q.
- 13. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix Q, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2022.

Inactive Transferring Plan Members

14. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix Q, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix Q. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 15. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 16. Any pension payable under Section 14 of this Appendix Q, once in payment after the Effective Date, shall increase pursuant to Article 11 of Appendix G of the Plan provided that no such increase shall be made in respect of the period prior to and including January 1, 2022.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 7-16 of this Appendix Q prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - i. in respect of any Active Transferring Plan Member:
 - to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - ii. under Sections 7, 8.b. and 9 of this Appendix Q in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - j. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or

in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX R

THE GEORGE HULL CENTRE FOR CHILDREN AND FAMILIES

Effective July 1, 2021, The George Hull Centre for Children and Families (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Pension Plan for Employees of The George Hull Centre for Children and Families (the "Transferring Plan") (Registration Number 0929547) are to be transferred to the Plan effective July 1, 2021.

This Appendix R shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member.

- 1. The following words and phrases shall, for the purposes of this Appendix R, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.

- d. Effective Date means July 1, 2021.
- e. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- f. **Inactive Transferring Plan Member** means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- g. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- h. **Transfer Date** means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- i. **Transferring Employer** has the meaning set forth in the introductory paragraph of this Appendix R.
- j. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix R and any incorporation by reference into this Appendix R of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- k. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix R that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix R; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix R of the Plan based on the definition of Annual Earnings under Appendix G.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3of this Appendix R in accordance with the following rate:

Effective	Contribution Rate	
For the period		
commencing on and		
after July 1, 2021	7%	

and

b. The Transferring Employer shall match contributions in respect of such Active Transferring Plan Members.

Accrued Transferring Plan Benefits

5. As of the Effective Date, and subject to the provisions of this Appendix R, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix R, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.

- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix R or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section 5 of this Appendix R, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section 5 of this Appendix R.
- 7. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 8. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 9. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 6a of this Appendix R; and

- b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6b of this Appendix R.
- 10. No indexation will be applied to the Active Accrued Transferring Plan Benefit upon commencement of a pension by an Active Transferring Plan Member other than any indexation applicable under the terms of the Transferring Plan, and, for the avoidance of doubt, any pension benefit accrued under Section 2 of this Appendix R will be subject to the indexation provisions under Article 11 of Appendix G.

Inactive Transferring Plan Members

- 11. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including, for the avoidance of doubt, monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix R, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix R. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 12. Any portability options in respect of an Inactive Accrued Transferring Plan
 Benefit credited to an Inactive Transferring Plan Member under the Plan shall be
 in accordance with the terms of the Plan and not the Transferring Plan.
- 13. No indexation will be applied to pensions payable under Section 10 of this Appendix R once in payment the Effective Date other than any indexation applicable under the terms of the Transferring Plan.

Transition

- 14. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5-12 of this Appendix R prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - k. in respect of any Active Transferring Plan Member:
 - i. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or

- ii. under Sections 5 and 6b of this Appendix R in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
- 1. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or

in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX S

THE GLOBE AND MAIL INC.

Effective May 1, 2021, The Globe and Mail Inc. (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of The Globe and Mail Employees' Retirement Plan (the "Transferring Plan") (Registration Number 1075704) are to be transferred to the Plan effective August 1, 2021, subject to applicable regulatory approval.

This Appendix S shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, subject to applicable regulatory approval, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- 1. The following words and phrases shall, for the purposes of this Appendix S, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means an annual adjustment, effective January 1, 2023 and annually thereafter through to termination, death or retirement, to Active Transferring Plan Benefits to reflect the most recent July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor. Notwithstanding the foregoing, the Adjustment Factor shall not exceed 1%. If in any year the increase is less

than 1%, the difference between the increase and 1% will be carried forward to the next year or any subsequent year when it will be permitted under applicable laws.

- d. **Deferred Transferring Plan Member** means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
- e. Effective Date means August 1, 2021.
- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Inactive Accrued Transferring Plan Benefits does not include any benefit or portion thereof that has been annuitized or otherwise settled under the Transferring Plan as of the Transfer Date.
- g. **Inactive Transferring Plan Member** means any individual whose pension benefit is not subject to Quebec pension standards legislation and is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. Participation Date means May 1, 2021.
- i. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- j. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- k. **Transferring Employer** has the meaning set forth in the introductory paragraph of this Appendix S.
- Transferring Plan has the meaning set forth in the introductory paragraph
 of this Appendix S and any incorporation by reference into this Appendix
 S of the terms of the Transferring Plan shall mean the terms of the
 Transferring Plan (and any amendments thereto) in effect immediately
 prior to the effective date of the asset transfer. The Transferring Plan text

(and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.

m. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.

Any capitalized terms used in this Appendix S that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. Each Active Transferring Plan Member shall:
 - a. On and after the Participation Date, participate in and accrue benefits under the Plan for service on and after the Participation Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix S; and
 - b. On and after the Participation Date, have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Participation Date.
- 3. On and after the Participation Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix S of the Plan based on the definition of Earnings under the defined contribution component of the Transferring Plan in which such Active Transferring Plan Member was participating immediately before the Effective Date.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix S in accordance with the following rate(s):

Contribution Rate Schedule	Transferring Employer Contribution Rate	Active Transferring Plan Member Contribution Rate
From May 1, 2021 to April 30, 2022	6%	4%
On and after May 1, 2022	5%	5%

- b. Notwithstanding the above, from the Transfer Date until a date that is 42 months after the Transfer Date, the Transferring Employer shall contribute at a rate of 1%, but for this 42-month period Active Transferring Plan Members shall accrue benefits under the Plan as if both Active Transferring Plan Members and the Transferring Employer were contributing to the Plan based on the applicable contribution rate set out in Section 4.a. of this Appendix S.
- c. After the 42-month period set out in Section 4.b. of this Appendix S, the Transferring Employer shall resume making contributions to the Plan in accordance Section 4.a. of this Appendix S.

Accrued Transferring Plan Benefits

- 5. As of the Effective Date, and subject to the provisions of this Appendix S, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member their Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix S, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 6. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Participation Date pursuant to Section 2 of this Appendix S or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to

- the Member pursuant to Section(s) 5 and 7 of this Appendix S, plus
- b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) 5 and 7 of this Appendix S.
- 7. For the purposes of this Appendix S, the Active Accrued Transferring Plan Benefit shall be enhanced by the Adjustment Factor.
- 8. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 9. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 10. On the death of any Active Transferring Plan Member between the Participation Date and the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a. of this Appendix S.
- 11. On the death of any Active Transferring Plan Member prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. Should the member die on or after the Participation Date, be in accordance with the terms of the Plan in respect of the benefit determined under Section 6.a. of this Appendix S; and
 - b. Should the member die on or after the Effective Date, be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b. of this Appendix S.
- 12. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix S, following a Normal Retirement Date, Early Retirement Date or Postponed

Retirement Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2031, provided that no such increase shall be made in respect of the period prior to January 1, 2023. After January 1, 2031, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix S, shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan.

Inactive Transferring Plan Members

- 13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix S, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix S. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 14. Any portability options in respect of an Inactive Accrued Transferring Plan
 Benefit credited to an Inactive Transferring Plan Member under the Plan shall be
 in accordance with the terms of the Plan and not the Transferring Plan.
- 15. On the commencement of pension by a Deferred Transferring Plan Member, the Inactive Accrued Transferring Plan Benefit, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2031, provided that no such increase shall be made in respect of the period prior to January 1, 2023. After January 1, 2031, the Inactive Accrued Transferring Plan Benefit shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan.
- 16. Any pension in pay to a Retired Transferring Plan Member, once in payment after the Effective Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, provided that no such increase shall be made in respect of the period prior to January 1, 2023. Notwithstanding the foregoing, if at any point a Retired Transferring Plan Member's pension or a portion thereof is ineligible to receive indexation increases under the Transferring Plan's provisions, such pension or portion thereof shall increase pursuant increase every January 1 pursuant to Article 11 of Appendix G of the Plan.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 5 to 16 of this Appendix S prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - m. in respect of any Active Transferring Plan Member:
 - i. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - ii. under Sections 5, 6.b and 7 of this Appendix S in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - n. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX T

MC LEGALMANAGEMENT LIMITED PARTNERSHIP

Effective March 28, 2022, MC LegalManagement Limited Partnership (the "Transferring Employer") became a participating employer in the Plan for the purposes of Appendix A.1 under an agreement whereby the assets and liabilities of the Pension Plan for the Non-Legal Staff of MC LegalManagement Limited Partnership (the "Transferring Plan") (Registration No. 0378000) are to be transferred to the Plan effective March 28, 2022.

This Appendix T shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- 1. The following words and phrases shall, for the purposes of this Appendix T, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the defined benefit provisions of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.
 - c. Adjustment Factor means an annual adjustment to reflect the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor. Notwithstanding the foregoing, the adjustment to such Active Accrued Transferring Plan Benefits shall not exceed 2% each calendar year.

- d. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
- e. Effective Date means March 28, 2022.
- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the defined benefit provisions of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
- g. **Inactive Transferring Plan Member** means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary.
- h. **Retired Transferring Plan Member** means any person in receipt of a pension from the Transferring Plan in respect of the defined benefit provisions as at the Effective Date.
- i. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- j. **Transferring Employer** has the meaning set forth in the introductory paragraph of this Appendix T.
- k. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix T and any incorporation by reference into this Appendix T of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a
 payment from or otherwise has an entitlement under the Transferring Plan
 in respect of the defined benefit provisions other than Active Transferring
 Plan Members, Retired Transferring Plan Members or Deferred
 Transferring Plan Members.

Any capitalized terms used in this Appendix T that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix T; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix T of the Plan based on the definition of Annual Earnings under Appendix G.
- 4. The contribution rate applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix T in accordance with the following rates:

Effective	Contribution Rate
For the period	
commencing on the	
Effective Date:	5%

and

- b. The Transferring Employer shall match contributions in respect of such Active Transferring Plan Members.
- 5. Any Active Transferring Plan Member in receipt of benefits from the Transferring Employer's long-term disability plan or from an applicable workers'

compensation board that was contributing to and accruing a benefit under the Transferring Plan immediately prior to the Effective Date shall contribute to the Plan at a rate of 5% of their Earnings, as defined under the Transferring Plan, immediately prior to their period of disability, with the Transferring Employer matching such contributions.

Accrued Transferring Plan Benefits

- 6. As of the Effective Date, and subject to the provisions of this Appendix T, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member, his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix T, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 7. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix T or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Sections 6 and 8 of this Appendix T, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Sections 6 and 8 of this Appendix T.
- 8. For the purposes of this Appendix T, the Active Accrued Transferring Plan Benefit shall be enhanced on and after January 1, 2024 each January 1 thereafter, prior to retirement, death or their Termination of Employment Date by the Adjustment Factor.
- 9. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in

accordance with the terms of the Plan and not the Transferring Plan.

- 10. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 11. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 7.a. of this Appendix T; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 7.b. of this Appendix T.
- 12. No indexation will be applied to the Active Accrued Transferring Plan Benefit upon commencement of a pension by an Active Transferring Plan Member other than any indexation applicable under the terms of the Transferring Plan (including any optional forms of pension available to the member on their retirement under the terms of the Transferring Plan), and, for the avoidance of doubt, any pension benefit accrued under Section 2 of this Appendix T will be subject to the indexation provisions under Article 11 of Appendix G.

Inactive Transferring Plan Members

13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix T, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix T. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 14. Any portability options in respect of an Inactive Accrued Transferring Plan
 Benefit credited to an Inactive Transferring Plan Member under the Plan shall be
 in accordance with the terms of the Plan and not the Transferring Plan.
- 15. No indexation will be applied to pensions payable under Section 13 of this Appendix T once in payment on and after the Effective Date other than any indexation applicable under the terms of the Transferring Plan (including any optional forms of pension available to or elected by the member on their retirement under the terms of the Transferring Plan).

Transition

- 16. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 6 to 15 of this Appendix T prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - iii. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - iv. under Sections 6, 7.b. and 8 of this Appendix T in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:
 - v. to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - vi. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX U

THE COLLEGE OF FAMILY PHYSICIANS OF CANADA

Effective August 1, 2022, The College of Family Physicians of Canada and the Nova Scotia College of Family Physicians (the "Transferring Employers") became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Pension Plan for Employees of The College of Family Physicians of Canada (the "Transferring Plan") (Registration Number 0236000) are to be transferred to the Plan effective August 1, 2022.

This Appendix U shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employers in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employers on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix U, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the defined benefit pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, the Active Accrued Transferring Plan Benefit shall include any Active Transferring Plan Member's Hybrid Benefit that has been converted into a defined benefit entitlement.
 - b. Active Transferring Plan Member means any individual that is an employee of any Transferring Employer who participates in the defined benefit component of the Transferring Plan or who is participating in defined contribution component of the Transferring Plan but has a defined benefit entitlement under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member. For greater certainty, an Active Transferring Plan Member does not include those members of the Transferring Plan whose only entitlement under the

Transferring Plan is under the defined contribution component of the Transferring Plan.

- c. Adjustment Factor means, in respect of any Active Transferring Plan Member who was designated a "Class C Member" under the terms of the Transferring Plan, the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor. For greater certainty, the Adjustment Factor will only apply to that part of an Active Accrued Transferring Plan Benefit that was determined based on an Active Accrued Transferring Plan Member's best average earnings.
- d. **Class A Member** means an Active Transferring Plan Member who is a Class A Member under the Transferring Plan.
- e. **Class C Member** means an Active Transferring Plan Member who is a Class C Member under the Transferring Plan.
- f. **Deferred Transferring Plan Member** means any person entitled to a defined benefit or Hybrid Benefit deferred pension or any lump sum from the Transferring Plan as at the Effective Date. For greater certainty, a Deferred Transferring Plan Member does not include those members of the Transferring Plan whose only entitlement under the Transferring Plan is under the defined contribution component of the Transferring Plan.
- g. Effective Date means August 1, 2022.
- h. **Hybrid Benefit** means benefits accrued under the Transferring Plan by Class A Members or Class C Members between January 1, 1993, and December 31, 2012. Such Hybrid Benefit will be determined in accordance with the terms of the Transferring Plan as a defined benefit entitlement as of the Effective Date and frozen as of the Effective Date.
- i. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada), and includes any Inactive Transferring Plan Member's Hybrid Benefit, which has been converted into a defined benefit entitlement. For greater certainty, Inactive Accrued Transferring Plan Benefits do not include those pension benefits accrued by or credited to those Inactive Transferring Plan Members whose benefits are subject to

Quebec pension standards legislation.

- j. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary. For greater certainty, Inactive Accrued Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- k. Retired Transferring Plan Member means any person in receipt of a pension from the Transferring Plan as at the Effective Date. For greater certainty, Retired Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- Transfer Date means the date determined by the Transferring Employers and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- m. **Transferring Employers** has the meaning set forth in the introductory paragraph of this Appendix U.
- n. **Transferring Plan** has the meaning set forth in the introductory paragraph of this Appendix U and any incorporation by reference into this Appendix U of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- o. **Transferring Plan Beneficiary** means any person who is entitled to a payment from or otherwise has a defined benefit or Hybrid Benefit entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members. For greater certainty, Transferring Plan Beneficiary does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.

Any capitalized terms used in this Appendix U that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix U; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix U of the Plan based on the definition of Annual Earnings under Appendix G.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Members shall contribute on their earnings as set out in Section 3 of this Appendix U in accordance with the following rate(s):

Member Class	Effective	Contribution Rate
Active Transferring Plan Members		
participating as Class A or Class C	For the period	
members of the Transferring Plan	commencing on August	9%
immediately prior to the Effective	1, 2022	
Date		
Active Transferring Plan Members		
participating in the defined	For the period	
contribution component of the	commencing on August	5%
Transferring Plan immediately	1, 2022	
prior to the Effective Date		

- b. The Transferring Employers shall match contributions in respect of such Active Transferring Plan Members.
- 5. Any Active Transferring Plan Members who, as of the Effective Date, were receiving benefits pursuant to a long-term disability policy of their Transferring Employer and were contributing to the Transferring Plan, shall have the option to contribute to the Plan. If such Active Transferring Plan Members opts to contribute to the Plan, they will receive a corresponding Transferring Employer contribution. Such contributions shall be based on the Active Transferring Plan Member's Earnings immediately prior to commencing receipt of such benefits.

Accrued Transferring Plan Benefits

- 6. As of the Effective Date, and subject to the provisions of this Appendix U, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix U, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 7. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix U or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section(s) 6 and 8 of this Appendix U, plus

- b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) Section 6 and 8 of this Appendix U.
- 8. For the purposes of this Appendix U, the Active Accrued Transferring Plan Benefit shall be enhanced on and after January 1, 2024 each January 1 thereafter, prior to the later of their Termination of Employment Date, retirement or death, by the Adjustment Factor.
- 9. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 10. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 11. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 7.a. of this Appendix U; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 7.b. of this Appendix U.
- 12. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix U, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2039, provided that no such increase shall be made in respect of the period prior to January 1, 2024. After January 1, 2039, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix U, shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan.

Inactive Transferring Plan Members

- 13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix U, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix U. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 14. Any portability options in respect of an Inactive Accrued Transferring Plan
 Benefit credited to an Inactive Transferring Plan Member under the Plan shall be
 in accordance with the terms of the Plan and not the Transferring Plan.
- 15. On the commencement of pension by a Deferred Transferring Plan Member, the Inactive Accrued Transferring Plan Benefit, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2039, provided that no such increase shall be made in respect of the period prior to January 1, 2024. After January 1, 2039, the Inactive Accrued Transferring Plan Benefit shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan.
- 16. Any pension in pay to a Retired Transferring Plan Member or a Transferring Plan Beneficiary, once in payment after the Effective Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2039, provided that no such increase shall be made in respect of the period prior to January 1, 2024. After January 1, 2039, the Inactive Accrued Transferring Plan Benefit shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan. Notwithstanding the foregoing, if a Retired Transferring Plan Member or Transferring Plan Beneficiary's pension or a portion thereof is entitled to receive contractual indexation under Transferring Plan on and after January 1, 2039, such Inactive Transferring Plan Benefit shall not be eligible to receive indexation increases pursuant to Article 11 of Appendix G of the Plan after January 1, 2039.

Transition

- 17. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 6 to 16 of this Appendix U prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - vii. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - viii. under Sections 6., 7.b. and 8 of this Appendix U in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:
 - ix. to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - x. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX V

ANGUS CONSULTING MANAGEMENT LIMITED

Effective September 1, 2022, Angus Consulting Management Limited and ACML Management Western Limited (the "Transferring Employers") became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Pension Plan of Angus Consulting Management Limited (the "Transferring Plan") (Registration Number 1217462) are to be transferred to the Plan effective September 1, 2022.

This Appendix V shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employers in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employers on and after the Effective Date who was not an Active Transferring Plan Member:

- 1. The following words and phrases shall, for the purposes of this Appendix V, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Active Accrued Transferring Plan Benefits do not include benefits in respect of members of the Transferring Plan subject to Quebec pension standards legislation.
 - b. Active Transferring Plan Member means any individual that is an employee of any Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member. For greater certainty, Active Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.

- c. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date. For greater certainty, Deferred Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- d. Effective Date means September 1, 2022.
- e. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Inactive Accrued Transferring Plan Benefits do not include those pension benefits accrued by or credited to those beneficiaries, deferred members or retired members, whose pension benefits are subject to Quebec pension standards legislation.
- f. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary. For greater certainty, Inactive Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- g. Retired Transferring Plan Member means any person in receipt of a pension from the Transferring Plan as at the Effective Date. For greater certainty, Retired Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- h. **Transfer Date** means the date determined by the Transferring Employers and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- i. **Transferring Employers** has the meaning set forth in the introductory paragraph of this Appendix V.
- j. Transferring Plan ERD has the meaning set forth in Section 6 of this Appendix V.

- k. **Transferring Plan** has the meaning set forth in the introductory paragraph of this Appendix V and any incorporation by reference into this Appendix V of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- Transferring Plan Beneficiary means any person who is entitled to a
 payment from or otherwise has an entitlement under the Transferring Plan
 other than Active Transferring Plan Members, Retired Transferring Plan
 Members or Deferred Transferring Plan Members. For greater certainty,
 Transferring Plan Beneficiary does not include any individual whose
 benefits under the Transferring Plan are subject to Quebec pension
 standards legislation.

Any capitalized terms used in this Appendix V that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

- 2. On and after the Effective Date, each Active Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix V; and
 - b. have any service credited to the Active Transferring Plan Member under the Transferring Plan recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix V, based on the definition of Pensionable Earnings under the Transferring Plan.
- 4. The contribution rate(s) applicable in respect of Active Transferring Plan Members is as set out below:
 - a. Active Transferring Plan Member and Transferring Employer contributions for Active Transferring Plan Members that commenced

participation in the Transferring Plan on or prior to May 30, 2022, shall be made as a percentage of their earnings as set out in Section 3 of this Appendix V in accordance with the following rate:

Effective	Active Transferring Plan Member Contribution Rate	Transferring Employer Contribution Rate
From September 1, 2022	6%	8%
to August 31, 2023:		
On and after September	6%	6%
1, 2023	0%	5 /6

b. Active Transferring Plan Member and Transferring Employer contributions for Active Transferring Plan Members that commenced participation in the Transferring Plan after May 30, 2022, shall be made as a percentage of their earnings as set out in Section 3 of this Appendix V in accordance with the following rate:

Effective	Active Transferring Plan Member Contribution Rate	Transferring Employer Contribution Rate
On and after September 1, 2022:	6%	6%

5. Any Active Transferring Plan Member who, as of the Effective Date, was receiving benefits pursuant to a long-term disability policy of their Transferring Employer or receiving benefits from an applicable workers' compensation board shall not be required to contribute to the Plan, but the applicable Transferring Employer shall make Transferring Employer and Active Transferring Plan Member contributions in accordance with Section 4 of this Appendix V for so long as required under the terms of the Transferring Plan. Such contributions shall be based on the Active Transferring Plan Member's earnings immediately prior to commencing receipt of such benefits.

- 6. Notwithstanding the foregoing, any Active Transferring Plan Member with an early retirement date under the Transferring Plan that is prior to age 50 ("Transferring Plan ERD") as at the Effective Date shall be eligible to retire in respect of the benefit they accrued under Section 2.a. of this Appendix V. If such Active Transferring Plan Member retires prior to age 50:
 - a. their Active Accrued Transferring Plan Benefit shall be adjusted in accordance with the terms of the terms of the Transferring Plan; and
 - b. the benefit they accrued under Section 2.a. of this Appendix V shall be reduced in accordance with Section 7.02 and 7.03 of Appendix G to the Plan between age 50 and 65, with a further actuarial reduction from age 50 between the Active Transferring Plan Member's Transferring Plan ERD and age 50.

Accrued Transferring Plan Benefits

- 7. As of the Effective Date, and subject to the provisions of this Appendix V, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix V, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.
- 8. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix V or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section 7 of this Appendix V, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan

pursuant to Section 7 of this Appendix V.

- 9. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 10. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 11. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 8.a. of this Appendix V; and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 8.b. of this Appendix V.
- 12. On the commencement of pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix V, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2040, provided that no such increase shall be made in respect of the period prior to January 1, 2026. After January 1, 2040, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix V, shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan. Notwithstanding the foregoing, the inflation protection factor so determined and applied pursuant to Article 11 of Appendix G shall increase as if such inflation protection factor was determined using 50% of the change in the Consumer Price Index.

Inactive Transferring Plan Members

13. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same

rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix V, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix V. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.

- 14. Any portability options in respect of an Inactive Accrued Transferring Plan
 Benefit credited to an Inactive Transferring Plan Member under the Plan shall be
 in accordance with the terms of the Plan and not the Transferring Plan.
- 15. Any pension payable under Section 13 of this Appendix V, once in payment after the Effective Date, shall increase pursuant to the indexation provisions of the Transferring Plan every January 1, with the last increase applied pursuant to the Transferring Plan's indexation provisions being January 1, 2040, provided that no such increase shall be made in respect of the period prior to January 1, 2026. After January 1, 2040, the Inactive Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix V, shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan. Notwithstanding the foregoing, the inflation protection factor so determined and applied pursuant to Article 11 of Appendix G shall increase as if such inflation protection factor was determined using 50% of the change in the Consumer Price Index.

Transition

- 16. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 7 to 15 of this Appendix V prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - a. in respect of any Active Transferring Plan Member:
 - xi. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - xii. under Sections 7 or 8.b. of this Appendix V in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - b. in respect of any Inactive Transferring Plan Member:

- xiii. to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
- xiv. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

APPENDIX W

HEALTHCARE EXCELLENCE CANADA

Effective October 1, 2022, Healthcare Excellence Canada (the "Transferring Employer") became a participating employer in the Plan for the purposes of Schedule A and Schedule A.1 under an agreement whereby the assets and liabilities of the Pension Plan for Healthcare Excellence Canada (Registration Number 1069277) (the "Transferring Plan") are to be transferred to the Plan effective October 1, 2022.

This Appendix W shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employer in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employer on and after the Effective Date who was not an Active Transferring Plan Member:

- The following words and phrases shall, for the purposes of this Appendix W, have the following meanings respectively, unless a different meaning is plainly required by the context:
 - a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada).
 - b. Active Transferring Plan Member means any individual that is an employee of the Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member.

- c. Adjustment Factor means, in respect of DBplus Transferring Plan Members, an annual adjustment to reflect the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor.
- d. **DBplus Transferring Plan Member** means an Active Transferring Plan Member that either:
 - i. was hired on or after October 1, 2020; or
 - was hired prior to October 1, 2020 and is employed by the Transferring Employer on and other than regular full-time basis.
- e. **DBprime Transferring Plan Member** means an Active Transferring Plan Member that was hired prior to October 1, 2020 and is employed on a regular full-time basis.
- f. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date.
- g. Effective Date means October 1, 2022.
- h. Inactive Accrued Transferring Plan Benefits means the pension
 benefits accrued by or credited to each Inactive Transferring Plan Member
 under the terms of the Transferring Plan immediately prior to the Effective
 Date, subject to the maximum pension benefit provisions of the Income
 Tax Act (Canada).
- i. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan

Member or a Transferring Plan Beneficiary.

- j. Retired Transferring Plan Member means any person in receipt of a pension from the Transferring Plan as at the Effective Date.
- k. Transfer Date means the date determined by the Transferring Employer and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- 1. **Transferring Employer** has the meaning set forth in the introductory paragraph of this Appendix W.
- m. Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix W and any incorporation by reference into this Appendix W of the terms of the Transferring Plan shall mean the terms of the Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.
- n. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members.
- Transition Date means the date a DBprime Transferring Plan Member becomes employed with the Transferring Employer in an other than regular full-time position.

Any capitalized terms used in this Appendix W that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

DBprime Transferring Plan Members

- 2. On and after the Effective Date, each DBprime Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of the Design One Provisions of the Plan, subject to the provisions of this Appendix W;
 - b. have any service credited to the DBprime Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.
- 3. On and after the Effective Date, any DBprime Transferring Plan Member shall be required to contribute to the Design One Provisions of the Plan in accordance with the terms of the Plan and the Transferring Employer shall match such contributions in accordance with the terms of the Plan, as may be amended from time to time.

DBplus Transferring Plan Members

- 4. On and after the Effective Date, each DBplus Transferring Plan Member shall:
 - a. commence participation in and accrue benefits under the Plan for service
 on and after the Effective Date in accordance with the terms of Appendix
 G of the Plan, subject to the provisions of this Appendix W; and
 - b. have any service credited to the DBplus Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or

entitlement to benefits recognized under the Plan, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.

- 5. On and after the Effective Date, any DBplus Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix W of the Plan based on the definition of Annual Earnings under Appendix G.
- 6. The contribution rate(s) applicable in respect of DBplus Transferring Plan Members is as set out below:
 - a. DBplus Transferring Plan Members shall contribute on their earnings as set out in Section 5 of this Appendix W in accordance with the following rate(s):

Effective	Contribution Rate
For the period	
commencing on	
October 1, 2022	9%

and

 The Transferring Employer shall match contributions in respect of such DBplus Transferring Plan Members.

Accrued Transferring Plan Benefits

7. As of the Effective Date, and subject to the provisions of this Appendix W, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her

Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix W, the form of payment, optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.

- 8. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Plan Member under the Plan on and after the Effective Date pursuant to Section 2 or 4 of this Appendix W, as applicable, or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section(s) 7 and 9 or 10, as applicable, of this Appendix W, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) 7 and 9 or 10, as applicable, of this Appendix W.
- 9. For the purposes of this Appendix W, the Active Accrued Transferring Plan Benefit for DBplus Transferring Plan Members shall be enhanced on and after January 1, 2024 each January 1 thereafter, prior to his or her death, retirement or Termination of Employment Date by the Adjustment Factor.
- 10. For greater certainty, the Active Accrued Transferring Plan Benefit for DBprime Transferring Plan Members shall be adjusted to account for future earnings received by the DBprime Transferring Plan Member from the Transferring Employer up to the DBprime Transferring Plan Member's Termination of Employment Date, Early Retirement Date, Normal Retirement Date or Postponed

Retirement Date, as applicable.

- 11. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 12. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 13. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death benefits so payable in respect of such person shall:
 - a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 7, 9 or 10, as applicable, of this Appendix W;
 and
 - b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 7,
 9 or 10, as applicable, of this Appendix W.
- 14. On the commencement of a pension by an Active Transferring Plan Member, the Active Accrued Transferring Plan Benefit, as adjusted pursuant to this Appendix W, following a Normal Retirement Date, Early Retirement Date or Postponed Retirement Date, shall increase every January 1 pursuant to Article 11 of Appendix G of the Plan, or Article 14 of the Plan, as applicable. Notwithstanding the above, such increase between January 1, 2024 and January 1, 2038 shall be

guaranteed and not conditional. Any increase after January 1, 2038 shall be conditional, as set out in Article 11 of Appendix G of the Plan, or Article 14 of the Plan, as applicable.

Inactive Transferring Plan Members

- 15. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix W, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix W. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 16. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 17. No indexation will be applied to pensions payable under Section 15 of this Appendix W once in payment the Effective Date other than any indexation applicable under the terms of the Transferring Plan.

Transition

18. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 7 to 17 of this Appendix W prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:

- a. in respect of any Active Transferring Plan Member:
 - xv. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - xvi. under Sections 7, 8.b, 9 and 10 of this Appendix W in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
- b. in respect of any Inactive Transferring Plan Member:
 - xvii. to the extent to which payment has been made from the

 Transferring Plan prior to the Transfer Date in respect of such
 Inactive Transferring Plan Member; or
 - xviii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.
- 19. In the event that a DBprime Transferring Plan Member ceases to be employed in a full-time position but continues as an employee of the Transferring Employer eligible for membership under the Design Two Provisions, such DBprime Transferring Plan Member shall cease participating in and accruing a benefit under the Design One Provisions of the Plan and commence participating in and accruing a benefit under the Design Two Provisions of the Plan, subject to the provisions of Appendix W, on and after the Transition Date, in accordance with Section 12.03 of Appendix G of the Plan, and in accordance with Section B.(1) of Appendix A.1 of the Plan.
- 20. For the avoidance of doubt, on and after the Transition Date, the total pension benefit entitlement under the Plan in respect of such DBprime Transferring Plan Member described in Section 21 of this Appendix W shall be:
 - a. The pension benefit accrued to the DBprime Transferring Plan Member under the Design One Provisions of the Plan on and after the Effective Date and up to the Transition Date, pursuant to Section 2 of this Appendix

W, plus

- b. The pension benefit accrued to the DBprime Transferring Plan Member under the Design Two Provisions of the Plan on and after the Transition Date, pursuant to Section 20 of this Appendix W, plus
- c. The DBprime Transferring Plan Member's Active Accrued Transferring
 Plan Benefit credited to the DBprime Transferring Plan Member under the
 Plan pursuant to Sections 7, 9 and 10 of this Appendix W.
- 21. For greater certainty, if the DBprime Transferring Plan Member transitions from the Design One Provisions to the Design Two Provisions, the DBprime Transferring Plan Member's Active Accrued Transferring Plan Benefit shall not continue to be adjusted to account for future earnings received by the DBprime Transferring Plan Member from the Transferring Employer on and after the Transition Date. However, on the January 1 next following the first January 1 after the Transition Date and each successive January 1 thereafter, the DBprime Transferring Plan Member's Active Accrued Transferring Plan Benefit shall be enhanced by the Adjustment Factor in accordance with Section 9 of this Appendix W.

APPENDIX X

LUTHERAN CHURCH—CANADA

Effective April 1, 2023, the Transferring Employers, as that term is defined in Exhibit 1 to this Appendix X, became participating employers in the Plan for the purposes of Schedule A.1 under an agreement whereby the assets and liabilities of the Lutheran Church—Canada Pension Plan (Registration No. 0355610) (the "Transferring Plan") are to be transferred to the Plan effective April 1, 2023.

This Appendix X shall apply with effect on and after the Effective Date in respect of the participation by the Transferring Employers in the Plan and the transfer of assets and liabilities from the Transferring Plan to the Plan, but shall not apply in respect of any employee of the Transferring Employers on and after the Effective Date who was not an Active Transferring Plan Member:

32.20. The following words and phrases shall, for the purposes of this Appendix X, have the following meanings respectively, unless a different meaning is plainly required by the context:

- a. Active Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Active Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date, subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Active Accrued Transferring Plan Benefits do not include benefits in respect of members of the Transferring Plan subject to Quebec pension standards legislation.
- b. Active Transferring Plan Member means any individual that is an employee of any Transferring Employer who participates in the Transferring Plan or who otherwise participates in and is accruing benefits under the Transferring Plan immediately prior to the Effective Date and who is not an Inactive Transferring Plan Member. For greater certainty, Active Transferring Plan Member does not include any individual whose benefits under the

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Transferring Plan are subject to Quebec pension standards legislation.

- c. Adjustment Factor means an annual adjustment to reflect:
 - i. for an Active Transferring Plan Member, the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor. Notwithstanding the foregoing, the adjustment to such Active Transferring Plan Benefits shall not exceed 2.3% each calendar year.
 - ii. for a Deferred Transferring Plan Member who was a US Member, or for an Active Transferring Plan Member who, as of the Effective Date, was in receipt of long-term disability benefits and was accruing a pension benefit under the Transferring Plan, the better of the July 1 to June 30 increase in the Average Industrial Wage, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor, up to a maximum of 2.3% each calendar year, and the July 1 to June 30 increase in the Consumer Price Index, if any, as reported by Statistics Canada in respect of the period immediately prior to the effective date of the adjustment factor, up to a maximum of 3% each calendar year.
- d. Deferred Transferring Plan Member means any person entitled to a deferred pension or any lump sum from the Transferring Plan as at the Effective Date. For greater certainty, Deferred Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- e. Effective Date means April 1, 2023.
- f. Inactive Accrued Transferring Plan Benefits means the pension benefits accrued by or credited to each Inactive Transferring Plan Member under the terms of the Transferring Plan immediately prior to the Effective Date,

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Right + Aligned at: 2.38" + Indent at: 2.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

subject to the maximum pension benefit provisions of the Income Tax Act (Canada). For greater certainty, Inactive Accrued Transferring Plan Benefits do not include those pension benefits accrued by or credited to those beneficiaries, deferred members or retired members, whose benefits are subject to Quebec pension standards legislation.

- g. Inactive Transferring Plan Member means any individual who is either a Retired Transferring Plan Member, a Deferred Transferring Plan Member or a Transferring Plan Beneficiary. For greater certainty, Inactive Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- h. **LCC DB Member** means any Active Transferring Plan Member who attained age 55 as at January 1, 2013 and whose age plus Credited Service, as defined under the Plan, as at December 31, 2012 exceeded 80 years.
- i. Retired Transferring Plan Member means any person in receipt of a pension from the Transferring Plan as at the Effective Date. For greater certainty, Retired Transferring Plan Member does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- j. Transfer Date means the date determined by the Transferring Employers and the Plan Manager following receipt of all necessary Regulatory Approvals on which the assets of the Transferring Plan are to be transferred from the Transferring Plan to the Plan, which date shall be no later than the date prescribed by the Framework Regulations.
- **k.** Transferring Employers has the meaning set forth in Exhibit 1 to this Appendix X.
- **I.** Transferring Plan has the meaning set forth in the introductory paragraph of this Appendix X and any incorporation by reference into this Appendix X of the terms of the Transferring Plan shall mean the terms of the

Transferring Plan (and any amendments thereto) in effect immediately prior to the effective date of the asset transfer. The Transferring Plan text (and any amendments thereto) are filed by the Plan pursuant to the Act as documents supporting the Plan.

- m. Transferring Plan Beneficiary means any person who is entitled to a payment from or otherwise has an entitlement under the Transferring Plan other than Active Transferring Plan Members, Retired Transferring Plan Members or Deferred Transferring Plan Members. For greater certainty, Transferring Plan Beneficiary does not include any individual whose benefits under the Transferring Plan are subject to Quebec pension standards legislation.
- n. US Member means any individual who participated in the defined benefit component of the Transferring Plan and who, as of the Effective Date, was employed by an affiliated employer of LCC in the United States and is accruing pension benefits in the Concordia Retirement Plan as defined under the Transferring Plan.

Any capitalized terms used in this Appendix X that are not defined herein, shall have the meaning ascribed to them under Article 2 and Appendix G of the Plan.

Active Transferring Plan Members

33.21. On and after the Effective Date, each Active Transferring Plan Member shall:

- a. commence participation in and accrue benefits under the Plan for service on and after the Effective Date in accordance with the terms of Appendix G of the Plan, subject to the provisions of this Appendix X; and
- b. have any service credited to the Active Transferring Plan Member under the Transferring Plan for the purposes of determining eligibility or entitlement to benefits recognized under the Plan for the purpose of determining eligibility or entitlement to benefits, but not for the accrual of benefits under the Plan in respect of the period on and after the Effective Date.

34.22. On and after the Effective Date, any Active Transferring Plan Member shall be required to contribute in respect of their participation under this Appendix **Formatted:** Indent: Left: 0.69", Hanging: 0.3", Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

X, based on the definition of Earnings under the Transferring Plan in which such Active Transferring Plan Member was participating immediately before the Effective Date.

The contribution rate(s) applicable in respect of Active Transferring Plan

Members is as set out below:

a. Active Transferring Plan Members and Transferring Employers shall contribute on the Active Transferring Plan Member's earnings as set out in Section 3 of this Appendix X in accordance with the following rate(s):

Employee Class	Effective	Transferring Employer	Active Transferring Plan Member Contribution Rate
LCC DB Members	For the period commencing on April 1, 2023:	12%	4%
All other Active Transferring Plan Members	For the period commencing on April 1, 2023:	6%	4%

in receipt of benefits from a Transferring Employer's long-term disability plan or from an applicable workers' compensation board shall not be required to contribute to the Plan, but the Transferring Employer shall continue to make the Transferring Employer Contribution for so long as the Active Transferring Plan Member continues to qualify for long-term disability benefits. Contributions shall be based on the definition of earnings used to determine the benefit accruing to such member under the Transferring Plan immediately prior to the Effective Date.

37.25. Any Active Transferring Plan Member who commences receipt of benefits from a Transferring Employer's long-term disability plan or from an applicable workers' compensation board on or after the Effective Date shall not be required 384

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

to contribute to the Plan, but the Employer shall contribute both the Transferring Employer contribution and the Active Transferring Plan Member contribution for so long as the Active Transferring Plan Member continues to qualify for long-term disability benefits. Active Transferring Plan Member and Transferring Employer contributions shall be based on the disability earnings received by the Active Transferring Plan Member.

38.26. If an Active Transferring Plan Member is absent on a statutory leave under employment standards legislation applicable to the Active Transferring Plan Member, the Active Transferring Plan Member's Participating Employer will continue to contribute Employer Contributions to the Plan based on the Active Transferring Plan Member's Deemed Annual Earnings during such period of statutory leave regardless of whether the Active Transferring Plan Member contributes to the Plan during such leave. The Active Transferring Plan Member may contribute Active Transferring Plan Member Contributions to the Plan in accordance with Section 4.03(2)(a) and 4.03(2)(b) of Appendix G to the Plan. If the Active Transferring Plan Member does not contribute Active Transferring Plan Member Contributions to the Plan in accordance with Section 4.03(2)(a) and 4.03(2)(b) of Appendix G to the Plan, they shall not be eligible to make a payment in accordance with Section 4.03(2)(c) of Appendix G to the Plan. For greater certainty, this Section 7 of Appendix X shall apply to any statutory leave under applicable employment standards legislation, regardless of whether the applicable employment standards legislation requires such Employer Non-Plan Member Employee to be allowed to continue participation in the Plan for the duration of such leave.

Accrued Transferring Plan Benefits

39.27. As of the Effective Date, and subject to the provisions of this Appendix X, the Active Accrued Transferring Plan Benefit of each Active Transferring Plan Member shall be assumed by the Plan and on the attainment of a Termination of Membership Date, Early Retirement Date, Normal Retirement Date or Postponed Retirement Date, as applicable, by an Active Transferring Plan Member his or her Active Accrued Transferring Plan Benefit shall be payable from the Plan subject to, unless otherwise provided for under this Appendix X, the form of payment,

optional form of payment and reductions, as provided for under the terms of the Transferring Plan in which such Active Transferring Plan Member participated immediately prior to the Effective Date.

- 40.28. For the avoidance of doubt, on and after the Effective Date, the total pension benefit entitlement under the Plan in respect of each Active Transferring Plan Member shall be:
 - a. The pension benefit accrued to the Active Transferring Pland Member under the Plan on and after the Effective Date pursuant to Section 2 of this Appendix X or otherwise credited to the person under the terms of the Plan other than any Active Accrued Transferring Plan Benefit credited to the Member pursuant to Section(s) 8 and 10 of this Appendix X, plus
 - b. The Active Transferring Plan Member's Active Accrued Transferring Plan Benefit credited to the Active Transferring Plan Member under the Plan pursuant to Section(s) 8 and 10 of this Appendix X.
- 41.29. For the purposes of this Appendix X, the Active Accrued Transferring Plan Benefit shall be enhanced on and after January 1, 2024 and each January 1 thereafter, prior to his or her Termination of Membership Date by the Adjustment Factor. For greater certainty, the first adjustment shall be prorated from the Effective Date.
- 42.30. Any portability options in respect of an Active Accrued Transferring Plan Benefit credited to an Active Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 43.31. The determination of entitlement to or eligibility for ancillary benefits in respect of an Active Transferring Plan Member's Active Accrued Transferring Plan Benefit shall take into account any period of service under the Plan after the Effective Date and any period of service that was credited to the Active Transferring Plan Member under the Transferring Plan immediately prior to the Effective Date.
- 44.32. On the death of any Active Transferring Plan Member on or after the Effective Date and prior to commencing a pension under the Plan, the death

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

benefits so payable in respect of such person shall:

- a. be in accordance with the terms of the Plan in respect of the benefit determined under Section 9.a. of this Appendix X;
- b. be in accordance with the terms of the Transferring Plan immediately prior to the Effective Date in respect of the benefit determined under Section 6.b. of this Appendix X.

45.33. No indexation will be applied to the Active Accrued Transferring Plan Benefit upon commencement of a pension by an Active Transferring Plan Member other than any indexation applicable under the terms of the Transferring Plan, and, for the avoidance of doubt, any pension benefit accrued under Section 2 of this Appendix X will be subject to the indexation provisions under Article 11 of Appendix G.

Inactive Transferring Plan Members

- 46.34. As of the Effective Date, the Inactive Accrued Transferring Plan Benefit (including for the avoidance of doubt monthly pensions then in pay at the same rate and on the same terms as under the Transferring Plan) in respect of each Inactive Transferring Plan Member shall be assumed by, and shall be payable from, the Plan in accordance with this Appendix X, and for the avoidance of doubt the provisions of the Transferring Plan in effect as of immediately prior to the Effective Date shall apply to the payment of such pensions, except as provided for under this Appendix X. And for the avoidance of doubt, such provisions of the Transferring Plan shall apply to determine the normal retirement date, earliest unreduced retirement date and early retirement date, pre-retirement death benefits, post-retirement death benefits, and reduction factors and form of pension.
- 47.35. Any portability options in respect of an Inactive Accrued Transferring Plan Benefit credited to an Inactive Transferring Plan Member under the Plan shall be in accordance with the terms of the Plan and not the Transferring Plan.
- 48-36. For the purposes of this Appendix X, the Inactive Accrued Transferring Plan Benefit of a US Member shall be enhanced by the Adjustment Factor on January 1, 2024 and each January 1 thereafter, until such US Member is no employed by an affiliated employer of LCC in the United States or is no longer accruing pension benefits in the Concordia Retirement Plan as defined under the Transferring Plan. For greater certainty, the first adjustment applied to an Inactive Transferring Plan Benefit shall be prorated from the Effective Date.

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 1.75" + Indent at: 2"

Formatted: Indent: Left: 0.25", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 1.25" + Indent at: 1.5"

49.37. No indexation will be applied to pensions payable under Section 15 of this Appendix X once in payment the Effective Date other than any indexation applicable under the terms of the Transferring Plan.

Transition

- 50.38. Notwithstanding the above, the Plan shall have no liability to make payment in respect of Sections 8-18 of this Appendix X prior to the Transfer Date and shall have no liability in respect of the payment of any benefit:
 - o. in respect of any Active Transferring Plan Member:
 - i. to the extent to which payment of such benefit has been made from the Transferring Plan prior to the Transfer Date; or
 - ii. under Sections 8, 9.b., and 10 of this Appendix X in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date; and,
 - p. in respect of any Inactive Transferring Plan Member:
 - to the extent to which payment has been made from the Transferring Plan prior to the Transfer Date in respect of such Inactive Transferring Plan Member; or
 - ii. in respect of whom the Transferring Plan has ceased to have any liability prior to the Transfer Date.

EXHIBIT 1For the purposes of this Appendix X, **Transferring Employers** means:

Emp	bloyer	Business Number	City	Province
	Shepherd of the Valley Lutheran Church	131909681RR0001	Canmore	AB
	Christ The King Evangelical Lutheran Church	106921943RR0001	100 Mile House	BC
	St. John /St.Peter Parish	131157836RR0001	Oxbow	SK
013	Good Shepherd Lutheran Church	140669656RR0001	Amherstburg	ON
	Lutheran Church Of Good Shepherd	133028308RR0001	Barrie	ON
	Redeemer Lutheran Church	107879264RR0001	Brampton	ON
020	Prince Of Peace Lutheran Church	107859167RR0001	Burlington	ON
022	Foothills Lutheran Church	107392516RR0001	Calgary	AB
023	Grace Lutheran Church	118944206RR0001	Calgary	AB
024	Lutheran Church Of The Good Shepherd	118862903RR0001	Calgary	AB
	Prince Of Peace Lutheran Church	107859217RR0001	Rocky View County	AB
027	St. Matthew Lutheran Church	108031428RR0001	Calgary	AB
029	Grace Lutheran Church	131964264RR0001	Camrose	AB
032	Our Saviours Evangelical Lutheran Church	132954769RR0001	Chatham	ON
033	Trinity Lutheran Church	129929154RR0001	Churchbridge	SK
034	St. John Lutheran Church	108023441RR0001	Clifford	ON
035	Christ Our Hope	889385548RR0001	Collingwood	ON
037	Faith Lutheran Church	136511961RR0001	Courtenay	BC
041	Zion Lutheran Church	135463081RR0001	Dashwood	ON
042	St. Paul Lutheran Church	119196079RR0001	Dawson Creek	BC
044	Saviour Lutheran Church	119142016RR0001	Delta	BC
045	Faith Lutheran Church	107373342RR0001	Desboro	ON
047	Redeemer Lutheran Church	129926473RR0001	Didsbury	AB
049	Grace Lutheran Church	118944180RR0001	Drumheller	AB
050	Our Savior Lutheran Church	132738352RR0001	Dryden	ON
052	Trinity Ev Lutheran Church	119270460RR0001	Duncan	BC
053	Faith Lutheran Church		Dunnville	ON
054	St. Paul's & Emmanuel Lutheran Churches	132504267RR0001	Southey	SK
056	All Saints Lutheran Church	106703101RR0001	Edmonton	AB
057	Bethlehem Lutheran Church	118807007RR0001	Edmonton	AB
058	Concordia Lutheran Church	892559469RR0001	Edmonton	AB
059	Concordia Lutheran Seminary	106966575RR0001	Edmonton	AB
	Cross of Christ Lutheran Church of the Deaf	106993843RR0001	Edmonton	AB
	Riverbend Lutheran Church	107897555RR0001	Edmonton	AB
068	St. Lukes/Zion Lutheran Churches	119308641RR0001	Eganville	ON
069	St. Paul Ev Lutheran Church	108036682RR0001	Elmira	ON
070	St. Peters Lutheran Church	119198604RR0001	Estevan	SK
071	Our Savior Lutheran Church	107808362RR0001	Etobicoke	ON
074	Trinity Evangelical Lutheran Church	133327403RR0001	Fisherville	ON

Employer	Business Number	City	Province
075 Concordia Evangelical Lutheran Church	106966526RR0001	Fonthill	ON
076 Church Of The Lutheran Hour	118863109RR0001	Fort Frances	ON
077 Trinity Lutheran Church	119270387RR0001	Fort McMurray	AB
078 Our Saviour Lutheran Church	136105145RR0001	Fort Qu'Appelle	SK
079 Bethany Lutheran Church	129428538RR0001	Fort Saskatchewan	AB
080 Christ Lutheran Church	106921158RR0001	Fort Saskatchewan	AB
081 Trail Peace-Trinity	108135765RR0001	Trail	BC
082 Walnut Grove Lutheran Church	127700508RR0001	Langley	BC
084 Immanuel Lutheran Church	107503625RR0001	Georgetown	ON
087 Trinity Lutheran Church	119270031RR0001	Gowanstown	ON
090 Faith Lutheran Church	118912112RR0001	Grande Prairie	AB
091 Peace Lutheran Church	131434060RR0001	Grenfell	SK
092 Christ Our Savior Lutheran Church	119201523RR0001	Grimsby	ON
093 Pilgrim Lutheran Church	107845133RR0001	Hamilton	ON
095 First St. Matthew Lutheran Church	107387862RR0001	Hanover	ON
097 Good Shepherd Lutheran Church	118941640RR0001	Valleyview	AB
098 St. John's Lutheran Church	106816036RR0001	Humboldt	SK
099 Holy Trinity & Hope Lutheran Church	130399520RR0001	Inglis	MB
100 Redeemer Lutheran Church	107879355RR0001	Kakabeka Falls	ON
101 Lord Of Life Lutheran Church	119024495RR0001	Kamloops	BC
103 Christ Risen Lutheran Church	118858638RR0001	Kanata	ON
104 First Lutheran Church	118918333RR0001	Kelowna	BC
106 Grace Lutheran Church	107444184RR0001	Brandon	MB
107 Southampton Lutheran Church	891456675RR0001	Southampton	ON
108 First Lutheran Church	118918374RR0001	Kingsville	ON
109 Bethel Lutheran Church	131978355RR0001	Kitchener	ON
111 Faith Lutheran Church	107373508RR0001	Kitchener	ON
112 Grace Ev Lutheran Church	129854949RR0001	Kitchener	ON
113 Historic St. Pauls Evangelical Lutheran Church	108036427RR0001	Kitchener	ON
114 Holy Cross Lutheran Church	130563695RR0001	Kitchener	ON
115 Hope Lutheran Church	107490161RR0001	Kitchener	ON
117 Luth Laymens League Of Canada	119026193RR0001	Kitchener	ON
118 La Ronge Lutheran Fellowship	119007201RR0001	La Ronge	SK
120 Christ Lutheran Church	890363393RR0001	Langenburg	SK
121 St. Peter Lutheran Church	130526015RR0001	Leduc	AB
125 First Lutheran Church	130170616RR0001	Lloydminster	AB
126 Faith Lutheran Church	107373359RR0001	London	ON
127 Good Shepherd Lutheran Church	130562853RR0001	London	ON
128 Our Saviour Lutheran Church	133594010RR0001	London	ON
136 St. Paul Lutheran Church	119195667RR0001	Medicine Hat	AB
138 Zion Lutheran Church	108230848RR0001	Melville	SK
139 Faith Lutheran Church	134869759RR0001	Middle Lake	SK

Employer	Business Number	City	Province
140 Christ Our King Luth Church	118858604RR0001	Mississauga	ON
142 Risen Christ Lutheran Church	131804395RR0001	Mississauga	ON
143 St. Marks Lutheran Church	137723730RR0001	Mississauga	ON
146 Grace Ev. Lutheran Church	107444622RR0001	Mitchell	ON
148 First Lutheran Church, Logan	119112753RR0001	Mitchell	ON
152 Emmanuel Lutheran Church	107303224RR0001	Moose Jaw	SK
154 St. John's Evang.Lutheran Ch.	131549339RR0001	Mt Forest	ON
158 St. Paul's Lutheran Church	128379666RR0001	Nanaimo	BC
159 Christ Evangelical Lutheran Church	118858265RR0001	Neepawa	MB
163 Mt. Calvary Lutheran Church	107726945RR0001	New Westminster	BC
164 Our Savior Lutheran Church	123579591RR0001	Niagara Falls	ON
165 Trinity Lutheran Church	891640682RR0001	Niagara-on-the-Lake	ON
166 Zion Lutheran Church	131920423RR0001	Nipawin	SK
171 Grace Evangelical Lutheran Church	131858037RR0001	Oshawa	ON
174 Our Saviour Lutheran Church	119074300RR0001	Ottawa	ON
176 Our Savior Lutheran Church	892635772RR0001	Parksville	BC
177 Prince Of Peace Luth Church	131809329RR0001	Peace River	AB
178 Christ Lutheran Church	106921000RR0001	Pembroke	ON
179 Grace Lutheran Church-Locksley	118943752RR0001	Pembroke	ON
180 St. John's Lutheran Church	108023813RR0001	Pembroke	ON
181 Concordia Lutheran Church and School	106966559RR0001	Penticton	BC
182 Emmanuel Lutheran Church	107303216RR0001	Petersburg	ON
183 Trinity Lutheran Church	132075573RR0001	Pincher Creek	AB
185 Grace Lutheran Church	107444820RR0001	Port Alberni	BC
189 St. Paul's Luth.Church (Port Hope)	129835773RR0001	Cobourg	ON
190 Redeemer Lutheran Church	119112704RR0001	Prince Albert	SK
191 Zion Lutheran Church	119308716RR0001	Prince George	BC
192 Trinity Lutheran Church	108140880RR0001	Quesnel	BC
193 Mt. Calvary Lutheran Church	130375496RR0001	Red Deer	AB
194 Christ Evangelical Lutheran Church	106918162RR0001	Red Lake	ON
196 Good Shepherd Lutheran Church	118941624RR0001	Regina	SK
197 Grace Lutheran Church	107444671RR0001	Regina	SK
198 Mount Olive Lutheran Church	130600216RR0001	Regina	SK
199 New Beginnings Lutheran Church	864218730RR0001	Regina	SK
200 Prince Of Peace Luth. Church	802604942RR0001	Regina	SK
203 Trinity Lutheran Church	108140914RR0001	Richmond	BC
205 Zion Lutheran Church	131819609RR0001	Rimbey	AB
206 St. Matthew Lutheran Church	119190528RR0001	Roblin	MB
208 Christ Lutheran Church	106921034RR0001	Sarnia	ON
210 Grace Lutheran Church, Inc	107444846RR0001	Saskatoon	SK
211 St. Pauls Lutheran Church	108036450RR0001	Saskatoon	SK
213 St. Matthews Lutheran Church	108031394RR0001	Scarborough	ON

Employer	Business Number	City	Province
215 Bethel Lutheran Church	106789100RR0001	Sherwood Park	AB
216 Peace Lutheran Church	107828618RR0001	Sidney	BC
217 St.Peter's Ev. Lutheran Church	108038977RR0001	Simcoe	ON
220 Peace Lutheran Church	119087286RR0001	Spruce Grove	AB
221 Christ Lutheran Church	118858190RR0001	St. Catharines	ON
222 Concordia Luth.Theol. Seminary	118871938RR0001	St. Catharines	ON
223 Grace Lutheran Church	118944065RR0001	St. Catharines	ON
224 Resurrection Lutheran Church	130919830RR0001	St. Catharines	ON
225 Grace LC / Redeemer LC	107879314RR0001	St. Thomas	ON
226 King Of Kings Lutheran Church	131537573RR0001	St. Albert	AB
227 St. John's Lutheran Church	119180842RR0001	Stevensville	ON
229 St. John Lutheran Church	119180990RR0001	Parkland County	AB
231 First St Johns Lutheran Church	133923771RR0001	Stratford	ON
232 St. Peters Lutheran Church	132171505RR0001	Stratford	ON
233 St. Pauls Lutheran Church	132003401RR0001	Tavistock	ON
234 Redeemer Lutheran Church	119112811RR0001	Sudbury	ON
235 St. John's Lutheran Church	853277903RR0001	Summerland	BC
236 Faith Lutheran Church	118912146RR0001	Surrey	BC
238 Mt. Calvary Evangelical Lutheran Church	131163990RR0001	Swift Current	SK
239 Peace Lutheran Church	107828550RR0001	Taber	AB
243 Christ Lutheran Church	118855386RR0001	Thunder Bay	ON
244 Epiphany Lutheran Church	107307043RR0001	Thunder Bay	ON
249 Peace Lutheran Church	131485351RR0001	Tillsonburg	ON
250 Christ Lutheran Church	118858240RR0001	Timmins	ON
254 Toronto Korean Lutheran Church	897622213RR0001	Toronto	ON
255 Trinity Ev Lutheran Church	132350810RR0001	Toronto	ON
256 The Church Of St. Luke	118862002RR0001	North York	ON
257 Trinity Lutheran Church	131523516RR0001	Torrington	AB
259 Bethlehem Lutheran Church	118807015RR0001	Vancouver	BC
261 Killarney Community Lutheran Church	889032462RR0001	Vancouver	BC
262 Trinity Luth Church Of Deaf	119269918RR0001	Vancouver	BC
267 St. Johns Lutheran Church	119181071RR0001	Vernon	BC
268 Hope Ev Lutheran Church	107490237RR0001	Victoria	BC
269 Redeemer Lutheran Church	892213042RR0001	Langford	BC
271 St. John's Lutheran Church	107949471RR0001	Warman	SK
273 Messiah Ev Lutheran Church	119039030RR0001	Waterloo	ON
274 Redeemer Lutheran Church	130331168RR0001	Waterloo	ON
275 First St Paul Luth Church	137406310RR0001	Wellesley	ON
276 St. John's Lutheran Church	119181014RR0001	Wembley	AB
277 Redeemer Lutheran Church	893043844RR0001	West Kelowna	BC
280 Zion Lutheran Church	108230707RR0001	Wetaskiwin	AB
282 Advent Evangelical Lutheran Church	106686603RR0001	Evansburg	AB

Employer	Business Number	City	Province
285 St. John Lutheran Church	125118620RR0001	Williams Lake	BC
287 First Lutheran Church	107386930RR0001	Windsor	ON
288 Gethsemane Lutheran Church	107430704RR0001	Windsor	ON
289 Peace Lutheran Church	107828519RR0001	Windsor	ON
290 Trinity Ev Lutheran Church	108140393RR0001	Winkler	MB
291 Beautiful Savior Lutheran Church	106780745RR0001	Winnipeg	MB
295 Immanuel Lutheran Church	118967447RR0001	Winnipeg	MB
297 Luth Church Of Good Shepherd	132017815RR0001	Winnipeg	MB
298 Luth. Church Of The Redeemer	131634719RR0001	Winnipeg	MB
299 Lutheran Church - Canada	133275479RR0001	Winnipeg	MB
302 Peace Lutheran Church	129584355RR0001	Winnipeg	MB
303 Saint James Lutheran Church	131506008RR0001	Winnipeg	MB
306 Zion Lutheran Church	108230897RR0001	Yorkton	SK
309 Faith Lutheran Church	107373110RR0001	Saskatoon	SK
312 Lutheran Bible Translators of Canada, Inc	107650129RR0001	Kitchener	ON
314 St. Paul's Evangelical Lutheran Church	108036476RR0001	Chilliwack	BC
316 Immanuel Lutheran Church	894356575RR0001	Aldergrove	BC
318 St. Luke Lutheran Church	108027798RR0001	Surrey	BC
319 Grace Lutheran Church	118944222RR0001	Edmonton	AB
320 B.C. Mission Boat Society	866291925RR0001	Courtenay	BC
321 LCC Financial Ministries	888730397RR0001	Winnipeg	MB
329 St. Johns Evangelical Lutheran Church	119181006RR0001	Barrhead	AB
330 Trinity Lutheran Church	129096574RR0001	Brooks	AB
331 Lord of All Lutheran Church	107643199RR0001	Strathmore	AB
	762008290MC000		
332 LCC Worker Benefits Services Inc.	1	Winnipeg	MB
333 Hope Lutheran Church	107490229RR0001	Port Coquitlam	BC
335 St. Matthew Evangelical Lutheran Church &			
Christian Academy	108031386RR0001	Stony Plain	AB
336 Grace Lutheran Church	107444788RR0001	Osoyoos	BC
337 Shepherd King Lutheran Church	132171299RR0001	Calgary	AB
338 St. Peter Lutheran Church	119197564RR0001	Stettler	AB
340 St. John's Lutheran Church	131934721RR0001	Sherwood Park	AB
341 St. Albert Evangelical Lutheran Church	108002452RR0001	St. Albert	AB
342 Peace Lutheran Church	132201229RR0001	Fort St. John	BC
343 Our Saviours Community LC/Bethlehem			
Lutheran	119074243RR0001	Hussar	AB
344 Bethlehem Evangelical Lutheran Church	132003922RR0001	Outlook	SK
345 St. Peter's Evangelical Lutheran Church	108038779RR0001	Winnipeg	MB
346 Ascension Lutheran Church	118790922RR0001	Calgary	AB
354 Kingman Lutheran Parish		Kingman	AB
355 Hope Lutheran Church	118962778RR0001	Edmonton	AB

Employer	Business Number	City	Province
356 Faith Evangelical Lutheran Church	107373433RR0001	Sundre	AB
357 Holy Trinity Lutheran Church	118961671RR0001	Luseland	SK